\_\_\_\_\_

# SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 1999

OF

[ ] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission File Number 1-9977

MERITAGE CORPORATION
(Exact Name of Registrant as Specified in Its Charter)

Maryland
(State or Other Jurisdiction)
of Incorporation or Organization)

86-0611231 (I.R.S. Employer Identification No.)

6613 North Scottsdale Road, Suite 200 Scottsdale, Arizona (Address of Principal Executive Offices)

85250 (Zip Code)

(480) 998-8700 (Registrant's Telephone Number, Including Area Code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days: Yes [X] No  $[\ ]$ .

As of November 7, 1999; 5,464,906 shares of Meritage Corporation common stock were outstanding.

\_\_\_\_\_\_

MERITAGE CORPORATION AND SUBSIDIARIES FORM 10-Q FOR THE QUARTER ENDED SEPTEMBER 30, 1999 TABLE OF CONTENTS

PAGE NO.

#### PART I. FINANCIAL INFORMATION

	Item 1.	Financial Statements:	
		Consolidated Balance Sheets as of September 30, 1999 and December 31, 1998	3
		Consolidated Statements of Earnings for the Three and Nine Months ended September 30, 1999 and 1998	4
		Consolidated Statements of Cash Flows for the Nine Months ended September 30, 1999 and 1998	5
		Notes to Consolidated Financial Statements	6
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	10
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk	15
ART	II. OTH	ER INFORMATION	
	Item 6.	Exhibits and Reports on Form 8-K	15
TONI	MIIDDO		

2

# MERITAGE CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (UNAUDITED)

	SEPTEMBER 30, 1999	DECEMBER 31, 1998
ASSETS		
Cash and cash equivalents	\$ 3,835,754	\$ 12,386,806
Real estate under development	180,151,276	104,758,530
Deposits on real estate under option or		
contract	13,665,743	7,338,406
Other receivables Deferred tax asset	2,004,365 571,436	2,460,966 6,935,000
Goodwill	19,008,403	14,640,712
Property and equipment, net	4,032,473	2,566,163
Other assets	1,581,314	1,163,737
Total Assets	\$224,850,764	\$152,250,320
	========	========
LIABILITIES	¢ 22 020 074	¢ 24 060 170
Accounts payable and accrued liabilities Home sale deposits	\$ 33,832,074 11,098,691	\$ 34,068,178 8,587,245
Notes payable	95,733,724	37,204,845
Minority interest in consolidated joint	30,700,721	0,,201,010
ventures	186,937	110,922
Total Liabilities	140,851,426	79 <b>,</b> 971 <b>,</b> 190
STOCKHOLDERS' EQUITY		
Common stock, par value \$.01 per share; 50,000,000 Shares authorized; issued and outstanding - 5,474,906 Shares at September 30, 1999, and 5,334,942 shares		
at December 31, 1998	54,749	53,349
Additional paid-in capital	100,258,418	99,319,669
Accumulated deficit	(16,200,867)	(27,093,888)
Less cost of shares held in treasury (10,000 shares)	(112,962)	
(10) ood shares)		
Total Stockholders' Equity	83 <b>,</b> 999 <b>,</b> 338	72,279,130
Total Liabilities and Stockholders' Equity	\$224,850,764 =======	\$152,250,320 ======

See accompanying notes to consolidated financial statements

3
MERITAGE CORPORATION AND SUBSIDIARIES
CONSOLIDATED STATEMENTS OF EARNINGS
(UNAUDITED)

<TABLE> <CAPTION>

CAFIION	THREE MONTHS ENDED SEPTEMBER 30,		NINE MONT SEPTEN	THS ENDED MBER 30,
	1999	1998	1999	1998
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Home sales revenues	\$ 76,786,202	\$ 68,416,768	\$ 204,739,270	\$ 160,538,271
Cost of home sales	(62,231,991)	(54,447,139)	(164,364,352)	(129,771,511)
Gross profit	14,554,211	13,969,629	40,374,918	30,766,760
Commissions and other sales costs	(4,572,040)	(3,661,365)	(12,479,899)	(8,555,753)
General and administrative expense	(3,531,611)	(3,062,087)	(10,355,955)	(7,242,025)
Interest expense	(1,725)	(119,030)	(4,404)	(314,624)
Other income, net	361,938	434,298	1,395,659	720,705
Residual interest and real estate				
loan interest income				5,230,549
Minority interest in net income of				
consolidated joint ventures		(1,395,443)		(1,395,443)
Earnings before income taxes	6,810,773	6,166,002	18,930,319	19,210,169
Income taxes	2,783,597	1,898,000	8,037,298	2,794,000
Net earnings	\$ 4,027,176	\$ 4,268,002	\$ 10,893,021	\$ 16,416,169
	========	========	========	=========

Basic earnings per share	\$ .74	\$ .80	\$ 2.00	\$ 3.09
Diluted earnings per share				

 \$ .67 | \$ .70 | \$ 1.80 | \$ 2.68 |See accompanying notes to consolidated financial statements.

# 4 MERITAGE CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

	NINE MONTHS ENDE	D SEPTEMBER 30, 1998
CASH FLOWS FROM OPERATING ACTIVITIES:  Net earnings Adjustments to reconcile net earnings to net	\$ 10,893,021	\$ 16,416,169
cash used in operating activities: Depreciation and amortization Minority interest in net income of	1,607,658	963,540
consolidated joint ventures		1,395,443
Deferred tax expense	6,363,564	1,534,000
Stock option compensation expense	444,987	1,040,342
Gain on sales of residual interests		(5,180,046)
Increase in real estate under development Increase in deposits on real estate under	(75,392,746)	(32,675,132)
option or contract (Increase) decrease in other receivables	(6,327,337)	(2,580,682)
and other assets Increase (decrease) in accounts payable and	39,024	(579 <b>,</b> 583)
accrued liabilities	1,638,776	(3,718,429)
Increase in home sale deposits	2,511,446	6,030,740
Net cash used in operating activities	(58,221,607)	(17,353,638)
CASH FLOWS FROM INVESTING ACTIVITIES: Cash acquired in merger/acquisition Cash paid for merger/acquisition Purchases of property and equipment Proceeds from sales of residual interest	(6,966,890) (2,273,634)	785,403 (9,744,607) (986,598) 6,600,000
Net cash used in investing activities	(9,240,524)	(3,345,802)
CASH FLOWS FROM FINANCING ACTIVITIES: Borrowings Repayment of borrowings Purchase of treasury shares Stock options exercised	213,469,099 (154,940,220) (112,962) 495,162	120,341,231 (102,456,502)  304,796
Net cash provided by financing activities	58,911,079 	18,189,525
Net decrease in cash and cash equivalents Cash and cash equivalents at beginning	(8,551,052)	(2,509,915)
of period	12,386,806	8,245,392
Cash and cash equivalents at end of period	\$ 3,835,754 =======	\$ 5,735,477

See accompanying notes to consolidated financial statements.

5
NOTES TO CONSOLIDATED FINANCIAL STATEMENTS
(UNAUDITED)

# NOTE 1 - ORGANIZATION AND BASIS OF PRESENTATION

We develop, construct and sell new high-quality, single family homes in the semi-custom luxury, move-up and entry-level markets. Our operations in the Phoenix and Tucson, Arizona metropolitan markets are under the Monterey Homes and Meritage Homes of Arizona brand names, in the Dallas/Fort Worth, Austin and Houston, Texas markets we use the Legacy Homes name and in the San Francisco Bay and Sacramento, California markets we are known as Meritage Homes of Northern California. We have recently undergone significant growth and are pursuing a strategy of expanding our operations.

BASIS OF PRESENTATION. The consolidated financial statements include the accounts of Meritage Corporation and its subsidiaries. Intercompany balances and transactions have been eliminated in consolidation and certain prior period amounts have been reclassified to be consistent with current financial statement presentation. Results for the first six months of 1998 do not include the operations of Meritage Homes of Northern California, which we acquired on July 1, 1998. In the opinion of management, the unaudited consolidated financial

statements reflect all adjustments, consisting only of normal recurring adjustments, necessary to fairly present our financial position and results of operations for the periods presented. The results of operations for any interim period are not necessarily indicative of results to be expected for a full fiscal year.

# NOTE 2 - REAL ESTATE UNDER DEVELOPMENT AND CAPITALIZED INTEREST

The components of real estate under development follow (in thousands):

	SEPTEMBER 30, 1999	DECEMBER 31, 1998
Homes under contract, in production	\$ 88,900	\$ 44,186
Finished lots and lots under development	63 <b>,</b> 932	46,558
Model homes and homes held for resale	27,319	14,015
	\$180 <b>,</b> 151	\$104 <b>,</b> 759
	=======	=======

We capitalize certain interest costs incurred during development and construction. Capitalized interest is allocated to real estate under development and charged to cost of home sales when the units are delivered. Summaries of interest capitalized and interest expensed follow (in thousands):

<TABLE> <CAPTION>

	QUARTER ENDED SEPTEMBER 30,		NINE MONTHS ENDE SEPTEMBER 30,	
	1999	1998	1999	1998
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Beginning unamortized capitalized interest	\$ 2,653	\$ 2,130	\$ 1,982	\$ 1,890
Interest capitalized	1,942	1,608	4,541	3,092
Interest amortized in cost of home sales	(1,118)	(1,199)	(3,046)	(2,443)
Ending unamortized capitalized interest	\$ 3,477	\$ 2,539	\$ 3,477	\$ 2,539
	======	======	======	======
Interest incurred	\$ 1,944	\$ 1 <b>,</b> 727	\$ 4,545	\$ 3,407
Interest capitalized		(1,608)		
Interest expensed	\$ 2	s 119	s 4	\$ 315
incorest expensed	======	======	======	======

  |  |  |  |MERITAGE CORPORATION AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS - (CONTINUED)

(UNAUDITED)

# NOTE 3 - NOTES PAYABLE

Notes payable consist of the following (in thousands):

<TABLE>

	SEPTEMBER 30, 1999	DECEMBER 31, 1998
<pre>\$ \$60 million bank construction line of credit, interest payable monthly approximating prime (8.25% at September 30, 1999) or LIBOR (30 day LIBOR 5.4% at September 30, 1999), plus 2.25% payable at the earlier of close of escrow, maturity date of individual homes within the line or June 9,</pre>	<c></c>	<c></c>
2000, secured by first deeds of trust on homes \$80 million bank construction line of credit, interest payable monthly approximating prime or LIBOR plus 2.25%, payable at the earlier of close of escrow, maturity date of individual homes within the line or July 31, 2000, secured by first deeds of trust on homes	\$27,362 34,982	\$4,641 10,925
\$20million bank acquisition and development credit facility, interest payable monthly approximating prime or LIBOR plus 2.25%, payable at the earlier of funding of construction financing, the maturity date of individual projects within the line or June 19, 2000, secured by first deeds of trust on land		3,314
\$15 million unsecured revolving line of credit, interest payable monthly approximating prime, maturing on December 17, 1999	11,116	

Other acquisition and development credit facilities with commitments totaling \$4.5 million, interest payable monthly, ranging from prime to prime plus .25%; payable at the earlier of funding of construction financing or the maturity date of the individual projects, secured by first deeds of		
trust on land	2,144	2,407
Senior unsecured notes, maturing September 15, 2005, annual interest of 9.10% payable quarterly, principal payable in three equal installments on September 15, 2003, 2004 and		
2005	15,000	15,000
Other	30	918
Total	\$95 <b>,</b> 734	\$37 <b>,</b> 205

</TABLE>

MERITAGE CORPORATION AND SUBSIDIARIES
NOTES TO CONSOLIDATED FINANCIAL STATEMENTS - (CONTINUED)
(UNAUDITED)

#### NOTE 4 - EARNINGS PER SHARE

A summary of the reconciliation from basic earnings per share to diluted earnings per share for the three and nine months ended September 30, 1999 and 1998 follows (in thousands, except per share amounts):

	QUARTER ENDED SEPTEMBER 30,			THS ENDED BER 30,
		1998	1999	1998
Net earnings Weighted average shares outstanding -	\$4 <b>,</b> 027	\$4,268	\$10,893	\$16,416
basic	5,463 	5,317 	5,448	5,313
Basic earnings per share	\$ .74	\$ .80	\$ 2.00	
Basic EPS - Weighted average shares outstanding	5,463	5,317	5,448	5,313
Effect of dilutive securities:				
Contingent shares Stock options	70 491	129 622	83 527	150 660
Weighted average shares outstanding -				
dilutive	6,024 	6,068 	6,058 	6,123 
Diluted earnings per share	\$ .67 =====	\$ .70 =====	\$ 1.80 =====	\$ 2.68

#### NOTE 5 - TREASURY STOCK

In June 1999, we began acquiring shares of our common stock in connection with a stock repurchase program announced in April 1999. The program authorizes us to purchase up to \$6 million of common stock from time to time on the open market or in privately negotiated transactions at price levels we consider attractive. As of September 30, 1999, we have purchased 10,000 shares of common stock at an aggregate cost of \$112,962. The purpose of the stock repurchase program is to help us achieve our long-term goal of enhancing stockholder value.

#### NOTE 6 - ACQUISITIONS

On June 15, 1998, we signed a definitive agreement with Sterling Communities, S.H. Capital, Inc., Sterling Financial Investments, Inc., Steven W. Hafener and W. Leon Pyle (together, the Sterling Entities), to acquire substantially all of the assets of Sterling Communities. The transaction was effective as of July 1, 1998. Assets acquired principally consist of real property and other residential homebuilding assets located in the San Francisco Bay and Sacramento areas of California. Operations of the Sterling Entities continue under the name Meritage Homes of Northern California.

Consideration paid for the assets and stock acquired, and various liabilities assumed, consisted of \$6.9 million in cash and additional consideration to be paid for up to four years after the transaction date. We used the purchase method of accounting and the purchase price was allocated among our net assets based on their estimated fair market value at the

transaction date. Goodwill of approximately \$2.2 million was recorded, which is being amortized over 20 years. The additional consideration will be equal to 20% of the pre-tax income of Meritage's California division and will be expensed as earned.

In connection with the 1997 acquisition of Legacy Homes, additional consideration of \$5.2 million was recorded as a non-cash addition to goodwill and accrued liabilities for the nine month period ended September 30, 1999. The purchase agreement provided that additional consideration was not to exceed \$15 million and was based on the Company's earnings. The additional consideration reached the \$15 million limit during the first quarter of 1999. The \$5.2 million final payment will be made in January 2000, and will represent additional cash paid for the acquisition of Legacy Homes.

8
MERITAGE CORPORATION AND SUBSIDIARIES
NOTES TO CONSOLIDATED FINANCIAL STATEMENTS - (CONTINUED)
(UNAUDITED)

#### NOTE 7 - INCOME TAXES

Components of income tax expense are as follows (in thousands):

	QUARTER ENDED SEPTEMBER 30,			ITHS ENDED IBER 30,
	1999	1998	1999	1998
Current taxes:				
Federal	\$ 724	\$ 99	\$ 964	\$ 325
State	250	265	709	935
	974	364	1,673	1,260
Deferred taxes:				
Federal	1,769	1,534	6,236	1,534
State	40		128	
	1,809	1,534	6,364	1,534
Total	\$2,783	\$1,898	\$8,037	\$2,794
	=====	=====	=====	=====

#### CARRYFORWARDS

At September 30, 1999, our federal net operating loss carryforward had been fully utilized.

#### NOTE 8 - SEGMENT INFORMATION

We classify our operations into three primary geographic segments: Arizona, Texas and California. These segments generate revenues through the sale of homes to external customers. We are not dependent on any one major customer.

Operational information relating to the different business segments follows. Information for the first six months of 1998 has not been included for the California operations that were acquired July 1, 1998. Certain information has not been included by segment due to the immateriality of the amount to the segment or in total. We evaluate segment performance based on several factors, of which the primary financial measure is earnings before interest and taxes (EBIT). The accounting policies of the business segments are the same as those described in Notes 1 and 2 for the Company. There are no significant transactions between segments.

<TABLE>

9

	QUARTER ENDED SEPTEMBER 30,		NINE MONTHS ENDED SEPTEMBER 30,	
	1999	1998	1999	1998
		(in tho	usands)	
<pre><s> HOME SALES REVENUE:</s></pre>	<c></c>	<c></c>	<c></c>	<c></c>
Texas Arizona California	\$ 43,604 26,056 7,126	\$ 34,767 19,911 13,739	\$ 116,399 75,740 12,600	\$ 93,613 53,186 13,739
Total	\$ 76,786 ======	\$ 68,417 ======	\$ 204,739 ======	\$160,538 ======
EBIT: Texas Arizona	\$ 5,790 2,565	\$ 5,040 1,822	\$ 15,344 8,270	\$ 13,069 4,594

California Corporate and other		515 (937)		1,731 (1,109)		1,288 (2,921)		1,731 2,574	
Total	\$	7,933	\$	7,484	\$	21,981	\$	21,968	
AMORTIZATION OF CAPITALIZED INTEREST: Texas Arizona California	\$	445 559 114	\$	316 433 450	\$	1,112 1,750 184	\$	818 1,175 450	
Total	\$ ==	1,118	\$ ==	1,199	\$ ==	3,046	- \$ =	2,443	
			AT SEI	PTEMBER 30 1999	),		AT D	ECEMBER 1998	31,
ASSETS: Texas Arizona California Corporate				103,646 77,355 42,109 1,741				64,448 58,758 12,321 16,723	
Total			\$2	224,851			\$	152,250	

</TABLE>

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

This Quarterly Report on Form 10-Q contains forward-looking statements. The words "believe," "expect," "anticipate," and "project" and similar expressions identify forward-looking statements, which speak only as of the date the statement was made. Such forward-looking statements are within the meaning of that term in Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. Such statements may include, but are not limited to, projections of revenues, income or loss, capital expenditures, plans for future operations, financing needs or plans, the impact of inflation, the impact of changes in interest rates, plans relating to our products or services, potential real property acquisitions, and new or planned development projects, as well as assumptions relating to the foregoing.

Statements in Exhibit 99 to this Quarterly Report on Form 10-Q and in our Annual Report on Form 10-K for the year ended December 31, 1998, including the Notes to the Consolidated Financial Statements and "Management's Discussion and Analysis of Financial Condition and Results of Operations," describe factors that could contribute to or cause such differences. Additional factors that could cause actual results to differ materially from those expressed in such forward-looking statements are set forth in "Business" and "Market for the Registrant's Common Stock and Related Stockholder Matters" in our December 31, 1998 Annual Report on Form 10-K.

### RESULTS OF OPERATIONS

The following discussion and analysis provides information regarding the results of operations of Meritage and its subsidiaries for the three and nine months ended September 30, 1999 and September 30, 1998. All material balances and transactions between Meritage and its subsidiaries have been eliminated. This discussion should be read in conjunction with our Annual Report on Form 10-K for the year ended December 31, 1998. In management's opinion, the data reflects all adjustments, consisting of only normal recurring adjustments, necessary to fairly present our financial position and results of operations for the periods presented. The results of operations for any interim period are not necessarily indicative of results expected for a full fiscal year. Comparative information for the first six months of 1998 has not been included for the California operations, which we acquired in July 1998.

# HOME SALES REVENUE

Home sales revenue is the product of the number of units closed during the period and the average sales price per unit. Comparative 1999 and 1998 housing revenues follow (dollars in thousands):

`	Τ.	A.	ο.	ЦΕ	٠,	_	
<	C	Α	P'	rı	ΙC	NC	

<caption></caption>								
PERCENTAGE	QUARTER	R ENDED	DOLLAR/UNIT	PERCENTAGE	NINE MONT	HS ENDED	DOLLAR/UNIT	
THICHNINGS	SEPTEME	BER 30,	INCREASE	INCREASE	SEPTEMB:	ER 30,	INCREASE	INCREASE
	1999	1998	(DECREASE)	(DECREASE)	1999	1998	(DECREASE)	
(DECREASE)								
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
<s> Total Dollars</s>	<c></c>	<c> \$68,417</c>	<c> \$ 8,369</c>	<c></c>	<c> \$204,739</c>	<c> \$160,538</c>	<c> \$ 44,201</c>	<c></c>

Units closed Average sales price	399 \$ 192.5	346 \$ 197.7	53 \$ (5.2)	15% (3)%	1,030 \$ 198.8	874 \$ 183.7	156 \$ 15.1	18% 8%
Texas								
Dollars	\$43,604	\$34,767	\$ 8,837	25%	\$116 <b>,</b> 399	\$ 93,613	\$ 22 <b>,</b> 786	24%
Units closed	283	255	28	11%	758	681	77	11%
Average sales price	\$ 154.1	\$ 136.3	\$ 17.8	13%	\$ 153.6	\$ 137.5	\$ 16.1	12%
Arizona								
Dollars	\$26,056	\$19,911	\$ 6,145	31%	\$ 75,740	\$ 53,186	\$ 22,554	42%
Units closed	97	59	38	64%	238	161	77	48%
Average sales price	\$ 268.6	\$ 337.5	\$ (68.9)	(20)%	\$ 318.2	\$ 330.3	\$ (12.1)	(4)%
California								
Dollars	\$ 7,126	\$13,739	\$(6,613)	(48)%	\$ 12,600	\$ 13,739	\$ (1,139)	(8)%
Units closed	19	32	(13)	(41)%	34	32	2	6%
Average sales price								

 \$ 375.1 | \$ 429.3 | \$ (54.2) | (13)% | \$ 370.6 | \$ 429.3 | \$ (58.7) | (14)% |The increase in revenues and number of units closed in 1999 compared to 1998 resulted mainly from our strong 1999 market performance in Arizona and Texas. The 1999 third quarter decrease in California revenue resulted from a 1998 third quarter which included revenue from close-out communities acquired in the Sterling acquisition, which were replaced by new communities that are just beginning to produce closing revenue. The average sales price in Arizona is decreasing as more homes are sold in our newer, lower priced communities increases.

#### GROSS PROFIT

Gross profit is home sales revenue, net of housing cost of sales. Housing cost of sales includes developed lot costs, unit construction costs, amortization of common community costs (such as the cost of model complexes and architectural, legal and zoning costs), interest, sales tax, warranty, construction overhead and closing costs. Comparative 1999 and 1998 housing gross profit follows (dollars in thousands):

	QUARTER	ENDED SEP	11 TEMBER 30,	NINE MONTHS	ENDED SE	PTEMBER 30,
	1999	1998	INCREASE/ (DECREASE)	1999	1998	INCREASE
Dollars Percentage of	\$14,554	\$13,970	\$ 584	\$40,375	\$30,767	\$9,608
housing revenues	19.0%	20.4%	(1.4)%	19.7%	19.2%	.5%

The dollar increase in gross profit for the three and nine months ended September 30, 1999 over the prior year's periods is attributable to the increase in number of units closed. In the September quarter there is an increase in the number of lower priced homes closed, especially in Arizona, and a corresponding decrease in margins.

# COMMISSIONS AND OTHER SALES COSTS

The increase in commissions and other sales costs in the third quarter and first nine months of 1999 compared with the same periods of 1998 are directly related to costs incurred to support growth in closing revenue. The growth in these costs has slightly exceeded revenue growth, primarily due to the need to incur model home operating and advertising costs associated with new communities, particularly in our more newly established Northern California region, several months prior to those communities producing closing revenue. Comparative 1999 and 1998 commissions and other sales costs follows (dollars in thousands):

	QUARTER	ENDED SEP	TEMBER 30,	NINE MONTH	IS ENDED	SEPTEMBER 30,
	1999	1998	INCREASE	1999	1998	INCREASE
Total						
	\$4,572	\$3,661	\$911	\$12,480	\$8,556	\$3,924
Percentage of	6.00	F 40	60	6 10	F 20	0.0
housing revenues	6.0%	5.4%	.6%	6.1%	5.3%	.8%

# GENERAL AND ADMINISTRATIVE EXPENSES

General and administrative expenses were approximately \$3.5 million (4.6% of revenue) in the third quarter of 1999 as compared with approximately \$3.1 million (4.5% of revenue) in the third quarter of 1998. The increase for the nine months ended September 30, 1999 over the 1998 period includes approximately \$600,000 related to an employment agreement buyout of a former managing director. Operating costs in the first nine months of 1999 have also increased due to overhead incurred related to our expansion into California and the start-up of our new Meritage Division in Phoenix, Arizona.

#### OTHER INCOME

Other income decreased for the three months ended September 30, 1999 compared with the same quarter last year due to fewer miscellaneous fees being collected. The increase in other income for the nine month period ended September 30, 1999, primarily is due to management fees earned by the California division, which is included for the full nine months, and an increase in revenue from the mortgage operations in Texas.

#### RESIDUAL INTEREST AND REAL ESTATE LOAN INTEREST INCOME

The remainder of our mortgage security portfolio was sold for a gain of approximately \$2.0 million in the second quarter of 1998, and as a result, there is no residual interest or real estate loan interest income in 1999.

#### INCOME TAXES

The increase in income taxes to \$2.8 million for the quarter ended September 30, 1999 from \$1.9 million in the prior year's period resulted from a higher effective tax rate in the current year due to the use of the net operating loss ("NOL") carryforward for accounting purposes in 1998. In future periods we expect to have an effective tax rate approximating the statutory federal and state tax rates as our NOL carryforward is now fully utilized.

#### SALES CONTRACTS

Sales contracts for any period represent the number of homes ordered by customers (net of cancellations) multiplied by the average sales price per unit ordered. Comparative 1999 and 1998 sales contracts follow (dollars in thousands):

12

		12						
<table></table>								
<caption></caption>								
	OHART	ER ENDED	DOLLAR/UNIT	PERCENTAGE	NINE MONT	HS ENDED	DOLLAR/UNIT	
PERCENTAGE	QUIII	DI DINDED	DOBLIN, ONII	IBROBIVITION	WIND HOW	IIO DIVDDD	DODDING ONLL	
FERCENTAGE	CEDME	WDED 20	TMODELLOE	TMODELLOE	CEDMEND	DD 20	TMODELLOE	TMODELLO
		MBER 30,	INCREASE	INCREASE	SEPTEMB	•	INCREASE	INCREASE
	1999	1998	(DECREASE)	(DECREASE)	1999	1998	(DECREASE)	
(DECREASE)								
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Total								
Dollars	\$93,955	\$65,327	\$ 28,628	44%	\$295,969	\$225,455	\$ 70,514	31%
Units ordered	405	309	96	31%	1,455	1,196	259	22%
Average sales price	\$ 232.0	\$ 211.4	\$ 20.6	10%	\$ 203.4	\$ 188.5	\$ 14.9	8%
invertage bares price	¥ 232.0	Y 211.1	¥ 20.0	100	¥ 200.1	4 100.0	ų 11.J	0 0
Texas								
Dollars	\$36,562	\$36,007	\$ 555	2%	\$156,178	\$134,440	\$ 21,738	16%
	219	234			996			10% 7%
Units ordered			(15)	(6)%		934	62	
Average sales price	\$ 167.0	\$ 153.9	\$ 13.1	9%	\$ 156.8	\$ 143.9	\$ 12.9	9%
Arizona								
Dollars	\$39 <b>,</b> 027	\$28 <b>,</b> 687	\$ 10,340	36%	\$100 <b>,</b> 265	\$ 90,382	\$ 9 <b>,</b> 883	11%
Units ordered	132	74	58	78%	344	261	83	32%
Average sales price	\$ 295.7	\$ 387.6	\$ (91.9)	(24)%	\$ 291.5	\$ 346.3	\$ (54.8)	(16)%
California								
Dollars	\$18,366	\$ 633	\$ 17,733	2,801%	\$ 39,526	\$ 633	\$ 38,893	6,144%
Units Ordered	54	1	53	5,300%	115	1	114	11,400%
Average sales price	\$ 340.1	\$ 633	\$ (292.9)	(46)%	\$ 343.7	\$ 633	\$ (289.3)	(46)%
<pre></pre>								

 A 240.T | 7 633 | Y (232.3) | (40) 0 | 7 243.7 | 7 000 | Y (209.3) | (40) 0 || / IADLE/ |  |  |  |  |  |  |  |  |
We do not include sales contingent upon the sale of a customer's existing home as a sales contract until the contingency is removed. Historically, we have experienced a cancellation rate of less than 20% of gross sales. Total sales contracts increased in 1999 compared to 1998 due mainly to the expansion into California and the increased sales of our lower priced homes in Arizona.

## NET SALES BACKLOG

Backlog represents net sales contracts that have not closed. Comparative 1999 and 1998 net sales backlog follows (dollars in thousands):

			DOLLAR/UNIT	PERCENTAGE
	AT SEPTI	EMBER 30,	INCREASE	INCREASE
	1999	1998	(DECREASE)	(DECREASE)
Total				
Dollars	\$236,524	\$182,461	\$ 54,063	30%
Units in backlog	1,113	835	278	33%
Average sales price	\$ 212.5	\$ 218.5	\$ (6)	(3)%

Dollars Units in backlog Average sales price	\$116,957 741 \$ 157.8	\$ 82,857 557 \$ 148.8	\$ 34,100 184 \$ 9.0	41% 33% 6%
Arizona				
Dollars	\$ 90,904	\$ 94,142	\$ (3,238)	(4)%
Units in backlog	286	268	18	7%
Average sales price	\$ 317.8	\$ 351.3	\$ (33.5)	(1)%
California				
Dollars	\$ 28,663	\$ 5,462	\$ 23,201	425%
Units in backlog	86	105	76	760%
Average sales price	\$ 333.3	\$ 546.2	(212.9)	(39)%

Total dollar backlog at September 30, 1999 increased 30% over the 1998 period due to a corresponding increase in new sales contracts entered into in 1999. Units in backlog at September 30, 1999 increased 33% over the same period in 1998 due to the increase in net orders caused in part by expansion into California, and to some extent, a lengthening of the construction cycle for our homes. The average sales price of homes in backlog in Arizona is decreasing as the percentage of lower priced homes sold in that region increases.

13

#### LIOUIDITY AND CAPITAL RESOURCES

Our principal uses of working capital are land purchases, lot development and home construction. We use a combination of borrowings and funds generated by operations to meet our working capital requirements.

At September 30, 1999 we had short-term secured revolving construction loan facilities totaling \$140 million and had \$24.5 million in acquisition and development facilities, of which approximately \$62.3 and \$7.2 million were outstanding, respectively. An additional \$18.8 million of unborrowed funds supported by approved collateral were available under these credit facilities at that date. We also have \$15 million outstanding in unsecured, senior subordinated notes due September 15, 2005, which were issued in October 1998, and \$15 million in an unsecured credit facility of which \$11.1 million is outstanding.

Management believes that the current borrowing capacity, cash on hand at September 30, 1999 and anticipated cash flows from operations are sufficient to meet our liquidity needs for the foreseeable future. There is no assurance, however, that future amounts available from our sources of liquidity will be sufficient to meet future capital needs. The amount and types of indebtedness that we incur may be limited by the terms of the indenture governing our senior subordinated notes and by the covenants and other terms of our credit agreements.

#### YEAR 2000 COMPLIANCE

The year 2000 presents potential concerns for business computing due to calculation problems from the use of a two-digit format as the year changes from 1999 to 2000. The problem affects certain computer software, hardware, and other systems containing processors and embedded chips. Consequently, information technology ("IT") systems and non-IT systems (collectively, "business systems") may not be able to accurately process certain transactions before, during, or after January 1, 2000. As a result, business and governmental agencies are at risk for potential disruption from business systems malfunctions or failures. This is commonly referred to as the Year 2000 ("Y2K") issue. We could be impacted by failures of our own business systems as well as those of our suppliers and business partners. We have implemented our Y2K compliance program that consisted of business systems identification, testing and remediation, assessments of critical suppliers, and contingency planning.

The compliance program's first component was the identification of our business systems for purposes of evaluating which systems are Y2K compliant and which will be replaced or remediated. This phase is complete.

The second part of the program is the evaluation and replacement or remediation of our business systems that are not Y2K compliant. We have converted to new versions of substantially all of our homebuilding database systems and our replacement or remediation program is substantially complete.

We have identified critical suppliers and business partners ("key business partners") and have taken steps to determine their Y2K readiness. These steps include various types of inquiries. Because of the number of business systems used by key business partners and the varying levels of Y2K readiness, it is difficult to assess the likelihood and impact of a malfunction due to this issue. We are not currently aware of any business relationships with key business partners that we believe will likely result in a significant disruption of our business. However, a Y2K failure could occur and have an adverse impact on us. Management currently believes that our greatest risk is with suppliers, banking and financial institutions, and suppliers of telecommunications services, all of which are operating within the United States. Potential consequences of Meritage or its key business partners having business systems

that are not Y2K compliant include delays in receiving  $\mbox{homebuilding}$  components and supplies.

Concurrent with the remediation and evaluation of our business systems and those of our key business partners, we have developed contingency plans to decrease the risks that could occur in the event of a Y2K related business disruption. Contingency plans include increasing the level of homebuilding components, securing additional financing or other actions management deems advisable. Estimated costs associated with developing and implementing contingency measures are expected to be minimal.

14

The remediation and testing of our business systems has cost approximately \$160,000. These costs have been expensed in the period incurred and funded through cash flows from operations. The future financial impact is not expected to be material to our financial position or results of operations.

#### ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We do not trade in derivative financial instruments and at September 30, 1999 had no significant derivative financial instruments. We do have other financial instruments in the form of notes payable and senior debt, which are at fixed interest rates. Our lines of credit and credit facilities are at variable interest rates and are subject to market risk in the form of interest rate fluctuations.

#### PART II OTHER INFORMATION

#### ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

#### (a) Exhibits

Exhibit Number	Description	Page or Method of Filing
10.23	\$15 Million Credit Agreement by and among Meritage Corporation and California Bank and Trust, Dated as of September 17, 1999	Filed herewith
27	Financial Data Schedule	Filed herewith
99	Private Securities Reform Act of 1995 Safe Harbor Compliance Statement for Forward-Looking Statements	Filed herewith

# (b) Reports on Form 8-K

No reports on form 8-K were filed during the quarter ended September 30, 1999.

#### 15 SIGNATURES

Pursuant to the requirements of Section 13 or 15 (d) of the Securities Exchange Act of 1934, the registrant has duly cause this report on Form 10-Q to be signed on its behalf by the undersigned, thereunto duly authorized, this 15th day of November 1999.

MERITAGE CORPORATION, a Maryland Corporation

By /s/ Larry W. Seay

Larry W. Seay Chief Financial Officer and Vice President-Finance (Principal Financial Officer and Duly Authorized Officer) \_\_\_\_\_\_

CREDIT AGREEMENT

by and among

MERITAGE CORPORATION

The Banks Named Herein

and

CALIFORNIA BANK & TRUST

as Administrative Agent and as Issuing Bank

Dated as of

September 17, 1999

\_\_\_\_\_\_

# TABLE OF CONTENTS

		Pag
RECITALS .		1
ARTICLE 1.	DEFINITION OF TERMS	2
1.1	Definitions Terms Generally	2 16
ARTICLE 2.	THE RLC	17
2.1	RLC Commitment	17
2.2	Revolving Line	17
2.3	RLC Notes	17
2.4	RLC	18
2.5	Excess Balance Repayment	20
2.6	Reduction of RLC Commitment	20
2.7	Conditions	20
2.8	Other RLC Advances	20
2.9	Assignment	21
2.10	Issuance of Letters of Credit	21
2.11	Issuance Procedure for Letters of Credit	22
2.12	Letter of Credit Fees and Costs	22
2.13	Disbursements	22
2.14	Reimbursement Obligations of Borrower	22
2.15	Nature of Reimbursement Obligations	23
2.16	Banks Obligation	23
2.17	Certain Requirements as to Letters of Credit	24
2.18	Risk Participations, Drawings, and Reimbursements	24
2.19	Repayment of Participations	26
2.20	Role of the Issuing Bank	27
ARTICLE 3.	PAYMENTS, FEES AND EURODOLLAR PROVISIONS	28
3.1	Payments	28
3.2	Fees	28
3.3	Computations	29
3.4	Maintenance of Accounts	29
3.5	Certain Contingencies	29
3.6	Increased Capital Requirements	29
3.7	Prepayment	30
ARTICLE 4.	EXTENSION	31

4 1	Dubonoine of the DIC Metamity Date	2.1
4.1	Extension of the RLC Maturity Date	31
ARTICLE 5.	CONDITIONS PRECEDENT	32
5.1	Closing	32
5.2	No Event of Default; No Material Adverse Change	33
5.3	Representations and Warranties	34
ARTICLE 6.	REPRESENTATIONS AND WARRANTIES	35
6.1	Organization and Good Standing	35
6.2	Authorization and Power	35
6.3 6.4	No Conflicts or Consents	35 35
6.5	Financial Condition	35
6.6	Full Disclosure	36
6.7	No Default	36
6.8	Significant Debt Agreements	36
6.9	No Litigation	36
6.10	Taxes	36
6.11	ERISA	36
6.12	Compliance with Law	36
6.13	Survival of Representations, Etc	36
6.14	Recitals	37
6.15	No Stock Purchase	37
6.16	Solvent	37
6.17	Advances	37
6.18	Year 2000 Compliance	37
6.19	Investment Company Act	38
6.20	Public Utility Holding Company Act	38
6.21	Title to Properties: Possession	38
6.22 6.23	Environmental and Safety Matters	38 38
6.24	Indebtedness - Borrower	38
6.25	Indebtedness - Subsidiaries	38
0.25	indebecaress substatutes	50
ARTICLE 7.	AFFIRMATIVE COVENANTS	39
7.1	Financial Statements, Reports and Documents	39
7.2	Payment of Taxes and Other Indebtedness	40
7.3	Maintenance of Existence and Rights; Conduct of Business	41
7.4	Notice of Default and Non-Compliance	41
7.5	Other Notices	41
7.6	Compliance with Credit Documents	41
7.7	Compliance with Significant Debt Agreements	41
7.8	Operations and Properties	41
7.9	Books and Records; Access	41
7.10	Compliance with Law	42
7.11 7.12	Authorizations and Approvals ERISA Compliance	42 42
7.12	Further Assurances	42
7.14	News Releases	42
7.15	Insurance	42
7.16	Change in Control or Management	43
7.17	Environmental Investigation	43
7.18	Inventory Inspection	43
7.19	Year 2000 Compliance	44
ARTICLE 8.	NEGATIVE COVENANTS	46
8.1	No Unsecured Debt Borrower	46
8.2	No Indebtedness Guarantors	46
8.3	Liens	47
8.4 8.5	Mergers, Consolidations, Transfers of Assets	47 47
8.6	Amendments to Organizational Documents  Margin Stock	47
8.7	Fiscal Year	47
8.8	Investments	47
8.9	Subordinated Debt Payments	47
8.10	Financial Covenants	48
ARTICLE 9.	EVENTS OF DEFAULT	49
9.1	Events of Default	49
9.2	Remedies Upon Event of Default	51
9.3	Performance by the Banks	53
9.4	Nature of the Obligations of Borrower	53
	-ii-	
ARTICLE 9A	BANKS; ADMINISTRATIVE AGENT; ISSUING BANK	54
9A.1	Appointment	54
9A.2	Liability	54
9A.3	Action by Administrative Agent	55
9A.4	Resignation	56
9A.5	Agent as Bank	56
9A.6	Ownership and Possession of Credit Documents	57
9A.7	Indemnification of Administrative Agent	57
9A.8	Independent Credit Analysis	57
9A.9	Process for Obtaining Approval of the Banks	58
9A.10 9A.11	Communications to the Banks	58 59
>₩• T T	Relationship with the Borrower	JJ

9A.12	Payments to or by the Banks	59
9A.13	Application of Payments	60
9A.14	Defaulting Banks	60
9A.15	Purchase of Defaulting Bank's Interest After Default	62
9A.16	Purchase Price and Payment for Defaulting Bank's Interest	62
9A.17	Issuing Bank	62
ARTICLE 10.	MISCELLANEOUS	63
10.1	Successors and Assigns	63
10.2	Waiver	66
10.3	Payment of Expenses; Indemnity	66
10.4	Notices	67
10.5	Governing Law	67
10.6	Invalid Provisions	68
10.7	Binding Effect	68
10.8	Entirety	68
10.10	Time of the Essence	68
10.11	Good Faith Standard	68
10.12	Headings	68
10.13	Survival	69
10.14	No Third Party Beneficiary	69
10.15	Schedules and Exhibits Incorporated	69
10.16	Setoff	69
10.17	JURY WAIVER	69
10.18	Counterparts	70

#### SCHEDULES:

Schedule 2.1 - Commitments of the Banks

Schedule 6.23 - Subsidiaries

Schedule 6.24 - Existing Indebtedness - Borrower

Schedule 6.25 - Existing Indebtedness - Subsidiaries

#### EXHIBITS:

Exhibit "A-1" - Form of Compliance Certificate per Section 7.1(c)

Exhibit "A-2" - Form of Quarterly Inventory Report per Section 7.1(g)

Exhibit "B-1" - Form of RLC Borrowing Base Inventory Report per Section 7.1(f)

Exhibit "B-2" - Form of RLC Borrowing Base Compliance Certificate per Section 7.1(f)

Exhibit "C" - Form of RLC Note per Section 2.3

Exhibit "D" - Form of Subdivision Admission Certificate

Exhibit "E" - Form of Continuing Guarantee

Exhibit "F" - Form of Assignment and Acceptance

Exhibit "G" - Administrative Details Reply Form

#### -iii-CREDIT AGREEMENT

BY THIS CREDIT AGREEMENT (together with any amendments or modifications, the "Credit Agreement"), entered into as of the 17th day of September, 1999 by and between MERITAGE CORPORATION, a Maryland corporation (the "Borrower"), the banks and financial institutions that are parties to this Credit Agreement from time to time (the "Banks"), and CALIFORNIA BANK & TRUST, a California banking corporation, as administrative agent for the Banks (in such capacity, together with any successor agent appointed hereunder, the "Administrative Agent") and as Issuing Bank (as hereinafter defined), in consideration of the mutual promises herein contained and for other valuable consideration, the parties hereto do hereby agree as follows:

### RECITALS

- A. Borrower has requested that the Banks establish a revolving line of credit (the "RLC") in the principal amount of \$15,000,000.00 to provide working capital financing.
- B. The Banks have agreed to do so upon the terms, conditions and provisions set forth herein.

Accordingly, the parties hereto agree as follows:  $\mbox{ARTICLE 1.}$ 

1.1 DEFINITIONS. For the purposes of this Credit Agreement, unless the context otherwise requires, the following terms shall have the respective meanings assigned to them in this Article 1 or in the section hereof referred to below:

"ADMINISTRATIVE AGENT": See the Preamble hereto.

"ADMINISTRATIVE DETAILS REPLY FORM" means an Administrative Details Reply Form substantially in the form of Exhibit "G".

"ADVANCE" means an RLC Advance.

"AFFILIATE" of any Person means any Person which, directly or indirectly, controls, is controlled by, or is under common control with, such Person. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.

"ASSIGNMENT AND ACCEPTANCE" means an Assignment and Acceptance substantially in the form of Exhibit "F", entered into by a Bank and an assignee.

"AUTHORIZED OFFICER" means one or more officers of the Borrower duly authorized (and so certified to the Banks by the corporate secretary of Borrower pursuant to a certificate of authority and incumbency from time to time satisfactory to the Banks in the exercise of the Banks' reasonable discretion), acting alone, to request Advances under the provisions of this Credit Agreement and execute and deliver documents, instruments, agreements, reports, statements and certificates in connection herewith.

"BANKS": See the Preamble hereto.

"BORROWER": See the Preamble hereto.

"BUSINESS DAY" means each day of the year, other than Saturdays, Sundays, Holidays and days on which banking institutions are generally authorized or obligated by law or executive order to close in California.

"CB&T" means CALIFORNIA BANK & TRUST, a California banking corporation.

"CHANGE IN CONTROL" means the occurrence or existence of either of the following events or conditions without the prior written consent of the Banks, if different than the state of affairs as of the Closing Date:

-2-

- (a) the acquisition by any Person or two or more Persons acting in concert of "beneficial ownership" (within the meaning of Rule 13d-3 promulgated by the SEC under the Exchange Act or as otherwise specified under the provisions of this Credit Agreement) of securities of Borrower having more than 50% of the ordinary voting power for the election of directors; or
- (b) the acquisition by any Person or two or more Persons acting in concert of Control of Borrower.

"CHIEF FINANCIAL OFFICER" means the chief financial officer of Borrower.

"CLOSING DATE" means September 17, 1999.

"CODE" means the Internal Revenue Code of 1986, as amended.

"CO-LENDER AGREEMENT" means collectively one or more Co-Lender Agreements, if any, hereafter entered into among the Administrative Agent, the Issuing Bank and the Banks with respect to this Credit Agreement and all amendments, modifications, supplements, and restatements thereof as entered into between the Administrative Agent and the Banks from time to time. As of the date hereof, there is no Co-Lender Agreement. With the agreement of Borrower, the Administrative Agent and the Banks may elect to amend this Credit Agreement in lieu of entering into a Co-Lender Agreement, in which case references to the Co-Lender Agreement shall also be amended to refer to the applicable provisions of this Credit Agreement.

"COLLATERAL" means RLC Borrowing Base Inventory for which Borrower shall have completed the Collateral Requirements to the satisfaction of the Administrative Agent and the Banks.

"COLLATERAL REQUIREMENTS" means:

(i) the delivery and recording of a deed of trust, assignment of rents, security agreement and fixture filing, executed by the applicable Obligor.

- (ii) the delivery and filing of UCC-1 Financing Statements, executed by the applicable Obligor.
- (iii) the delivery of an environmental  $\,$  indemnity agreement,  $\,$  executed by Borrower and all Guarantors.
- (iv) Either a current survey of the Collateral by a licensed surveyor acceptable to the Banks or a recorded plat map (final or preliminary), describing the boundaries of the Collateral and showing all means of ingress and egress, rights-of-way, easements (each of which shall be identified by docket and page or recording number where recorded) and all other customary and relevant information pursuant

-3-

to ALTA standards and any title company requirements. All surveys shall be certified to the Administrative Agent for the Banks and the title company issuing the Title Policy.

(v) An ALTA extended coverage mortgagee's title insurance policy [ALTA Loan Policy - 1970 (Rev. 10-17-70)] or similar policy acceptable to the Banks (the "Title Policy"), with such endorsements as the Banks may require, issued by a title insurance company satisfactory to the Banks insuring the lien of the applicable deed of trust to be a first and prior lien upon the Collateral as security for all Advances pursuant to the terms of this Credit Agreement, subject only to such exceptions as the Banks may expressly approve in writing.

"COMMITMENT" means the RLC Commitment.

"COMPLIANCE CERTIFICATE": See Section 7.1.(c) hereof.

"CONSOLIDATED INDEBTEDNESS" means the sum of all Indebtedness of Borrower, determined on a consolidated basis in accordance with GAAP, including without limitation, if not otherwise included, Subordinated Debt, current amounts due under development agreements and other funding obligations, and the stated amount of any outstanding letters of credit.

"CONSOLIDATED INTANGIBLE ASSETS" means Borrower's intangible assets as determined on a consolidated basis in accordance with GAAP, including without limitation goodwill and trademarks, but excluding any Tax Asset.

"CONSOLIDATED INTEREST INCURRED" means the sum of all interest paid or accrued by Borrower, including without limitation, that on Subordinated Debt, whether expensed or capitalized, for the prior four fiscal quarters, determined on a consolidated basis.

"CONSOLIDATED TANGIBLE NET WORTH" means Borrower's consolidated stockholders equity as determined on a consolidated basis in accordance with GAAP less its Consolidated Intangible Assets.

"CONSOLIDATED TANGIBLE NET WORTH REQUIREMENT" means the greater of (a) \$45,000,000.00 or (b) the sum of (i) ninety percent (90.0%) of the Consolidated Tangible Net Worth of Borrower as of March 31, 1999, (ii) ninety percent (90.0%) of any new stated capital or paid in capital acquired by Borrower during such fiscal period, and (iii) fifty percent (50.0%) of the Net Profit of Borrower for each fiscal year, commencing with that ending December 31, 1999, without giving effect to any losses reported for any such fiscal year.

"CONTROL" when used with respect to any Person means the power, directly or indirectly, to direct the management policies of such Person, whether through the ownership of voting securities, by contract or otherwise; and the terms "controlling" and "controlled" have meanings correlative to the foregoing.

-4-

"CONTROLLED GROUP" means, severally and collectively, the members of the group controlling, controlled by and/or in common control of Borrower, within the meaning of Section 4001(b) of ERISA.

"CREDIT AGREEMENT": See the Preamble hereto.

"CREDIT DOCUMENTS" means this Credit Agreement, the Notes (including any renewals, extensions, restatements and refundings thereof), any Guaranties and any written agreements, certificates or documents (and with respect to this Credit Agreement, the Notes and such other written agreements and documents, any amendments or supplements thereto or modifications thereof) executed or delivered pursuant to the terms of this Credit Agreement.

"DEFAULT RATE" means an interest rate per annum equal to three percent (3.0%) above the rate that would otherwise be payable under the terms of the Notes.

"DISBURSEMENT": See Section 2.13 hereof.

"DISBURSEMENT DATE": See Section 2.13 hereof.

"DOLLARS" and the sign \$\$ mean lawful currency of the United States of America.

"EBITDA" means Net Profit Before Tax, plus the sum of all interest expense and capitalized interest expensed in cost of sales, depreciation and amortization deducted in computing such Net Profit Before Tax, for the prior four quarters, determined on a consolidated basis.

"ENTITLED DEVELOPING LAND" means land:

- (a) that is zoned for the development of single family residences,
- (b) whose final or preliminary plat has been fully approved by the applicable municipality,
  - (c) that is in a Qualified Subdivision,
  - (d) that is under development, and
  - (e) that does not qualify as a Finished Lot.

"ENTITLED LAND" means land:

- (a) that is zoned for the development of single family residences,
- (b) whose final or preliminary plat has been fully approved by the applicable municipality,

-5-

- (c) that is in a Qualified Subdivision, and
- (d) that does not qualify as a Finished Lot or as  ${\tt Entitled}$   ${\tt Developing}$   ${\tt Land.}$

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, together with all final and permanent regulations issued pursuant thereto. References herein to sections and subsections of ERISA are deemed to refer to any successor or substitute provisions therefor.

"EVENT OF DEFAULT": See Section 9.1 hereof.

"EXCHANGE ACT" means the Securities Exchange Act of 1934, as amended.

"EXTENSION FEE" means \$15,000.00.

"FEDERAL FUNDS RATE" means, as of any date of determination, the rate set forth in the weekly statistical release designated as H.15(519), or any successor publication, published by the Federal Reserve Board (including any such successor, "H.15(519)") for such date opposite the caption "Federal Funds (Effective)." If for any relevant date such rate is not yet published in H.15(519), the rate for such date will be the rate set forth in the daily statistical release designated as the Composite 3:30 p.m. Quotations for U.S. Government Securities, or any successor publication, published by the Federal Reserve Bank of New York (including any such successor, the "Composite 3:30 p.m. Quotation") for such date under the caption "Federal Funds Effective Rate." If on any relevant date the appropriate rate for such date is not yet published in either H.15(519) or the Composite 3:30 p.m. Quotations, the rate for such date will be the arithmetic mean of the rates for the last transaction in overnight Federal funds arranged prior to 9:00 a.m. (New York City time) on that date by each of three leading brokers of Federal funds transactions in New York City selected by the Administrative Agent.

"FINANCIAL COVENANTS": See Section 8.10 hereof.

"FINISHED LOT" means land that has been subdivided and developed.

"FIXED CHARGE COVERAGE RATIO" means the ratio of the sum of the EBITDA of Borrower and its rental and operating lease expenses to its Fixed Charges, for its prior four fiscal quarters, determined on a consolidated basis.

"FIXED CHARGES" means the sum of the Consolidated Interest Incurred, rental and operating lease expenses, principal amortization payments and distributions by Borrower to its preferred shareholders, including dividends and distributions, but excluding any principal payments at the maturity of any loan and any principal amortization payments on the Senior Notes.

"GAAP" means those generally accepted accounting principles and practices which are recognized as such by the American Institute of Certified Public Accountants acting through its Accounting Principles Board or by the Financial

-6

Accounting Standards Board or through other appropriate boards or committees thereof and which are consistently applied for all periods after the date hereof so as to properly reflect the financial condition, and the results of operations

and changes in the financial position, of Borrower, including without limitation accounting rules promulgated pursuant to Regulations S-X and S-K, except that any accounting principle or practice required to be changed by the said Accounting Principles Board or Financial Accounting Standards Board (or other appropriate board or committee of the said Boards) in order to continue as a generally accepted accounting principle or practice may be so changed.

"GOVERNMENTAL AUTHORITY" means any government (or any political subdivision or jurisdiction thereof), court, bureau, agency or other governmental authority having jurisdiction over Borrower or any of its business, operations or properties.

"GUARANTIES," each a Guaranty, means a Continuing Guarantee substantially in the form attached hereto as Exhibit "E".

"GUARANTOR" means a Subsidiary of Borrower which has executed and delivered a Guaranty to the Administrative Agent.

"HILTON" means Steven J. Hilton.

"HONOR DATE" has the meaning specified in Section 2.18(b) hereof.

"INDEBTEDNESS" means, with respect to any Person, all of its monetary liabilities and any other liabilities required to be reported pursuant to GAAP, including without limitation each of the following (without duplication): (a) obligations of that Person to any other Person for payment of borrowed money, (b) capital lease obligations, (c) notes and drafts drawn or accepted by that Person payable to any other Person, whether or not representing obligations for borrowed money (but without duplication of indebtedness for borrowed money), (d) any obligation for the purchase price of property the payment of which is deferred for more than one year or evidenced by a note or equivalent instrument, or for which the sales agreement provides that, in the event the purchase is not consummated, liquidated damages in excess of twenty percent (20.0%) of the purchase price are to be paid to the seller, (e) guarantees of Indebtedness of third parties, and (f) a recourse or nonrecourse payment obligation of any other Person that is secured by a Lien on any property of the first Person, whether or not assumed by the first Person, up to the fair market value (from time to time) of such property (absent manifest evidence to the contrary, the fair market value of such property shall be the amount determined under GAAP for financial reporting purposes), but excluding any trades accounts payables of less than sixty (60) days, any accruals incurred in the ordinary course of business and any off-site bonds provided by a surety and any guaranties related thereto.

"INSOLVENCY PROCEEDING" means any proceeding undertaken under the Debtor Relief Laws.

-7-

"INTEREST COVERAGE RATIO" means the ratio of the EBITDA of Borrower to its Consolidated Interest Incurred, for its prior four fiscal quarters, determined on a consolidated basis.

"INVENTORY" means all real property fully owned by Borrower and all real property fully owned by its Subsidiaries which are Guarantors.

"INVENTORY INSPECTION" means the inspection  $% \left( 1\right) =1$  performed in accordance with Section 7.18 hereof.

"ISSUANCE DATE" means the date on which a Letter of Credit is delivered to the beneficiary thereof.

"ISSUANCE REQUEST" means a request for a Letter of Credit duly executed by Borrower in a form satisfactory to the Issuing Bank.

"ISSUE" means, with respect to any Letter of Credit, to issue or, by amendment or otherwise, to extend the expiry of, or to renew or increase or decrease the amount of, such Letter of Credit; and the terms "ISSUED," "ISSUING" and "ISSUANCE" have corresponding meanings.

"ISSUING BANK" means CB&T in its capacity as issuer of one or more Letters of Credit hereunder, together with any successor or replacement Letter of Credit issuer arising under this Credit Agreement.

"LANDON" means John R. Landon.

"LC BORROWING" means an extension of credit resulting from a drawing under any Letter of Credit which shall not have been reimbursed on the date when made nor converted into a Variable Rate Advance.

"LC OBLIGATIONS" means at any time the sum of (a) the Outstanding LC Balance under the RLC, plus (b) the amount of all unreimbursed drawings under all Letters of Credit, including all outstanding LC Borrowings.

"LETTER OF CREDIT" means a letter of credit, either as a standby financial or a performance letter of credit, issued by the Issuing Bank for the account of Borrower pursuant to Article 2 hereof.

"LETTER OF CREDIT FEE" means a fee to be determined by the Issuing Bank.

"LEVERAGE RATIO" means the ratio of (i) the Consolidated Indebtedness of Borrower to (ii) its Consolidated Tangible Net Worth.

"LIEN" means any lien, mortgage, security interest, tax lien, pledge, encumbrance, conditional sale or title retention arrangement, or any other interest in property designed to secure the repayment of Indebtedness whether arising by agreement or under any statute or law, or otherwise.

-8-

"LOANS" means the RLC.

"LOAN FEES" means the fees described in Section 3.2 hereof.

"MAJORITY BANKS" means, at any time, Banks having Pro Rata Shares representing at least 51% of the aggregate Commitment.

"MATERIAL ADVERSE EFFECT" means any circumstance or event which (i) has any material adverse effect upon the validity or enforceability of any Credit Document, (ii) materially impairs the ability of Borrower to fulfill its obligations under the Credit Documents, or (iii) causes an Event of Default or any event which, with notice or lapse of time or both, would become an Event of Default.

"MATURITY DATE" means the RLC Maturity Date.

"MAXIMUM LC COMMITMENT" means \$0.

"MODEL HOME" means a Unit constructed initially for inspection by prospective purchasers that is not intended to be sold until all or substantially all other Units in the Subdivision are sold.

"MORTGAGE LIENS" means a Lien on real property for the benefit of a lender to secure a monetary obligation owed to said lender.

"NET PROFIT" means for any period the post-income tax net income of Borrower for such period in accordance with GAAP, determined on a consolidated basis.

"NET PROFIT BEFORE TAX" means for any period the pre-income tax net income of Borrower for such period in accordance with GAAP, determined on a consolidated basis.

"NET UNIT SALES" means the number of Units sold by Borrower pursuant to signed purchase contracts on a consolidated basis during a fiscal quarter, less any cancellations during such period of any purchase contract signed either in said period or in prior periods.

"NON-RECOURSE  $\,$  DEBT" means Indebtedness that is non-recourse to Borrower or any Guarantor.

"NONENTITLED LAND" means land that does not qualify as a Finished Lot, as Entitled Land or as Entitled Developing Land.

"NOTE" means an RLC Note.

-9-

"OBLIGATIONS" means all present and future indebtedness, obligations and liabilities of Borrower to the Banks, and all renewals and extensions thereof, or any part thereof, arising pursuant to this Credit Agreement or represented by the Notes, including without limitation the Loans and all interest accruing thereon, and attorneys' fees incurred in the enforcement or collection thereof, regardless of whether such indebtedness, obligations and liabilities are direct, indirect, fixed, contingent, joint, several or joint and several; together with all indebtedness, obligations and liabilities of Borrower evidenced or arising pursuant to any of the other Credit Documents, and all renewals and extensions thereof, or part thereof.

"OBLIGOR" means an owner of RLC Borrowing Base Inventory.

"OUTSTANDING LC BALANCE" in effect at any time means the maximum aggregate amount available to be drawn at such time under all outstanding Letters of Credit, the determination of such maximum amount to assume compliance with all conditions for a Disbursement.

"OUTSTANDING PERIOD": See Section 2.12(a) hereof.

"PAYMENT DATE" means as to interest the first day of each calendar month after the Closing Date, provided, however, that at Lender's option, the first Payment Date may be the first day of the second calendar month after the Closing Date.

"PAYMENT OFFICE" means Central Note Department, 9775 Clairemont Mesa

Boulevard, San Diego, California 92124.

"PBGC" means the Pension Benefit Guaranty Corporation, and any successor to all or substantially all of the Pension Benefit Guaranty Corporation's functions under ERISA.

"PERMITTED LIENS" means Liens which consist of the following:

- (a) Liens for taxes, assessments or governmental and special district charges not yet due and payable;
- (b) Liens to which the Banks shall consent in writing, in their sole and absolute discretion;
  - (c) lot premiums payable to the seller of any land; and
- (d) performance deeds of trust and other Liens payable to the seller of any land.

"PERSON" includes an individual, a corporation, a joint venture, a partnership, a trust, a limited liability company, an unincorporated organization or a government or any agency or political subdivision thereof.

-10-

"PLAN" means an employee defined benefit plan or other plan maintained by Borrower for employees of Borrower and covered by Title IV of ERISA, or subject to the minimum funding standards under Section 412 of the Code.

"PRESOLD HOME" means a Unit (i) for which a bona fide, non-contingent purchase contract has been signed or (ii) for which a contingent contract has been signed within the past sixty (60) days, in either case for which an earnest money deposit has been made in an amount not less than two percent (2.0%) of the deposit.

"PRO RATA SHARE" with respect to any individual Bank, or Pro Rata Shares with respect to all of the Banks, as the case may be, means the applicable percentage or percentages of the Commitment assigned to each of the Banks as set forth on Schedule 2.1 hereto or in the Co-Lender Agreement, as applicable.

"QUALIFIED INVENTORY" means Inventory that:

- (a) Is located in a Qualified State;
- (b) Is located in a Qualified Subdivision;
- (c) Is owned by Borrower or a Guarantor, and is free of any Liens.
- (e) Is free of any Mortgage Liens.

"QUALIFIED STATES" means Arizona, California and Texas.

"QUALIFIED SUBDIVISION" means a Subdivision for which the Administrative Agent has received a fully executed Subdivision Admission Certificate substantially in the form of Exhibit "D" attached hereto, signed by an Authorized Officer, certifying that it has completed its due diligence and that the Obligor has the following items:

- (i) A Qualified Title Policy together with evidence that fee simple title to the Subdivision is held by the Obligor or in a subdivision trust for the benefit of Obligor insured by the Qualified Title Policy, subject only to commercially acceptable exceptions.
- (ii) A Phase I Environmental report requiring no further environmental investigation, or a completed environmental questionnaire that concludes that environmental contamination is not probable.
- (iii) A commercially acceptable soils and engineering report, indicating that the soil is capable of supporting the purported development.

-11-

"QUALIFIED TITLE POLICY" means an ALTA owner's title insurance policy, subject only to commercially acceptable exceptions from an insurance company reasonably acceptable to the Banks.

"REFERENCE RATE" means the interest rate per annum designated by California Bank & Trust, a California banking corporation, or its successors, as its "prime rate" as publicly announced by that bank from time to time as a means of pricing credit extensions to some customers and is neither tied to any extended interest rate or index nor necessarily the lowest rate of interest charged by that bank at any given time for any particular class of customer or credit extension.

"REGULATION U" means Regulation U promulgated by the Board of Governors of the Federal Reserve System, 12 C.F.R. Part 221, or any other regulation hereafter promulgated by said Board to replace the prior Regulation U and having substantially the same function.

"REGULATORY CHANGE" means any change effective after the date of this Credit Agreement in United States federal, state, or foreign law, regulations, or rules or the adoption or making after such date of any interpretation, directive, or request applying to a class of banks including the Banks, of or under any United States federal, state, or foreign law, regulation or rule (whether or not having the force of law) by any court or governmental or monetary authority charged with the interpretation or administration thereof.

"REPORTABLE EVENT" means any "reportable event" as described in Section 4043(b) of ERISA with respect to which the thirty (30) day notice requirement has not been waived by the PBGC.

"REQUIRED BANKS" means, at any time, Banks having Pro Rata Shares representing at least 66 2/3% of the aggregate Commitment.

"RLC": See Recital A hereto.

"RLC ADVANCE" means a disbursement of the proceeds of the RLC.

"RLC BALANCE" means the sum of (i) with respect to the RLC on any date, the aggregate outstanding principal amount thereof, after giving effect to any borrowings and prepayments or repayments of RLC Advances occurring on such date; plus (ii) with respect to any outstanding LC Obligations on any date, the aggregate amount of such LC Obligations on such date, after giving effect to any Issuances of Letters of Credit occurring on such date and any other changes in the aggregate amount of the LC Obligations as of such date, including changes occurring as a result of any reimbursements of outstanding unpaid drawings under any Letters of Credit or any reductions in the maximum amount available for drawing under Letters of Credit taking effect on such date.

"RLC BORROWING AVAILABILITY" means (i) an amount equal to the lesser of the RLC Commitment or the RLC Borrowing Base as calculated each month, less (ii) an amount equal to the sum of the RLC Balance, Senior Unsecured Debt (other than

-12-

the Senior Notes) and Subordinated Debt to the extent it exceeds in the aggregate \$20,000,000.00.

"RLC BORROWING BASE" means the Dollar sum of the value of the following Inventory (the "RLC Borrowing Base Inventory") as calculated each month on the RLC Borrowing Base Inventory Report:

- A. Entitled Land that has been owned for more than ninety (90) days, but less than twelve (12) months, valued at thirty percent (30.0%) of its costs determined on the basis of GAAP.
- B. Entitled Land that has been owned for ninety (90) days or less valued at sixty percent (60.0%) of its costs determined on the basis of
- C. Entitled Developing Land valued at sixty percent (60.0%) of its costs determined on the basis of GAAP.
- D. Finished Lots valued at sixty percent (60.0%) of their costs determined on the basis of GAAP.
- E. Pre-Sold Homes valued at ninety percent (90.0%) of its costs determined on the basis of GAAP.
- F. Model Homes valued at eighty percent (80.0%) of their costs determined on the basis of GAAP.
- G. Spec Homes valued at the following percent of their costs determined on the basis of GAAP:
  - (i) During construction and until twelve (12) months after completion of construction, 80.0%.
  - (ii) After twelve (12) months after its completion, but until eighteen (18) months after its completion, 70.0%.
  - (iii) After eighteen (18) months after its completion, but until twenty-four (24) months after its completion, 50.0%.
    - (iv) After twenty-four (24) months after its completion, 0.0%.

subject to the following conditions:

(a) All property included in the RLC Borrowing Base Inventory must be Qualified Inventory.

(b) There may not be more than \$8,000,000 of Entitled Developing Land

costs determined on the basis of GAAP per Subdivision.

- (c) Not more than six (6) Model Homes located in any single Subdivision of a single product type may be included in the RLC Borrowing Base Inventory.
- (d) The aggregate value of Spec Homes in the RLC Borrowing Base Inventory may not exceed the lesser of \$20,000,000 or thirty percent (30.0\$) of the sale value of the average backlog at the end of the prior four quarters.
- (e) Not more than fifteen (15) Spec Homes in California and twelve (12) elsewhere located in any single Subdivision of a single product type may be included in the RLC Borrowing Base Inventory.

"RLC BORROWING BASE COMPLIANCE CERTIFICATE": See Section 7.1(f) hereof.

"RLC BORROWING BASE INVENTORY": See the definition of RLC Borrowing Base.

"RLC BORROWING BASE INVENTORY REPORT": See Section 7.1(f) hereof.

"RLC COMMITMENT" means FIFTEEN MILLION AND NO/100 DOLLARS (\$15,000,000.00).

"RLC COMMITMENT FEE": See Section 3.2(a) hereof.

"RLC MATURITY DATE" means December 17, 1999, unless extended pursuant to Section 4.1.

"RLC NOTE" means a Revolving Promissory Note of even date herewith substantially in the form attached hereto as Exhibit "C", in the amount of a Bank's Pro Rata Share of the RLC Commitment, executed by Borrower and delivered to a Bank pursuant to the terms of this Credit Agreement, together with any renewals, extensions, modifications, restatements or replacements thereof.

"SEC" means the Securities and Exchange Commission.

"SENIOR NOTES" means those \$15,000,000 9.10% Senior Notes due September 1, 2005 issued by Borrower pursuant to its Note Agreement dated as of September 1, 1998.

"SENIOR UNSECURED DEBT" means Indebtedness of Borrower whose payment is not subordinated in writing to the payment of the Obligations to the satisfaction of the Banks and which is not secured by a Lien on any property of Borrower, other than the Senior Notes.

"SIGNIFICANT DEBT AGREEMENT" means all documents, instruments and agreements executed by Borrower, evidencing, securing or ensuring any Indebtedness of Borrower or any guaranty in excess of \$1,000,000 in outstanding

-14-

principal (or principal equivalent) amount, excluding any Indebtedness that is Non-Recourse Debt.

"SPEC HOME" means a Unit immediately available for sale, but for which a purchase contract reasonably satisfactory to the Administrative Agent has not been signed.

"STATED  $\,$  AMOUNT" of a Letter of Credit means the stated amount as stated in the Letter of Credit.

"STATED EXPIRY DATE" of a Letter of Credit means the Stated Expiry Date as stated in the Letter of Credit.

"SUBDIVISION" means a group of Finished Lots being sold or to be sold from a single complex of Model Homes of a single product type.

"SUBORDINATED DEBT" means Unsecured Debt of Borrower whose payment is subordinated in writing to the payment of the Loans and the other obligations of Borrower under this Credit Agreement, to the satisfaction of the Banks.

"SUBSIDIARIES" means all business associations the majority of whose equity interests is owned by Borrower or that are controlled by Borrower.

"TAX ASSET" means prior losses  $% \left( 1\right) =\left( 1\right) +\left( 1\right) =\left( 1\right) +\left( 1\right) +\left( 1\right) =\left( 1\right) +\left( 1\right) +\left($ 

"UNIT" means a single family residence.

"UNSECURED DEBT" means  $\,$  Indebtedness of a Person which is not secured by a Lien on any of its property.

"VARIABLE RATE" means the rate per annum equal to the Reference Rate per annum as in effect from time to time. The Variable Rate will change on each day that the "Reference Rate" changes.

"VARIABLE RATE ADVANCE" means an Advance that bears, or is requested to bear, interest at the Variable Rate.

"VARIABLE RATE RLC ADVANCE" means an RLC Advance that bears or that is requested to bear interest at the Variable Rate.

#### 1.2 TERMS GENERALLY.

(a) The definitions in Section 1.1 shall apply equally to both the singular and plural forms of the terms defined.

-15-

- (b) Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms.
- (c) All references herein to Articles, Sections, Exhibits and Schedules shall be deemed references to Articles and Sections of, and Exhibits and Schedules to, this Agreement unless the context shall otherwise require.
- (d) Except as otherwise expressly provided herein, all terms of an accounting or financial nature shall be construed in accordance with GAAP, as in effect from time to time.

-16-ARTICLE 2.

THE RLC

2.1 RLC COMMITMENT. Each Bank agrees, severally but not jointly, to loan to or for the benefit of Borrower, and Borrower shall be entitled to draw upon and borrow, in the manner and upon the terms and conditions contained in this Credit Agreement, an amount that shall not exceed that Bank's Pro Rata Share of the RLC Commitment.

#### 2.2 REVOLVING LINE.

- (a) Subject to the terms and conditions set forth in this Credit Agreement, each Bank shall provide to Borrower a revolving line of credit (each, a "RLC"), against which a Bank shall fund its Pro Rata Share of each RLC Advance to be made to Borrower, repaid by Borrower, and readvanced to Borrower, as Borrower may request, and the Issuing Bank shall issue such Letters of Credit as Borrower shall request, which may be terminated or repaid by Borrower and reissued, provided that (i) there is no Event of Default under any provision of this Credit Agreement, (ii) no RLC Advance shall be made or Letter of Credit issued that would exceed the RLC Borrowing Availability, (iii) no Bank shall be obligated under any circumstances to fund an RLC Advance in excess of that Bank's Pro Rata Share of the requested RLC Advance, (iv) the aggregate amount of a Bank's funding of the RLC Balance at any one time outstanding shall not exceed its Pro Rata Share of the RLC Commitment, and (v) no Letter of Credit shall be issued with a Stated Expiry Date later than the RLC Maturity Date. The Banks shall not be obligated to fund their Pro Rata Share of any RLC Advance if, after giving effect thereto, any of the foregoing limitations would be exceeded.
- (b) The failure of any Bank to fund its Pro Rata Share of an RLC Advance in accordance with its Pro Rata Share of the RLC Commitment shall not relieve any other Bank of its several obligations hereunder, but no Bank shall be liable with respect to the obligation of any other Bank hereunder.
- (c) RLC Advances may be made for the purpose of providing working capital financing.
- 2.3 RLC NOTES. The RLC of each Bank shall be evidenced by an RLC Note and shall bear interest and be payable to the order of such Bank upon the terms and conditions contained therein. The aggregate amount funded by a Bank under its RLC Note less all repayments of principal thereof shall be the principal amount owing and unpaid on its RLC Note and its RLC. The principal amount funded by a Bank and all principal payments and prepayments thereof may be noted by such Bank on a schedule attached to its RLC Note and shall be entered by the Bank on its ledgers and computer records; provided that the failure of the Bank to make such notations or entries shall not affect the principal amount owing and unpaid on its RLC Note. The entries made in the ordinary course of business by a Bank on its ledgers and computer records and any notations made in the ordinary

-17-

course of business by a Bank on any such schedule annexed to its RLC Note shall be presumed to be accurate until the contrary is established. If requested, Borrower shall confirm in writing to the Administrative Agent each RLC Advance.

2.4 RLC. The RLC shall bear interest and be payable to the Banks upon the terms and conditions contained therein, which include the following provisions:

- (a) Interest shall accrue:
- (i) On the unpaid  $\mbox{principal}$  of each RLC Advance at the Variable Rate.
  - (ii) [Intentionally left blank.]
- (b) All interest shall be computed on the basis of a 360-day year and accrue on a daily basis for the actual number of days elapsed. All accrued interest shall be due and payable on each Payment Date.
- (c) The entire unpaid principal balance, all accrued and unpaid interest, and all other amounts payable under the RLC Note shall be due and payable in full on the RLC Maturity Date.
- (d) Each request for an RLC Advance shall, in addition to complying with the other requirements in this Credit Agreement, (i) specify the date and amount of the requested RLC Advance, (ii) be delivered to the Administrative Agent before 9:00 a.m. (San Diego, California local time) at least one Business Day prior to the date of the requested RLC Advance and (iii) be in a minimum amount of \$250,000.00. No more than one RLC Advance per Bank shall be made in any single week.
- (e) After receiving a request for an RLC Advance in the manner provided herein, the Administrative Agent shall promptly, before 11:30 a.m. (San Diego, California local time) on the date an RLC Advance is requested, notify each Bank by telephone (confirmed promptly in writing), telefacsimile or cable of the terms of such request and such Bank's Pro Rata Share of the requested RLC Advance. Each Bank shall, before 1:00 p.m. (San Diego, California local time) on the date an RLC Advance is to be made as specified in a request for an RLC Advance, deposit with the Administrative Agent such Bank's Pro Rata Share of the requested RLC Advance in immediately available funds. Upon fulfillment of all applicable conditions set forth herein and after receipt by the Administrative Agent of such funds, the Administrative Agent shall pay or deliver all funds so received to the order of Borrower at the principal office of the Administrative Agent. The failure of any Bank to fund its Pro Rata Share of any RLC Advance required of it hereunder shall not relieve any other Bank of its obligation to fund its Pro Rata Share of any RLC Advance hereunder. If any Bank fails to fund its Pro Rata Share of the requested RLC Advance

-18-

and if all conditions to such RLC Advance have apparently been satisfied, the Administrative Agent will make available to Borrower the funds received by it from the other Banks. Neither the Administrative Agent nor any Bank shall be responsible for the performance by any other Bank of its obligations hereunder.

Unless the Administrative Agent shall have received notice from a Bank prior to the date of any RLC Advance that such Bank will not make available to the Administrative Agent such Bank's Pro Rata Share of the requested RLC Advance, the Administrative Agent may assume that such Bank has made such amount available to the Administrative Agent on the date of such RLC Advance in accordance with this Section and the Administrative Agent may, in reliance upon such assumption, make available a corresponding amount to or on behalf of Borrower on such date. If and to the extent any Bank shall not have so made its Pro Rata Share of the requested RLC Advance available to the Administrative Agent (the "Principal Shortfall Amount"), Borrower agrees to repay the Principal Shortfall Amount to the Administrative Agent forthwith on demand, together with interest thereon for each day from (and including) the date such amount is made available to or on behalf of Borrower to (but excluding) the date such amount is repaid to the Administrative Agent, at the rate per annum equal to the rate otherwise applicable to the RLC Advance in question.

- (f) [Intentionally left blank.]
- (g) [Intentionally left blank.]
- (h) Each request for an RLC Advance shall be irrevocable and binding on Borrower once the request is received by the Administrative Agent and the Administrative Agent notifies the Banks of the request. Prior to the Administrative Agent's notice of the request to the Banks, Borrower may revoke the request. Borrower shall indemnify each Bank against any cost, loss or expense incurred by any Bank as a result of Borrower's failure to fulfill, on or before the date specified for an RLC Advance in any request for an RLC Advance, the conditions to such RLC Advance set forth herein, including any cost, loss or expense incurred by reason of the liquidation or reemployment of deposits or other funds acquired by a Bank to fund such RLC Advance when such RLC Advance, as a result of such failure, is not made on the date so specified.
  - (i) [Intentionally left blank.]
  - (j) Nothing herein shall be deemed to relieve any Bank from its

obligation to fulfill its Pro Rata Share of the RLC Commitment hereunder or to prejudice any right which the Administrative Agent or the Borrower may have against any Bank as a result of any default by such Bank hereunder.

-19-

- (k) Upon an Event of Default, including failure to pay upon final maturity, the Banks, at their option, may also, if permitted under applicable law, do one or both of the following: (a) increase the applicable interest rate to the Default Rate, and/or (b) add any unpaid accrued interest to principal and such sum will bear interest therefrom until paid at the rate provided herein (including any increased rate). The interest rate shall not exceed the maximum rate permitted by applicable law.
- 2.5 EXCESS BALANCE REPAYMENT. In the event that the RLC Borrowing Base Availability has a negative balance at the end of any month, there shall be due and payable from Borrower to the Banks, and Borrower shall immediately repay to the Banks, without notice or demand, from time to time, an amount equal to the lesser of the negative balance of the RLC Borrowing Base Availability or the RLC Balance.
- 2.6 REDUCTION OF RLC COMMITMENT. Borrower shall have the right at any time upon at least seven days' prior written notice to the Administrative Agent to reduce the aggregate amount of the RLC Commitment; PROVIDED, that the amount of each such reduction shall be in a minimum aggregate amount of \$1,000,000.00 and an integral aggregate multiple of \$100,000.00 in excess thereof and that no such reduction shall reduce (i) the amount of the RLC Commitment to less than the RLC Balance, or (ii) the amount of a Bank's Pro Rata Share of the RLC Commitment to less than the amount of the RLC Balance funded by such Bank. Any reduction in the aggregate amount of the RLC Commitment shall reduce each Bank's share of the RLC Commitment by its Pro Rata Share of the aggregate amount of such reduction. The Administrative Agent shall promptly notify each Bank of any such notice of reduction received from the Borrower. Any reduction in the RLC Commitment may not be reinstated without the mutual prior consent of the Borrower and the Banks.
- 2.7 CONDITIONS. The Banks shall have no obligation to fund their Pro Rata Shares of any RLC Advance unless and until all of the conditions and requirements of this Credit Agreement are fully satisfied. However, the Banks in their sole and absolute discretion may elect to make one or more RLC Advances prior to full satisfaction of one or more such conditions and/or requirements. Notwithstanding that such an RLC Advance or RLC Advances are made, such unsatisfied conditions and/or requirements shall not be waived or released thereby. Borrower shall be and continue to be obligated to fully satisfy such conditions and requirements, and the Banks, at any time, in their sole and absolute discretion, may stop making RLC Advances until all conditions and requirements are fully satisfied.
- 2.8 OTHER RLC ADVANCES. The Administrative Agent, at the direction of the Banks, after giving written notice to Borrower, after the occurrence of an Event of Default and during the continuation thereof, may make RLC Advances in any amount in payment of (i) insurance premiums, taxes, assessments, liens or encumbrances existing against Borrower's property, other than Permitted Liens, (ii) interest accrued and payable upon the RLC, (iii) any indebtedness, charges and expenses that are the obligation of Borrower under this Credit Agreement, and (iv) any charges or matters necessary to cure any Event of Default.
- 2.9 ASSIGNMENT. Borrower shall have no right to any RLC Advance other than to have the same disbursed by the Administrative Agent in accordance with the disbursement provisions contained in this Credit Agreement. Any assignment or

-20-

transfer, voluntary or involuntary, of this Credit Agreement or any right hereunder shall not be binding upon or in any way affect the Banks without their written consent; the Administrative Agent, at the direction of the Banks may make RLC Advances under the disbursement provisions herein, notwithstanding any such assignment or transfer.

#### 2.10 ISSUANCE OF LETTERS OF CREDIT.

- (a) Subject to the terms and conditions of this Credit Agreement, (i) the Issuing Bank agrees from time to time before the RLC Maturity Date to issue Letters of Credit for the account of the Borrower; and (ii) the Banks severally agree to participate in Letters of Credit issued for the account of the Borrower, subject to the prior approval by each Bank of the provisions of each Letter of Credit. Each reference in this Credit Agreement to the "issue" or "issuance" or other forms of such words in relation to Letters of Credit shall be deemed to include any extension or renewal of a Letter of Credit.
- (b) Each Letter of Credit shall (i) by its terms be issued in a Stated Amount; (ii) have a Stated Expiry Date no later than the RLC Maturity Date; (iii) expire or be terminated by the beneficiary thereunder on or before its Stated Expiry Date; (iv) not exceed the RLC Borrowing Base Availability; and (v) not cause the Outstanding LC Balance after the

issuance of said Letter of Credit to exceed the Maximum LC Commitment.

- (c) In addition to the conditions otherwise specified in this Section, the obligation of the Issuing Bank to issue a Letter of Credit shall be subject to the further condition precedent that the following statements shall be correct, and each application for such Letter of Credit and the issuance of such Letter of Credit shall constitute a representation and warranty by Borrower that on the date of the issuance of such Letter of Credit such statements are correct:
  - (i) The representations and warranties in Article 6 are correct on and as of the date of the issuance of such Letter of Credit, before and after giving effect to such issuance, as though made on and as of such date;
    - (ii) No Event of Default has occurred and is continuing; and
  - (iii) The conditions in Section 2.2(a) are satisfied as of the date of issuance of the Letter of Credit, before and after giving effect to such issuance.
- 2.11 ISSUANCE PROCEDURE FOR LETTERS OF CREDIT. By delivery to the Issuing Bank of an Issuance Request on or before 9:00 a.m. (San Diego, California time) five (5) Business Days prior to the requested Issuance Date, and the execution

-21-

of such applications and agreements as the Issuing Bank may reasonably request, Borrower may request the issuance of a Letter of Credit in such form as Borrower may reasonably request. Each Issuance Request shall include the form of the Letter of Credit, the amount and other terms thereof. Subject to the terms and conditions of this Credit Agreement, the Issuing Bank will issue such Letter of Credit on the Issuance Date specified in the Issuance Request submitted in connection therewith. The Issuing Bank and Borrower agree that all Letters of Credit issued pursuant to the terms of this Article shall be subject to the terms and conditions and entitled to the benefits of this Credit Agreement and the other Credit Documents.

- 2.12 LETTER OF CREDIT FEES AND COSTS.
- (a) Borrower agrees to pay to the Administrative Agent, for distribution to the Banks pursuant to Section 9A.8 hereof, a non-refundable fee equal to the Letter of Credit Fee per annum on the Stated Amount of each Letter of Credit, computed on a daily basis, from and including the Issuance Date of such Letter of Credit to the Stated Expiry Date (the "Outstanding Period"). The Letter of Credit Fee shall be payable in advance upon the issuance of a Letter of Credit. Upon an Event of Default, the Letter of Credit Fee shall be equal to three hundred basis points (3.0%) above the Letter of Credit Fee.
- (b) Borrower further agrees to pay to the Issuing Bank for its account a charge for all reasonable administrative expenses of the Issuing Bank in connection with the issuance, amendment or modification (if any) and administration of the Letter of Credit upon demand from time to time.
- 2.13 DISBURSEMENTS. The Issuing Bank will notify Borrower of the presentment for payment of a Letter of Credit by any beneficiary thereto, together with notice of the date (the "Disbursement Date") such payment shall be made. Subject to the terms and provisions of the Letter of Credit, the Issuing Bank shall make such payment (a "Disbursement") to the beneficiary of the Letter of Credit. Each such Disbursement shall be deemed to be an RLC Advance hereunder.
- 2.14 REIMBURSEMENT OBLIGATIONS OF BORROWER. Borrower's obligation under Section 2.13 to reimburse the Banks with respect to each Disbursement (including interest thereon) in respect of any Letter of Credit shall be absolute and unconditional under any and all circumstances and irrespective of any setoff, counterclaim, or defense to payment which Borrower may have or have had against the Banks, the Issuing Bank, the Administrative Agent or the beneficiary thereof, including any defense based upon the occurrence of any Event of Default, any draft, demand or certificate or other document presented under the Letter of Credit proving to be forged, fraudulent, invalid or insufficient, the failure of any Disbursement to conform to the terms of the Letter of Credit (if, in Issuing Bank's good faith opinion, such Disbursement is determined to be appropriate) or any non-application or misapplication by the beneficiary of the proceeds of such Disbursement, or the legality, validity, form, regularity or enforceability of the Letter of Credit; PROVIDED, HOWEVER, that nothing herein shall adversely affect the right of Borrower to commence any proceeding against Issuing Bank for any wrongful Disbursement made by Issuing Bank under the Letter of Credit as a result of acts or omissions constituting gross negligence or wilful misconduct on the part of Issuing Bank.

2.15 NATURE OF REIMBURSEMENT OBLIGATIONS. Borrower shall assume all risks of the acts, omissions or misuse of any Letter of Credit by the beneficiary thereof. Neither the Banks nor the Issuing Bank (except to the extent of its own

- (a) the form, validity, sufficiency, accuracy, genuineness or legal effect of any Letter of Credit or any document submitted by any party in connection with the issuance of any Letter of Credit, even if such document should in fact prove to be in any or all respects invalid, insufficient, inaccurate, fraudulent or forged;
- (b) the form, validity, sufficiency, accuracy, genuineness or legal effect of any instrument transferring or assigning or purporting to transfer or assign any Letter of Credit;
- (c) failure of any beneficiary of any Letter of Credit to comply fully with conditions required in order to demand payment under a Letter of Credit;
- (d) errors, omissions, interruption or delays in transmission or delivery of any messages, by mail, cable, telegraph, telex or otherwise; or
- (e) any loss or delay in the transmission or otherwise of any document or draft required by or from a beneficiary of a Letter of Credit in order to make a Disbursement under a Letter of Credit or of the proceeds thereof.

None of the foregoing shall affect, impair or prevent the vesting of any of the rights or powers granted the Banks or the Issuing Bank hereunder. In furtherance and extension, and not in limitation or derogation of any of the foregoing, any action taken or omitted to be taken by the Banks or the Issuing Bank in good faith shall be binding upon the Borrower and shall not put the Banks or the Issuing Bank under any resulting liability to Borrower.

- 2.16 BANKS OBLIGATION. Nothing herein shall be deemed to relieve any Bank from its obligations to fulfill its Pro Rata Share of the RLC Commitment hereunder or to prejudice any right which the Administrative Agent or the Borrower may have against any Bank as a result of any default by such Bank hereunder.
- 2.17 CERTAIN REQUIREMENTS AS TO LETTERS OF CREDIT. The Issuing Bank is under no obligation to Issue any Letter of Credit if:
  - (i) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain the Issuing Bank from Issuing such Letter of Credit, or any requirement of law applicable to the Issuing Bank or any request or directive (with

-23-

which it is customary for banks in the relevant jurisdiction to comply whether or not having the force of law) from any Governmental Authority with jurisdiction over the Issuing Bank shall prohibit, or request that the Issuing Bank refrain from, the Issuance of letters of credit generally or such Letter of Credit in particular or shall impose upon the Issuing Bank with respect to such Letter of Credit any restriction, reserve, or capital requirement (for which the Issuing Bank is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon the Issuing Bank any unreimbursed loss, cost, or expense which was not applicable on the Closing Date and which the Issuing Bank in good faith deems material to it;

- (ii) the Issuing Bank has received written notice from any Bank, the Administrative Agent or Borrower, on or prior to the Business Day prior to the requested date of Issuance of such Letter of Credit, that one or more of the applicable conditions contained in Article 5 is not then satisfied;
- (iii) the Stated Expiry Date of any requested Letter of Credit is not in accord with the requirements of Section 2.10(b), unless all of the Banks have approved such Stated Expiry Date;
- (iv) any requested Letter of Credit does not provide for drafts, or is not otherwise in form and substance acceptable to the Issuing Bank, or the Issuance of a Letter of Credit shall violate any applicable policies of the Issuing Bank; or
- (v) such Letter of Credit is to be used for a purpose  $\,$  other than as provided herein or denominated in a currency other than Dollars.
- 2.18 RISK PARTICIPATIONS, DRAWINGS, AND REIMBURSEMENTS.
- (a) Immediately upon the Issuance of each Letter of Credit, each Bank shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Issuing Bank a participation in such Letter of Credit and each drawing thereunder in an amount equal to the product of (i) the Pro Rata Share of such Bank, times (ii) the maximum amount available to be drawn under such Letter of Credit and the amount of any drawing, respectively. For purposes of the RLC Commitment, each Issuance of a Letter of Credit shall be deemed to utilize each Bank's Pro Rata Share of the RLC

-21-

- (b) In the event of any request for a drawing under a Letter of Credit by the beneficiary or transferee thereof, the Issuing Bank will promptly notify the Borrower. The Issuing Bank shall honor any Disbursement request under any Letter of Credit only if (i) such request is delivered to the Issuing Bank by the beneficiary of such Letter of Credit, and (ii) such request is accompanied by the original documents, if any, required by the Letter of Credit for any Disbursement. Except as otherwise provided herein, the Borrower shall reimburse the Issuing Bank prior to 11:00 a.m. (San Diego, California local time) on each date that any amount is paid by the Issuing Bank under any Letter of Credit (each such date, an "Honor Date"), in an amount equal to the amount so paid by the Issuing Bank. In the event the Borrower is required but fails to reimburse the Issuing Bank for the full amount of any drawing under any Letter of Credit by 11:00 a.m. (San Diego, California local time) on the Honor Date, the Issuing Bank will promptly notify the Administrative Agent and the Administrative Agent will promptly notify each Bank thereof. Any notice given by the Issuing Bank or the Administrative Agent pursuant to this Section may be oral if immediately confirmed in writing (including by facsimile); provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice.
- (c) Each Bank shall upon any notice pursuant to this Section make available to the Administrative Agent for the account of the Issuing Bank an amount in Dollars and in immediately available funds equal to its Pro Rata Share of the amount of the drawing, whereupon the Banks shall (subject to paragraph (d)) each be deemed to have made a Variable Rate RLC Advance to the Borrower in that amount. If any Bank so notified fails to make available to the Administrative Agent for the account of the Issuing Bank the amount of such Bank's Pro Rata Share of the amount of the drawing by no later than 3:00 p.m. (San Diego, California local time) on the Honor Date, then interest shall accrue on such Bank's obligation to make such payment, from the Honor Date to the date such Bank makes such payment, at a rate per annum equal to the Federal Funds Rate in effect from time to time during such period and such amount and interest shall be immediately due and payable to the Administrative Agent; the obligation of such Bank to make such payment to the Administrative Agent shall not be waived by the Administrative Agent without the prior written consent of the Borrower. The Administrative Agent will promptly give notice of the occurrence of the Honor Date, but failure of the Administrative Agent to give any such notice on the Honor Date or in sufficient time to enable any Bank to effect such payment on such date shall not relieve such Bank from its obligations under this Section.
- (d) With respect to any unreimbursed drawing, the Borrower shall be deemed to have incurred from the Issuing Bank a Variable Rate RLC Advance in the amount of such drawing.
- (e) Each Bank's obligation in accordance with this Credit Agreement to make a Variable Rate RLC Advance, as contemplated by this Section, as a result of a drawing under a Letter of Credit, shall be absolute and

-25-

unconditional and without recourse to the Issuing Bank and shall not be affected by any circumstance, including (i) any set-off, counterclaim, recoupment, defense, or other right which such Bank may have against the Issuing Bank, the Borrower, or any other Person for any reason whatsoever, (ii) the occurrence or continuance of a default, an Event of Default, or a Material Adverse Effect, or (iii) any other circumstance, happening, or event whatsoever, whether or not similar to any of the foregoing.

#### 2.19 REPAYMENT OF PARTICIPATIONS.

- (a) Upon (and only upon) receipt by the Administrative Agent for the account of the Issuing Bank of immediately available funds from the Borrower (i) in reimbursement of any payment made by the Issuing Bank under a Letter of Credit with respect to which any Bank has paid the Administrative Agent for the account of the Issuing Bank for such Bank's participation in such Letter of Credit pursuant to Section 2.18, or (ii) in payment of interest thereon, the Administrative Agent will pay to each Bank, in the same funds as those received by the Administrative Agent for the account of the Issuing Bank, the amount of such Bank's Pro Rata Share of such funds, and the Issuing Bank shall receive the amount of the Pro Rata Share of such funds of any Bank that did not so pay the Administrative Agent for the account of the Issuing Bank.
- (b) If the Administrative Agent or the Issuing Bank is required at any time to return to the Borrower, or to a trustee, receiver, liquidator, custodian, or any official in any Insolvency Proceeding, any portion of the payments made by the Borrower to the Administrative Agent for the account of the Issuing Bank pursuant to paragraph (a) in reimbursement of a payment made under a Letter of Credit or interest or fee thereon, each Bank shall, on demand of the Administrative Agent, forthwith return to the

Administrative Agent or the Issuing Bank the amount of its Pro Rata Share of any amounts so returned by the Administrative Agent or the Issuing Bank plus interest thereon from the date such demand is made to the date such amounts are returned by such Bank to the Administrative Agent or the Issuing Bank, at a rate per annum equal to the Federal Funds Rate in effect from time to time.

#### 2.20 ROLE OF THE ISSUING BANK.

- (a) Each Bank and Borrower agree that, in paying any drawing under a Letter of Credit, the Issuing Bank shall not have any responsibility to obtain any document (other than any sight draft and certificates expressly required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document.
- (b) No Administrative Agent-related Person nor any of the respective correspondents, participants or assignees of the Issuing Bank shall be liable to any Bank for: (i) any action taken or omitted in connection herewith at the request or with the approval of the Banks or (ii) any action taken or omitted in the absence of gross negligence or willful misconduct.

#### ARTICLE 3

#### PAYMENTS, FEES AND EURODOLLAR PROVISIONS

#### 3.1 PAYMENTS.

- (a) All payments and prepayments by the Borrower of principal of and interest on the Notes and all fees, expenses and any other Obligations payable to the Administrative Agent or the Banks in connection with the Loans shall be nonrefundable and made in Dollars or immediately available funds to the Banks not later than 11:00 a.m. (San Diego, California local time) on the dates called for under this Credit Agreement, at the Payment Office of the Administrative Agent. Funds received after such hour shall be deemed to have been received by the Administrative Agent on the next Business Day. Payment to the Administrative Agent as aforesaid shall be deemed payment to the Banks as well, regardless of whether the Administrative Agent makes the distributions contemplated by the Co-Lender Agreement.
- (b) Unless otherwise agreed to, in writing, or otherwise required by applicable law, payments will be applied first to any unpaid collection costs, late charges and other charges, then to accrued, unpaid interest, with any remaining amount to principal; provided, however, upon delinquency or other default, the Banks reserve the right to apply payments among principal, interest, late charges, collection costs and other charges at their discretion.
- (c) Interest shall be due and payable on each Loan on each Payment Date and on the RLC Maturity Date.
- (d) Whenever any payment to be made hereunder shall be stated to be due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day, and such extension of time shall in such case be included in the computation of interest, commission or fee, as the case may be.

#### 3.2 FEES.

- (a) RLC COMMITMENT FEE. Borrower agrees to pay the Administrative Agent for distribution to the Banks pursuant to Article 9A a fee (the "RLC Commitment Fee") in an amount equal to \$5,000.00 on the Closing Date.
- (b) EXTENSION FEE. Borrower agrees to pay the Administrative agent for distribution to the Banks pursuant to Article 9A the Extension Fee in the event Borrower elects to extend the RLC Maturity Date.
  - (c) LETTER OF CREDIT FEE. See Section 2.12(a) hereof.

-28-

- (d) OTHER FEES. Borrower agrees to pay such other fees as may be separately agreed to between Borrower, the Administrative Agent, the Issuing Bank and CB&T.
- $3.3\,$  COMPUTATIONS. All fees and interest on the Notes shall be computed on the basis of a year of 360-days/year and accrue on a daily basis for the actual number of days elapsed.
- 3.4 MAINTENANCE OF ACCOUNTS. The Banks shall maintain, in accordance with their usual practice, an account or accounts evidencing the indebtedness of the Borrower and the amounts payable and paid from time to time hereunder. In any legal action or proceeding in respect of this Credit Agreement, the entries made in the ordinary course of business in such account or accounts shall be

presumptive, absent manifest error, evidence of the existence and amounts of the obligations of the Borrower therein recorded. The failure to record any such amount shall not, however, limit or otherwise affect the obligations of the Borrower hereunder to repay all amounts owed hereunder, together with all interest accrued thereon as provided in the Notes.

#### 3.5 CERTAIN CONTINGENCIES.

- (a) If the contingency contemplated by Section 3.6 hereof should occur, the Borrower may at any time after receipt of such notice, and as long as the circumstances giving rise to the relevant claim continue, require the Banks to terminate upon not less than thirty days' notice the participation agreement with such Bank, unless such Bank has waived any claim to payment under those provisions.
- (b) If circumstances arise which would (or would upon the giving of notice) entitle a Bank to receive additional payments pursuant to Section 3.6 hereof, then the Bank shall promptly, upon becoming aware of such circumstances, notify the Borrower and, to the extent that it can legally do so without material prejudice to its own position, the Bank shall take such reasonable steps as may be available to it to mitigate the effects of such circumstances.
- 3.6 INCREASED CAPITAL REQUIREMENTS. In the event that, as a result of any Regulatory Change, compliance by any Bank with any applicable law or governmental rule, requirement, regulation, guideline or order (with which it is customary for banks in the relevant jurisdiction to comply whether or not having the force of law) has or would have the effect of reducing the rate of return on the capital of the Bank or any institution controlling the Bank as a consequence of or with reference to the issuance of a Letter of Credit to a level below that which the Bank or such other corporation could have achieved but for such change or compliance (taking into consideration the policies of the Bank or such other corporation with respect to capital), then from time to time the Borrower shall pay to such Bank such additional amount or amounts as will compensate the Bank for such reduction. The Bank will notify the Borrower of any Regulatory Change that will entitle the Bank to compensation pursuant to this Section as promptly as practicable, but in any event within thirty (30) days after the Bank obtains knowledge thereof; provided, however, that if the Bank fails to give such notice within thirty (30) days after it obtains knowledge of such a Regulatory Change, the Bank shall, with respect to compensation payable in respect of any costs resulting from such Regulatory Change, only be entitled to payment for costs incurred from and after the date that the Bank does give such notice. Such Bank shall deliver to the Borrower a written certificate which states the additional amount(s) due and payable, showing in reasonable detail the calculation of such amount and provide evidence to substantiate the Bank's claim for such amount(s).
- 3.7 PREPAYMENT. Borrower may prepay any Variable Rate Advances without penalty or premium at any time upon at least two (2) Business Days' notice.

-29-ARTICLE 4.

#### EXTENSION

# 4.1 EXTENSION OF THE RLC MATURITY DATE.

- (a) At the option of the Banks, in their sole and absolute discretion, the RLC Maturity Date may be extended an additional ninety (90) days once subject to the following terms and conditions:
  - (i) Borrower at least thirty (30) days prior to the current Maturity Date shall have submitted to the Administrative Agent a written request that the RLC Maturity Date be extended an additional ninety (90) days;
  - (ii) There has been no material adverse change in the financial condition of Borrower;
  - (iii) Borrower shall be in compliance with all representations, warranties, Financial Covenants and other covenants contained herein;
  - (iv) No Event of Default and no event, that with the giving of notice or the passage of time, or both, would be an Event of Default, shall have occurred and be continuing on the date of Borrower's request for the extension to the Administrative Agent or on the commencement of the extension period;
    - (v) Borrower shall have paid the Extension Fee; and
  - (vi) Borrower shall have completed the Collateral Requirements to the satisfaction of the Administrative Agent and the Banks with respect to all property included in the RLC Borrowing Base Inventory.
- (b) If the Administrative Agent shall fail to notify the Borrower in writing before the current RLC Maturity Date that the extension of the RLC

Maturity Date has been approved by all the Banks, the Banks shall be deemed not to have affirmed such request and the RLC Maturity Date shall be deemed unchanged.

(c) If the Administrative Agent shall notify the Borrower in writing that all the Banks have approved the extension of the RLC Maturity Date, the RLC Maturity Date shall be deemed to have been modified accordingly.

-30-ARTICLE 5.

#### CONDITIONS PRECEDENT

The obligation of the Banks to make the Loans is subject to the full prior satisfaction of each of the following conditions precedent and, as to their obligation to make the initial Advance hereunder and to make any future Advance or the Issuing Bank's obligation to issue the initial Letter of Credit, to the full prior satisfaction at each such time of each of the conditions precedent in Sections 5.2 and 5.3 hereof:

- 5.1 CLOSING. The Administrative Agent shall have received the following, each in form and substance satisfactory to the Required Banks:
  - (a) THIS CREDIT AGREEMENT. This Credit Agreement, duly executed by Borrower.
  - (b) THE NOTES. The Notes, each duly executed, as provided in Article 2 hereof.
  - (c) OFFICER'S CERTIFICATE. A certificate signed by an Authorized Officer of the Borrower, stating that (to the best knowledge and belief of the Borrower, after reasonable inquiry and review of matters pertinent to the subject matter of such certificate): (i) all of the representations and warranties contained in Article 6 of this Credit Agreement and in the other Credit Documents are, in all material respects, true and correct as of the date hereof (other than those of such representations which by their express terms speak to a date prior to such date, which representations are, in all material respects, true and correct as of such respective dates); (ii) no event has occurred and is continuing, or would result from the advance of the proceeds of the Loans, which would constitute an Event of Default, and (iii) no change or changes having a Material Adverse Effect have occurred in the business or financial condition of Borrower since the date of the last financial statements of Borrower heretofore delivered to the Banks.
  - (d) ORGANIZATIONAL DOCUMENTS. A copy of the current organizational documents of Borrower and of each Guarantor, including all amendments thereto, certified as current and complete by the appropriate authority of the state of said corporation's incorporation, together with evidence of said corporation's good standing in said corporation's state of incorporation and in every other state in which it is doing business or the conduct of said corporation's business requires such standing for the enforcement of material contracts.
  - (e) SECRETARY'S CERTIFICATES AND RESOLUTIONS. A certificate of the corporate secretary or assistant secretary of Borrower and of each Guarantor, signed by the duly appointed secretary thereof and issued as of the Closing Date, certifying that (i) attached thereto is a true and complete copy of the corporate by-laws of said corporation in effect on the

\_21\_

date of passage of the corporate resolutions described immediately below and at all subsequent times to and including the date of the certificate, (ii) no change has been made to said corporation's charter documents other than as reflected in the certified copies submitted in connection with the delivery of this Credit Agreement or as approved in writing by the Administrative Agent, and (iii) attached thereto are proper resolutions, authorizations and certificates relating to the authority of any person executing documents on behalf of such entity.

- (f) COMPLIANCE CERTIFICATE. A Compliance Certificate executed by the Borrower, indicating that Borrower is in compliance with all Financial Covenants as of June 30, 1999.
- (g) OPINION OF COUNSEL. An opinion of counsel to Borrower and each Guarantor as to those matters reasonably required by the Banks.
- (h) FINANCIAL STATEMENTS. Borrower's audited financial statements as of December 31, 1998.
- (i) FEES AND COSTS. Payment of the RLC Commitment Fee and the costs of the Administrative Agent and the Banks.
- (j) LIENS. No Liens other than Permitted Liens shall be outstanding. All other Liens shall have been terminated.

- (k) GUARANTIES. Fully executed Guaranties from all Guarantors.
- (1) QUARTERLY INVENTORY REPORT. A Quarterly Inventory Report executed by Borrower as of June 30, 1999.
- (m) RLC BORROWING BASE INVENTORY REPORT. An RLC Borrowing Base Inventory Report executed by Borrower.
- (n) INVENTORY INSPECTION. An Inventory Inspection shall have been completed in accordance with Section 7.18.
- (o) ADDITIONAL INFORMATION. Such other information and documents as may reasonably be required by the Banks or their counsel.
- 5.2 NO EVENT OF DEFAULT; NO MATERIAL ADVERSE CHANGE.
- (a) No Event of Default  $\,$  known to Borrower  $\,$  shall have occurred and be continuing, or result from the making of the Loans.

-32-

- (b) Since the date of the most recent financial statements provided to the Banks by Borrower, no change shall have occurred in the business or financial condition of Borrower that could have a Material Adverse Effect.
- 5.3 REPRESENTATIONS AND WARRANTIES. The representations and warranties contained in Article 6 hereof shall be true and correct in all material respects, with the same force and effect as though made on and as of the Closing Date (other than those of such representations which by their express terms speak to a date prior to that date, which representations shall, in all material respects, be true and correct as of such respective date).

-33-ARTICLE 6.

#### REPRESENTATIONS AND WARRANTIES

To induce the Banks to make the Loans,  $\,$  Borrower represents and warrants to the Banks that:

- 6.1 ORGANIZATION AND GOOD STANDING. It is duly organized under the laws of the state of its organization, is validly existing and is in good standing, to the extent required by law, in each state in which it is doing business. It has the legal power and authority to own its properties and assets and to transact the business in which it is engaged and is or will be qualified in those states wherein the nature of its proposed business and property will make such qualifications necessary or appropriate in the future.
- 6.2 AUTHORIZATION AND POWER. It has the corporate power and requisite authority to execute, deliver and perform this Credit Agreement, the Notes and the other Credit Documents to be executed by it; it is duly authorized to, and has taken all action, corporate or otherwise, necessary to authorize it to, execute, deliver and perform this Credit Agreement, the Notes and such other Credit Documents and is and will continue to be duly authorized to perform this Credit Agreement, the Notes and such other Credit Documents.
- 6.3 NO CONFLICTS OR CONSENTS. Neither the execution and delivery of this Credit Agreement, the Notes or the other Credit Documents to which it is a party, nor the consummation of any of the transactions herein or therein contemplated, nor compliance with the terms and provisions hereof or with the terms and provisions thereof, (a) will contravene or conflict with: (i) any provision of law, statute or regulation to which it is subject, (ii) any judgment, license, order or permit applicable to it, (iii) any indenture, loan agreement, mortgage, deed of trust, or other agreement or instrument to which it is a party or by which it may be bound, or to which it may be subject, or (b) will violate any provision of its organizational documents. No consent, approval, authorization or order of any court or Governmental Authority or other Person is required in connection with the execution and delivery by it of the Credit Documents or to consummate the transactions contemplated hereby or thereby, or if required, such consent, approval, authorization or order shall have been obtained.
- 6.4 ENFORCEABLE OBLIGATIONS. This Credit Agreement, the Notes and the other Credit Documents are the legal, valid and binding obligations of Borrower, enforceable against Borrower in accordance with their respective terms, except as limited by bankruptcy, insolvency or other laws or equitable principles of general application relating to the enforcement of creditors' rights.
- 6.5 FINANCIAL CONDITION. It has delivered to the Banks copies of the Borrower's most recent audited consolidated financial statements. Such financial statements, in all material respects, fairly present the financial position of Borrower as of such date and have been prepared in accordance with GAAP. Since the date thereof, Borrower has not discovered any obligations, liabilities or indebtedness (including contingent and indirect liabilities and obligations or unusual forward or long-term commitments) which in the aggregate are material

and adverse to the financial position or business of Borrower that should have been but were not reflected in such financial statements.

- 6.6 FULL DISCLOSURE. There is no material fact that it has not disclosed to the Banks that would have a Material Adverse Effect. No certificate or statement delivered herewith or heretofore by it to the Banks in connection with negotiations of this Credit Agreement, contains any untrue statement of a material fact or omits to state any material fact necessary to keep the statements contained herein or therein from being misleading.
- 6.7 NO DEFAULT. No event or condition has occurred and is continuing that constitutes an Event of Default.
- 6.8 SIGNIFICANT DEBT AGREEMENTS. It is not in default in any material respect under any Significant Debt Agreement.
- 6.9 NO LITIGATION. There are no actions, suits or legal, equitable, arbitration or administrative proceedings pending, or to its actual knowledge overtly threatened, against Borrower that would, if adversely determined, have a Material Adverse Effect.
- 6.10 TAXES. It has filed or caused to be filed all returns and reports which are required to be filed by any jurisdiction, and has paid or made provision for the payment of all taxes, assessments, fees or other governmental charges imposed upon its properties, income or franchises, as to which the failure to file or pay would have a Material Adverse Effect, except such assessments or taxes, if any, which are being contested in good faith by appropriate proceedings.
- 6.11 ERISA. (a) No Reportable Event has occurred and is continuing with respect to any Plan; (b) PBGC has not instituted proceedings to terminate any Plan; (c) neither the Borrower, any member of the Controlled Group, nor any duly-appointed administrator of a Plan (i) has incurred any liability to PBGC with respect to any Plan other than for premiums not yet due or payable or (ii) has instituted or intends to institute proceedings to terminate any Plan under Section 4041 or 4041A of ERISA; and (d) each Plan of Borrower has been maintained and funded in all material respects in accordance with its terms and in all material respects in accordance with all provisions of ERISA applicable thereto. The Borrower does not participate in, or is not required to make contributions to, any Multi-employer Plan (as that term is defined in Section 3(37) of ERISA).
- 6.12 COMPLIANCE WITH LAW. It is in substantial compliance with all laws, rules, regulations, orders and decrees that are applicable to it, or its properties, noncompliance with which would have a Material Adverse Effect.
- 6.13 SURVIVAL OF REPRESENTATIONS, ETC. All representations and warranties by Borrower herein shall survive the making of the Loans and the execution and delivery of the Notes; any investigation at any time made by or on behalf of the Banks shall not diminish the Banks' right to rely on the representations and warranties herein.

-35-

- 6.14 RECITALS. The recitals and statements of intent appearing in this Credit Agreement are true and correct.
- $6.15\,$  NO STOCK PURCHASE. No part of the proceeds of any financial accommodation made by the Banks in connection with this Credit Agreement will be used to purchase or carry "margin stock," as that term is defined in Regulation U, or to extend credit to others for the purpose of purchasing or carrying such margin stock.
- 6.16 SOLVENT. It (both before and after giving effect to the Loans contemplated hereby) is solvent, has assets having a fair value in excess of the amount required to pay its probable liabilities on its existing debts as they become absolute and matured, and has, and will have, access to adequate capital for the conduct of its business and the ability to pay its debts from time to time incurred in connection therewith as such debts mature.
- 6.17 ADVANCES. Each request for an Advance or for the extension of any financial accommodation by the Banks whatsoever shall constitute an affirmation that the representations and warranties contained herein are, true and correct as of the time of such request. All representations and warranties made herein shall survive the execution of this Credit Agreement, all advances of proceeds of the Loans and the execution and delivery of all other documents and instruments in connection with the Loans and/or this Credit Agreement, so long as any Bank has any commitment to lend hereunder and until the Loans have been paid in full and all of Borrower's obligations under this Credit Agreement and the Notes been fully discharged.
  - 6.18 YEAR 2000 COMPLIANCE.
    - (a) All devices, systems, machinery, information technology, computer

software and hardware, and other date sensitive technology (jointly and severally the "Systems") necessary for Borrower to carry on its business as presently conducted and as contemplated to be conducted in the future are Year 2000 Compliant or will be Year 2000 Compliant within a period of time calculated to result in no material disruption of any of Borrower's business operations. For purposes of these provisions, "Year 2000 Compliant" means that such Systems are designed to be used prior to, during and after the Gregorian calendar year 2000 A.D. and will operate during each such time period without error relating to date data, specifically including any error relating to, or the product of, date data which represents or references different centuries or more than one century.

(b) Borrower has: (1) undertaken a detailed inventory, review, and assessment of all areas within its business and operations that could be adversely affected by the failure of Borrower to be Year 2000 Compliant on a timely basis; (2) developed a detailed plan and time line for becoming Year 2000 Compliant on a timely basis, and (3) to date, implemented that plan in accordance with that timetable in all material respects.

-36-

- (c) As stated in its SEC 10K and 10Q  $\,$  reports,  $\,$  Borrower  $\,$  is Year 2000 Compliant.
- 6.19 INVESTMENT COMPANY ACT. It is not an "investment company" or a company "controlled" by an "investment company" within the meaning of the Investment Company Act of 1940, as amended.
- 6.20 PUBLIC UTILITY HOLDING COMPANY ACT. It is not a "holding company," or a "Subsidiary company" of a "holding company," or an "affiliate" of a "holding company" or of a "Subsidiary company" of a "holding company," within the meaning of the Public Utility Holding Company Act of 1935, as amended. TITLE TO PROPERTIES: POSSESSION. It has fee simple title to, or valid leasehold interests in, all its respective material properties and assets, and properties included in the RLC Borrowing Base are subject only to Permitted Liens. It has all licenses and rights necessary to enable it to use all material technology used by it in its respective operations.
- 6.22 ENVIRONMENTAL AND SAFETY MATTERS. It has no material undisclosed environmental liabilities.
- 6.23 SUBSIDIARIES. It has no Subsidiaries except for the Subsidiaries listed on Schedule 6.23 hereto.
- 6.24 INDEBTEDNESS BORROWER. It has no Indebtedness except for Indebtedness described in Section 8.1 hereof and on Schedule 6.24 hereto as of the date thereof.
- 6.25 INDEBTEDNESS SUBSIDIARIES. No Subsidiary has any Indebtedness except for Indebtedness described in Section 8.2 hereof and on Schedule 6.25 hereto as of the date thereof.

#### -37-ARTICLE 7.

#### AFFIRMATIVE COVENANTS

Until payment in full of the Notes and the complete performance of the Obligations, and so long as any Bank has any Commitment outstanding to Borrower, the Borrower, agrees that:

- 7.1 FINANCIAL STATEMENTS, REPORTS AND DOCUMENTS. It shall deliver, or cause to be delivered, to the Banks each of the following:
  - (a) CONSOLIDATED QUARTERLY STATEMENTS OF THE BORROWER. As soon as available, and in any event within sixty (60) days after the end of each fiscal quarter of the Borrower, copies of the consolidated balance sheet of the Borrower as of the end of such fiscal quarter, and consolidated statements of income and cash flows of the Borrower for that fiscal quarter and for the portion of the fiscal year ending with such fiscal quarter, in each case setting forth in comparative form the figures for the corresponding period of the preceding fiscal year, all in reasonable detail and fairly stated and prepared by the Borrower in accordance with GAAP.
  - (b) CONSOLIDATED ANNUAL STATEMENTS OF THE BORROWER. As soon as available and in any event within one hundred five (105) days after the close of each fiscal year of the Borrower, audited consolidated financial statements of the Borrower, including its consolidated balance sheet as of the close of such fiscal year and consolidated statements of income and cash flows of the Borrower for such fiscal year, in each case setting forth in comparative form the figures for the preceding fiscal year, all in reasonable detail and accompanied by an unqualified opinion thereon of independent public accountants of recognized national standing selected by the Borrower and acceptable to the Banks, to the effect that such financial statements have been prepared in accordance with GAAP (except for changes

in which such accountants concur) and that the examination of such accounts in connection with such financial statements has been made in accordance with generally accepted auditing standards and, accordingly, includes such tests of the accounting records and such other auditing procedures as were considered necessary in the circumstances.

(c) COMPLIANCE CERTIFICATE OF THE BORROWER. Within sixty (60) days after the end of each of the first three fiscal quarters of each fiscal year of the Borrower hereafter and within one hundred five (105) days after the end of the final fiscal quarter of each fiscal year of the Borrower hereafter, a certificate (the "Compliance Certificate") substantially in the form of Exhibit "A-1" attached hereto signed by the Authorized Officer that is the Chief Financial Officer of the Borrower, (i) stating that a review of the activities of Borrower during such quarter has been made under his/her supervision, that, as of such date, Borrower has observed, performed and fulfilled each and every obligation and covenant contained herein and no Event of Default exists under any of the same or, if any Event of Default shall have occurred, specifying the nature and status

-38-

thereof, and stating that all financial statements delivered to the Banks during the respective period pursuant to Section 7.1(a) and 7.1(b) hereof, to such officer's knowledge after due inquiry, fairly present in all material respects the financial position of the Borrower and the results of its operations at the dates and for the periods indicated, and have been prepared in accordance with GAAP, subject to year end audit and adjustments, (ii) setting forth in such level of detail as the Banks shall reasonably require a calculation of the Financial Covenants as of the end of that fiscal quarter, and (iii) certifying that the Borrower is in compliance with the most recent RLC Borrowing Base Inventory Report.

- (d) MANAGEMENT LETTERS. With the audited fiscal year-end statements submitted under Section 7.1(b) above, the management letter, if any, of the Borrower's certified public accountants issued in connection with such audit.
- (e) FISCAL YEAR PROJECTIONS. No later than one hundred five (105) days after the end of each fiscal year of the Borrower commencing with that fiscal quarter ending December 31, 1999, consolidated financial statements of the Borrower for the new fiscal year of the Borrower, based on its current financial projections for such period, to include balance sheet and statement of income, all in detail reasonably acceptable to the Banks.
- (f) RLC BORROWING BASE. No later than twenty (20) days after the end of each month hereafter, (i) a report (the "RLC Borrowing Base Inventory Report") substantially in the form of Exhibit "B-1" attached hereto, signed by an Authorized Officer and (ii) a certificate (the "RLC Borrowing Base Compliance Certificate") substantially in the form of Exhibit "B-2" attached hereto, signed by an Authorized Officer.
- (g) QUARTERLY INVENTORY REPORT. No later than sixty (60) days after the end of each fiscal quarter of the Borrower hereafter, a consolidated inventory and pipeline report (the "Quarterly Inventory Report") substantially in the form of Exhibit "A-2" attached hereto, signed by an Authorized Officer.
- (h) SEC REPORTS. Promptly when filed with the SEC, copies of all  $10 \, \mathrm{K}$ ,  $10 \, \mathrm{Q}$  and any other reports files by the Borrower with the SEC.
- (i) OTHER INFORMATION. Such other information concerning the business, properties or financial condition of Borrower as the Banks shall reasonably request.
- 7.2 PAYMENT OF TAXES AND OTHER INDEBTEDNESS. It shall pay and discharge (i) all income taxes and payroll taxes, (ii) all taxes, assessments, fees and other governmental charges imposed upon it or upon its income or profits, or upon any property belonging to it, before delinquent, which become due and payable, (iii) all lawful claims (including claims for labor, materials and supplies), which, if unpaid, might become a Lien upon any of its property, and (iv) all of its

-39-

Indebtedness as it becomes due and payable, except as prohibited hereunder; provided, however, that it shall not be required to pay any such tax, assessment, charge, levy, claims or Indebtedness if and so long as the amount, applicability or validity thereof shall currently be contested in good faith by appropriate actions and appropriate accruals and reserves therefor have been established in accordance with GAAP.

- 7.3 MAINTENANCE OF EXISTENCE AND RIGHTS; CONDUCT OF BUSINESS. It shall preserve and maintain its corporate existence and all of its rights, privileges and franchises necessary or desirable in the normal conduct of its business, and conduct its business in an orderly and efficient manner consistent with good business practices.
  - 7.4 NOTICE OF DEFAULT AND NON-COMPLIANCE. It shall furnish to the Banks

immediately upon becoming actually aware of the existence of any event or condition that constitutes an Event of Default, of any non-compliance with any Financial Covenant or other covenant, and of any negative balance of the RLC Borrowing Base Availability, together with a written notice specifying the nature and period of existence thereof and the action which it is taking or proposes to take with respect thereto.

- 7.5 OTHER NOTICES. It shall promptly notify the Banks of (a) any Material Adverse Effect, (b) any waiver, release or default under any Significant Debt Agreement, (c) except as to any claim not covered as a result of an insurance deductible provision, any claim not covered by insurance against Borrower or any of Borrower's properties, and (d) the commencement of, and any material determination in, any litigation with any third party or any proceeding before any Governmental Authority affecting it, except litigation or proceedings which, if adversely determined, would not have a Material Adverse Effect.
- $7.6\,$  COMPLIANCE WITH CREDIT DOCUMENTS. It shall comply with any and all covenants and provisions of this Credit Agreement, the Notes and all other Credit Documents.
- 7.7 COMPLIANCE WITH SIGNIFICANT DEBT AGREEMENTS. It shall comply in all material respects with all Significant Debt Agreements.
- 7.8 OPERATIONS AND PROPERTIES. It shall keep in good working order and condition, ordinary wear and tear excepted, all of its assets and properties which are necessary to the conduct of its business.
- 7.9 BOOKS AND RECORDS; ACCESS. It shall give any authorized representative of the Banks access during normal business hours to, and permit such representative to examine, copy or make excerpts from, any and all books, records and documents in its possession of and relating to the Loans, and to inspect any of its properties. It will maintain complete and accurate books and records of its transactions in accordance with good accounting practices.
- 7.10 COMPLIANCE WITH LAW. It shall comply with all applicable laws, rules, regulations, and all final, nonappealable orders of any Governmental Authority applicable to it or any of its property, business operations or transactions,

-40-

including without limitation any environmental laws, rules and regulations, a breach of which could result in a Material Adverse Effect.

- 7.11 AUTHORIZATIONS AND APPROVALS. It shall promptly obtain, from time to time at its own expense, all such governmental licenses, authorizations, consents, permits and approvals as may be required to enable it to comply with its obligations hereunder and under the other Credit Documents and to operate its businesses as presently or hereafter duly conducted.
- 7.12 ERISA COMPLIANCE. With respect to its Plans, it shall (a) at all times comply with the minimum funding standards set forth in Section 302 of ERISA and Section 412 of the Code or shall have duly obtained a formal waiver of such compliance from the proper authority; (b) at the Banks' request, within thirty (30) days after the filing thereof, furnish to the Banks copies of each annual report/return (Form 5500 Series), as well as all schedules and attachments required to be filed with the Department of Labor and/or the Internal Revenue Service pursuant to ERISA, in connection with each of its Plans for each year of the plan; (c) notify the Banks within a reasonable time of any fact, including, but not limited to, any Reportable Event arising in connection with any of its Plans, which constitutes grounds for termination thereof by the PBGC or for the appointment by the appropriate United States District Court of a trustee to administer such Plan, together with a statement, if requested by the Banks, as to the reason therefor and the action, if any, proposed to be taken with respect thereto; and (d) furnish to the Banks within a reasonable time, upon the Banks' request, such additional information concerning any of its Plans as may be reasonably requested.
- 7.13 FURTHER ASSURANCES. It shall make, execute or endorse, and acknowledge and deliver or file or cause the same to be done, all such notices, certifications and additional agreements, undertakings or other assurances, and take any and all such other action, as the Banks may, from time to time, deem reasonably necessary or proper to fully evidence the Loan.
- 7.14 NEWS RELEASES. It shall promptly forward to the Banks copies of all news releases made by it to the news media as to anything of material significance with respect to its financial status.
  - 7.15 INSURANCE. It shall maintain in full force and effect at all times:
  - (a) Policies of all risk coverage insurance covering (i) its real property of every kind and description, and wherever located, in coverage amounts not less than, from time to time, the full replacement value of all insurable improvements situated thereon and (ii) its tangible personalty in respective coverage amounts not less than, from time to time, the fair market value thereof.

(b) Policies of insurance evidencing personal liability and property damage liability coverages in amounts not less than \$5,000,000 (combined single limit for bodily injury and property damage), and an umbrella excess liability coverage in an amount not less than \$5,000,000 shall be in effect with respect to Borrower.

-41-

(c) Policies of workers' compensation insurance in amounts and with coverages as legally required.

Without limitation of the foregoing, it shall at all times maintain insurance coverages in scope and amount not less than, and not less extensive than, the scope and amount of insurance coverages customary in the trades or businesses in which it is from time to time engaged. All of the aforesaid insurance coverages shall be issued by insurers reasonably acceptable to the Banks. Copies of all policies of insurance evidencing such coverages in effect from time to time, or certificates of such policies satisfactory to the Banks, shall be delivered to the Banks prior to the initial Advance under this Credit Agreement and upon reasonable notice upon issuance of new policies thereafter. From time to time, promptly upon the Banks' request, it shall provide evidence satisfactory to the Banks that required coverage in required amounts is in effect. At the Banks' option, it shall deliver to the Banks certified copies of all such policies of insurance in effect from time to time, to be retained by the Banks so long as the Banks shall have any commitment to lend hereunder and/or any portion of the Obligations shall be outstanding or unsatisfied. All such insurance policies shall provide for at least thirty (30) days prior written notice of the cancellation or modification thereof to the Banks.

- 7.16 CHANGE IN CONTROL OR MANAGEMENT. Should there be a Change in Control or a material change in management as to the Borrower, the Loans shall be immediately due and payable unless the Required Banks should consent to the substitute management team. The termination of both Landon and Hilton as co-chief executive officers of the Borrower shall be deemed a material change in management.
- 7.17 ENVIRONMENTAL INVESTIGATION. Borrower shall cause any real property being added to the Inventory or being developed, to be reviewed by a qualified environmental engineering firm and shall comply with the recommendations of the environmental report of such firm.

## 7.18 INVENTORY INSPECTION.

- (a) The Administrative Agent, or a Person at the direction of the Administrative Agent (the "RLC Borrowing Base Inspector") may, each fiscal quarter, review at Borrower's expense, its Inventory status from its purchase order records. A random sample of Units shall be selected from each division's list of purchase orders, not to be less than four (4) per Qualified Subdivision. A field inspection of these Units will then be conducted to verify the accuracy of the invoices to the actual physical construction status.
- (b) In the event that the above review indicates that there are in fact material deviations, at the sole and absolute discretion of the Banks the Banks may require a complete due diligence analysis of the Inventory.
- (c) Borrower shall provide to the Administrative Agent and the RLC Borrowing Base Inspector such information as necessary to enable it to ascertain the GAAP costs expended as of the end of the applicable reporting period for each Unit selected for inspection.

-42-

(d) Borrower shall reimburse the Administrative Agent, within twenty (20) Business Days of its demand, for the costs of performing all Inventory Inspections.

## 7.19 YEAR 2000 COMPLIANCE.

- (a) Furnish such additional information, statements and other reports with respect to Borrower's activities, course of action and progress towards becoming Year 2000 Compliant as the Banks may request from time to time.
- (b) In the event of any change in circumstances that causes or will likely cause any of Borrower's representations and warranties with respect to its being or becoming Year 2000 Compliant to no longer be true (hereinafter, referred to as a "Change in Circumstances") then Borrower shall promptly, and in any event within ten (10) days of receipt of information regarding a Change in Circumstances, provide the Banks with written notice (the "Notice") that describes in reasonable detail the Change in Circumstances and how such Change in Circumstances caused or will likely cause Borrower's representations and warranties with respect to being or becoming Year 2000 Compliant to no longer be true. Borrower shall, within ten (10) days of a request, also provide the Banks with any additional information the Banks request of Borrower in connection with the Notice and/or a Change in Circumstances.

- (c) Promptly upon its becoming available, furnish to the Banks one copy of each financial statement, report, notice, or proxy statement sent by Borrower to stockholders generally and of each regular or periodic report, registration statement or prospectus filed by Borrower with any securities exchange or the Securities and Exchange Commission or any successor agency, and of any order issued by any Governmental Authority in any proceeding to which Borrower is a party. For purposes of these provisions, "Governmental Authority" shall mean any government (or any political subdivision or jurisdiction thereof), court, bureau, agency or other governmental entity having or asserting jurisdiction over Borrower or any of its business, operations or properties.
- (d) Give any representative of the Banks access during all business hours to, and permit such representative to examine, copy or make excerpts from, any and all books, records and documents in the possession of Borrower and relating to its affairs, and to inspect any of the properties and Systems of Borrower, and to project test the Systems to determine if they are Year 2000 Compliant in an integrated environment, all at the sole cost and expense of the Banks.

## -43-ARTICLE 8.

#### NEGATIVE COVENANTS

Until payment in full of the Notes and the performance of the Obligations, and so long as any Bank has any Commitment outstanding to Borrower, Borrower, agrees that:

- 8.1 NO UNSECURED DEBT -- BORROWER. It shall not become or remain obligated either directly or as a guarantor or surety for any Unsecured Debt, except:
  - (a) Indebtedness to the Banks hereunder.
  - (b) Unsecured trade, customer deposits, utility or accounts payable arising in the ordinary course of its business.
    - (c) Off-balance sheet leases and option agreements.
    - (d) Any Unsecured Debt described on Schedule 6.24 attached hereto.
  - (e) Future Senior Unsecured Debt so long as the RLC Borrowing Base Availability, after giving effect to such Indebtedness, does not have a negative balance.
  - (f) Subordinated Debt not in excess in the aggregate of \$20,000,000.00, all of which Subordinated Debt shall have an initial maturity date extending beyond the RLC Maturity Date.
  - (g) Subordinated Debt in excess in the aggregate of \$20,000,000.00 so long as the RLC Borrowing Base Availability, after giving effect to such excess, does not have a negative balance.
- $8.2\ \text{NO}$  INDEBTEDNESS GUARANTORS. No Guarantor shall become or remain obligated either directly or as a guarantor or surety of any Unsecured Debt, except:
  - (a) The Guaranties to the Banks hereunder.
  - (b) Unsecured trade, customer deposits, utility or accounts payable arising in the ordinary course of its business.
    - (c) Off-balance sheet leases and option agreements.
    - (d) Any Unsecured Debt described on Schedule 6.25 attached hereto.

## -44-

- 8.3 LIENS. On and after the date hereof, neither Borrower nor any Guarantor shall create or suffer to exist Liens upon the RLC Borrowing Base Inventory, real or personal, except (i) Liens, if any, for the benefit of the Banks, and (ii) Permitted Liens.
- 8.4 MERGERS, CONSOLIDATIONS, TRANSFERS OF ASSETS. Neither Borrower nor any Guarantor shall (i) merge into or consolidate with or acquire any other Person, or permit any other Person to merge into or consolidate with it, unless Borrower or Guarantor is the surviving entity, or (ii) sell, transfer, lease or otherwise dispose of (in one transaction or in a series of related transactions) all or substantially all of its assets whether now owned or hereafter acquired except for any transfer of assets undertaken in the ordinary course of business.
- 8.5 AMENDMENTS TO ORGANIZATIONAL DOCUMENTS. Neither Borrower nor any Guarantor shall amend its organizational documents if the result thereof could result in the occurrence directly or indirectly of a Material Adverse Effect.

- 8.6 MARGIN STOCK. It shall not use any proceeds of the Loans, or any proceeds of any other or future financial accommodation from the Banks for the purpose, whether immediate, incidental or ultimate, of purchasing or carrying any "margin stock" as that term is defined in Regulation U or to reduce or retire any indebtedness undertaken for such purposes within the meaning of said Regulation U, and will not use such proceeds in a manner that would involve Borrower in a violation of Regulation U or of any other Regulation of the Board of Governors of its Federal Reserve System, nor use such proceeds for any purpose not permitted by Section 7 of the Exchange Act, as amended, or any of the rules or regulations respecting the extensions of credit promulgated thereunder.
- 8.7 FISCAL YEAR. Unless consented to by the Required Banks, it shall not change the times of commencement or termination of its fiscal year or other accounting periods; or change its methods of accounting other than to conform to GAAP applied on a consistent basis. After any such changes, its method of accounting shall conform to GAAP.

#### 8.8 INVESTMENTS.

- (a) It shall not invest in any non-consolidated joint ventures that would result in the aggregate of such joint venture investments, whether in the form of equity or debt, exceeding twenty percent (20.0%) of its Consolidated Tangible Net Worth.
- (b) It shall not acquire, engage or invest in any business or business activity unrelated to the home-building business that would result in the aggregate of such investments exceeding five percent (5.0%) of its Consolidated Tangible Net Worth.
- 8.9 SUBORDINATED DEBT PAYMENTS. Upon the occurrence of an Event of Default and during the continuation thereof, it shall not make any principal and/or interest payments on Subordinated Debt.

-45-

# 8.10 FINANCIAL COVENANTS. It will not permit:

- (a) Its Leverage Ratio to be more than 1.75 to 1.0 at the end of any fiscal quarter;
- (b) Its Consolidated Tangible Net Worth to be less than the Consolidated Tangible Net Worth Requirement at the end of any fiscal quarter; Its Interest Coverage Ratio to be less than 2.50 to 1.0 at the end of any fiscal quarter;
- (c) Its Fixed Charge Coverage Ratio to be less than 1.50 to 1.0 at the end of any fiscal guarter;

#### provided that:

- (i) With respect to Consolidated Tangible Net Worth, the sum of its Nonentitled Land, Entitled Land, Entitled Developing Land, Finished Lots and contract option deposits shall not exceed one hundred fifty percent (150.0%) of Consolidated Tangible Net Worth, not more than eighty percent (80.0%) of Consolidated Tangible Net Worth shall consist of Entitled Land and Entitled Developing Land (excluding Finished Lots), and not more than twenty-five percent (25.0%) shall consist of Nonentitled Land and Entitled Land not under development, in each case based on its cost in accordance with GAAP.
- (ii) With respect to Consolidated Tangible Net Worth, all land owned, including without limitation Finished Lots but excluding Finished Lots under option contract, shall not exceed a 3-year supply, based on the most recent quarterly report of its four-quarter sales rate, net of cancellations.

-46-ARTICLE 9.

## EVENTS OF DEFAULT

- 9.1 EVENTS OF DEFAULT. An "Event of Default" shall exist if any one or more of the following events (herein collectively called "Events of Default") shall occur and be continuing:
  - (a) Borrower shall fail to pay any principal of, or interest on, either Note when the same shall become due or payable and such failure continues for two (2) days after notice thereof to Borrower.
  - (b) Any failure to observe any Financial Covenant and such failure continues unremedied for a period of ten (10) days after notice thereof to Borrower.
  - (c) Any failure or neglect to perform or observe any of the covenants, conditions, provisions or agreements of Borrower contained herein, or in

any of the other Credit Documents (other than a failure or neglect described in one or more provisions of this Section 9.1) and such failure or neglect continues unremedied for a period of thirty (30) days after notice thereof to Borrower so long as such remedy is commenced within ten (10) days after notice thereof to Borrower and is diligently pursued to completion within such thirty (30) day period.

- (d) Any material warranty, representation or statement contained in this Credit Agreement or any of the other Credit Documents, or which is contained in any certificate or statement furnished or made to the Banks pursuant hereto or in connection herewith or with the Loans, shall be or shall prove to have been false when made or furnished.
- (e) The occurrence of any "event of default" or "default" by Borrower under any agreement, now or hereafter existing, to which any Bank, or an Affiliate of any Bank, and Borrower or an Affiliate of Borrower are a party, in each case after the expiration of any notice and cure period.
- (f) Borrower or any Guarantor shall (i) fail to pay any Indebtedness thereof (other than the Notes or Indebtedness that is Non-Recourse Debt), due under any agreement or agreements, for any Indebtedness of \$1,000,000.00 or more or for Indebtedness aggregating \$5,000,000.00 or more, or any interest or premium thereon, when due (whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise) or within any applicable grace period, (ii) fail to perform or observe any term, covenant, or condition on its part to be performed or observed under any agreement or instrument relating to such Indebtedness, within any applicable grace period when required to be performed or observed, if the effect of such failure to perform or observe is to accelerate the maturity of such Indebtedness, or any such Indebtedness shall be declared to be due and payable, or required to be prepaid (other than by a regularly scheduled

\_17\_

prepayment), prior to the stated maturity thereof, or (iii) allow the occurrence of any material event of default with respect to such Indebtedness.

- (g) Any one or more of the Credit Documents shall have been determined to be invalid or unenforceable against Borrower or any Guarantor executing the same in accordance with the respective terms thereof, or shall in any way be terminated or become or be declared ineffective or inoperative, so as to deny the Banks the substantial benefits contemplated by such Credit Document or Credit Documents.
- (h) Borrower or any Guarantor shall (i) apply for or consent to the appointment of a receiver, trustee, custodian, intervenor or liquidator of itself or of all or a substantial part of its assets, (ii) file a voluntary petition in bankruptcy or admit in writing that it is unable to pay its debts as they become due, (iii) make a statutory general assignment for the benefit of creditors, (iv) file a petition or answer seeking reorganization of an arrangement with creditors or to take advantage of any bankruptcy or insolvency laws, (v) file an answer admitting the material allegations of, or consent to, or default in answering, a petition filed against it in any bankruptcy, reorganization or insolvency proceeding, or (vi) take corporate action for the purpose of effecting any of the foregoing.
- (i) An involuntary petition or complaint shall be filed against Borrower or any Guarantor, seeking bankruptcy or reorganization of Borrower or any Guarantor, or the appointment of a receiver, custodian, trustee, intervenor or liquidator of Borrower or any Guarantor, or all or substantially all of its assets, and such petition or complaint shall not have been dismissed within sixty (60) days of the filing thereof; or an order, order for relief, judgment or decree shall be entered by any court of competent jurisdiction or other competent authority approving a petition or complaint seeking reorganization of Borrower, appointing a receiver, custodian, trustee, intervenor or liquidator of Borrower, or all or substantially all of its assets, and such order, judgment or decree shall continue unstayed and in effect for a period of sixty (60) days.
- (j) Any final judgment(s) (excluding those the enforcement of which is suspended pending appeal) for the payment of money in excess of the sum of \$1,000,000.00 in the aggregate (other than any judgment covered by insurance where coverage has been acknowledged by the insurer) shall be rendered against Borrower or any Guarantor, and such judgment or judgments shall not be satisfied, settled, bonded or discharged at least ten (10) days prior to the date on which any of its assets could be lawfully sold to satisfy such judgment.
- (k) Either (i) proceedings shall have been instituted to terminate, or a notice of termination shall have been filed with respect to, any Plans (other than a Multi-Employer Pension Plan as that term is defined in Section 4001(a)(3) of ERISA) by Borrower, any member of the Controlled Group, PBGC or any representative of any thereof, or any such Plan shall be terminated, in each case under Section 4041 or 4042 of ERISA, and such

termination shall give rise to a liability of the Borrower or the Controlled Group to the PBGC or the Plan under ERISA having an effect in excess of \$1,000,000.00 or (ii) a Reportable Event, the occurrence of which would cause the imposition of a lien in excess of \$1,000,000.00 under Section 4062 of ERISA, shall have occurred with respect to any Plan (other than a Multi-Employer Pension Plan as that term is defined in Section 4001(a)(3) of ERISA) and be continuing for a period of sixty (60) days.

- (1) Any of the following events shall occur with respect to any Multi-Employer Pension Plan (as that term is defined in Section 4001(a)(3) of ERISA) to which Borrower contributes or contributed on behalf of its employees and the Banks determine in good faith that the aggregate liability likely to be incurred by Borrower, as a result of any of the events specified in Subsections (i), (ii) and (iii) below, will have an effect in excess of \$1,000,000.00; (i) Borrower incurs a withdrawal liability under Section 4201 of ERISA; (ii) any such plan is "in reorganization" as that term is defined in Section 4241 of ERISA; or (iii) any such Plan is terminated under Section 4041A of ERISA.
- (m) The occurrence of a Change in Control or a material change in the management of the Borrower without the written consent of the Required Banks.
- (n) The dissolution, liquidation, sale, transfer, lease or other disposal of all or substantially all of the assets or business of Borrower or any Guarantor.
- (o) Any attachment, garnishment, levy or execution upon, or judicial seizure of, any property of Borrower that has a fair market value in excess of \$1,000,000.00, that is not bonded over within ten (10) days of notice thereof and whose termination is not diligently pursued thereafter.
- (p) The institution of any legal action or proceedings to enforce a lien or security interest in any property of Borrower or any Guarantor that has a fair market value in excess of \$1,000,000.00 that is not bonded over within ten (10) days of notice thereof and whose termination is not diligently pursued thereafter.
- (q) The occurrence of any material adverse change in the financial condition of Borrower that results in a Material Adverse Effect, or if the Banks in good faith shall believe that the prospect of payment or performance of the Obligations is impaired.
- 9.2 REMEDIES UPON EVENT OF DEFAULT. If an Event of Default shall have occurred and be continuing, then the Administrative Agent, at the request of the Required Banks may, at their sole option, exercise any one or more of the following rights and remedies, and any other remedies provided in any of the Credit Documents, as the Required Banks in their sole discretion may deem necessary or appropriate, all of which remedies shall be deemed cumulative, and not alternative:

-49-

- (i) Cease making Advances or extensions of financial accommodations in any form to or for the benefit of Borrower and declare the principal of, and all interest then accrued on, the Notes and any other liabilities hereunder to be forthwith due and payable, whereupon the same shall become immediately due and payable without presentment, demand, protest, notice of default, notice of acceleration or of intention to accelerate or other notice of any kind all of which Borrower hereby expressly waives, anything contained herein or in the Notes to the contrary notwithstanding;
  - (ii) Reduce any claim to judgment;
- (iii) Without notice of default or demand, pursue and enforce any of the Banks' and the Administrative Agent's rights and remedies under the Credit Documents, or otherwise provided under or pursuant to any applicable law or agreement; or
- (iv) Require Borrower to deposit with the Administrative Agent cash in an amount equal to the Outstanding LC Balance;

provided, however, that if any Event of Default specified in Sections 9.1(g) and 9.1(h) hereof shall occur, the principal of, and all interest on, the Notes and other liabilities hereunder shall thereupon become due and payable concurrently therewith, without any further action by the Banks and without presentment, demand, protest, notice of default, notice of acceleration or of intention to accelerate or other notice of any kind, all of which Borrower hereby expressly waives.

Upon the occurrence and during the continuance of any Event of Default, the Administrative Agent on behalf of the Banks, at the request of the Required Banks is hereby authorized at any time and from time to time, without notice to Borrower (any such notice being expressly waived by Borrower), to set off and

apply any and all moneys, securities or other property of Borrower and the proceeds therefrom, now or hereafter held or received by or in transit to the Banks or their agents, from or for the account of Borrower, whether for safe keeping, custody, pledge, transmission, collection or otherwise, and also upon any and all deposits (general or special) and credits of Borrower, and any and all claims of Borrower against the Banks at any time existing. The Banks agree to notify Borrower promptly after any such setoff and application, provided that the failure to give such notice shall not affect the validity of such setoff and application. The rights of the Banks under this Section are in addition to other rights and remedies (including, without limitation, other rights of setoff) which the Banks may have.

Upon the occurrence and during the continuance of any Event of Default, the Banks are also authorized to disclose such Event of Default in reference to an inquiry from any Person.

9.3 PERFORMANCE BY THE BANKS. Upon the occurrence and during the continuance of any Event of Default, should Borrower fail to perform any covenant, duty or agreement with respect to the payment of taxes, obtaining licenses or permits, or any other requirement contained herein or in any of the Credit Documents within the period provided herein, if any, for correction of

-50-

such failure, the Banks may, at their option, perform or attempt to perform such covenant, duty or agreement on behalf of Borrower. In such event, Borrower shall, at the request of the Required Banks, promptly pay any amount expended by the Banks and/or the Administrative Agent in such performance or attempted performance to the Administrative Agent at its Payment Office, together with interest thereon at the Default Rate, from the date of such expenditure until paid. Notwithstanding the foregoing, it is expressly understood that neither the Banks nor the Administrative Agent assume any liability or responsibility for the performance of any duties of Borrower hereunder or under any of the Credit Documents or other control over the management and affairs of Borrower.

9.4 NATURE OF THE OBLIGATIONS OF BORROWER. Except to the extent secured pursuant to Section 4.1, the obligations of Borrower hereunder shall be deemed PARI PASSU with the obligations of Borrower under the Senior Notes.

-51-ARTICLE 9A

## BANKS; ADMINISTRATIVE AGENT; ISSUING BANK

9A.1 APPOINTMENT. In order to expedite the transactions contemplated by this Credit Agreement, CB&T, a national banking association, is hereby appointed to act as Administrative Agent on behalf of the Banks. Each of the Banks, and each subsequent holder of any Note by its acceptance thereof, hereby irrevocably authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are specifically delegated to the Administrative Agent by the terms and provisions hereof and of the other Credit Documents, together with such actions and powers as are reasonably incidental thereto. The Administrative Agent is hereby expressly authorized by the Banks, without hereby limiting any implied authority, (a) to receive on behalf of the Banks all payments of principal of and interest on the Loans and all other amounts due to the Banks hereunder, and promptly to distribute to each Bank its proper share of each payment so received in accordance with this Credit Agreement; (b) to give written notice on behalf of each of the Banks to the Borrower of any Potential Default or Event of Default specified in this Credit Agreement of which the Administrative Agent has actual knowledge acquired in connection with its agency hereunder; and (c) to distribute to each Bank copies of all written notices, financial statements and other materials delivered by the Borrower pursuant to this Credit Agreement as received by the Administrative Agent.

9A.2 LIABILITY. Neither the Administrative Agent nor any of its directors, officers, employees or agents shall be liable as such for any action taken or omitted by any of them except for its or his own gross negligence or wilful misconduct, or be responsible for any statement, warranty or representation herein or the contents of any document delivered in connection herewith, or be required to ascertain or to make any inquiry concerning the performance or observance by the Borrower of any of the terms, conditions, covenants or agreements contained in any Loan Document. The Administrative Agent shall not be responsible to the Banks or the holders of the Notes for the due execution, genuineness, validity, enforceability or effectiveness of this Credit Agreement, the Notes or any other Credit Documents or other instruments or agreements. The Administrative Agent may deem and treat the payee of any Note as the owner thereof for all purposes hereof until it shall have received from the payee of such Note written notice, given as provided herein, of the transfer thereof. The Administrative Agent shall in all cases be fully protected in acting, or refraining from acting, in accordance with written instructions signed by either the Majority Banks or the Required Banks (if the consent of only the Majority Banks or the Required Banks is required by the provisions of this Credit Agreement with respect to an issue) or all the Banks (if the consent of all the Banks is required by the provisions of this Credit Agreement with respect to an issue), as applicable, and, except as otherwise specifically provided herein, such instructions and any action or inaction pursuant thereto shall be binding

on all the Banks and each subsequent holder of any Note. The Administrative Agent shall, in the absence of knowledge to the contrary, be entitled to rely on any instrument or document believed by it in good faith to be genuine and correct and to have been signed or sent by the proper Person or Persons. Neither the Administrative Agent nor any of its directors, officers, employees or agents shall have any responsibility to the Borrower on account of the failure of or delay in performance or breach by any Bank of any of its obligations hereunder

-52

or to any Bank on account of the failure of or delay in performance or breach by any other Bank or the Borrower of any of their respective obligations hereunder or under any other Loan Document or in connection herewith or therewith. The Administrative Agent may execute any and all duties hereunder by or through agents or employees and shall be entitled to rely upon the advice of legal counsel selected by it with respect to all matters arising hereunder and shall not be liable for any action taken or suffered in good faith by it in accordance with the advice of such counsel. Notwithstanding anything to the contrary, none of the foregoing provisions in this Section 9A.2 shall be applicable to Administrative Agent when acting in its capacity as a Bank.

#### 9A.3 ACTION BY ADMINISTRATIVE AGENT.

- (a) The Banks hereby acknowledge that the Administrative Agent shall be under no duty to take any discretionary action permitted to be taken by it pursuant to the provisions of this Credit Agreement unless it shall be requested in writing to do so by the Majority Banks. The Banks agree that, to the extent the Borrower must obtain the consent of any or all of the Banks pursuant to this Credit Agreement, such consent shall be deemed to have been given unless the Banks from whom such consent must be obtained fail to deny such consent within ten (10) days after the Borrower has requested such consent.
- (b) Unless in each case consented to in writing by all the Banks, the Administrative Agent shall not (i) agree to the modification or waiver of any of the terms of any of the Credit Documents, or (ii) consent to any act or omission by the Borrower, or (iii) exercise any rights which the Administrative Agent may have with respect to the Loans, the Notes, or any of the other Credit Documents, if any such agreement, modification, waiver, consent or exercise would:
  - (i) change or modify the interest rate and repayment provisions set forth in the Credit Documents;
    - (ii) increase the RLC Commitment;
    - (iii) extend the Maturity Date of the Loans;
  - (iv) postpone any date for payment or forgive the payment of principal of, or interest on, the Loans or the payment of any other sum due under the Credit Documents;
    - (v) waive or change any Event of Default;
  - (vi) allow any assignment by Borrower of any right or interest in the Credit Documents;
    - (vii) change or modify the Fees or the payment of such Fees; or
    - (viii) amend this Section 9A.3.

-53-

- (c) Upon receipt of an Advance notice from the Borrower, the Administrative Agent shall provide to each Bank a telecopy notice of such Advance (or telephone notice promptly confirmed by telecopy) not later than 9:30 a.m., San Diego, California local time, one Business Day before a proposed Advance. (d) The Administrative Agent agrees to distribute to each Bank by 9:00 a.m., San Diego, California local time on the immediately following Business Day, its pro rata share of each payment or prepayment of principal of any Loan, each payment of interest on the Loans, and each payment of the Fees that is received from the Borrower prior to 3:00 p.m., San Diego, California local time. Any such payments received after 3:00 p.m., San Diego, California local time, on any Business Day shall be made available to the Banks on or before 1:00 p.m., San Diego, California local time, on the immediately following Business Day.
- (e) The Administrative Agent agrees to distribute promptly to each Bank a copy of all information received from the Borrower and all amendments and modifications of the Credit Documents.
- (f) The Administrative Agreement agrees to distribute or cause to be distributed no later than thirty (30) days after the Closing Date to each Bank a copy of the Credit Documents.
- 9A.4 RESIGNATION. The Administrative Agent may not, without the written consent of the Banks and the Borrower which consent shall not be unreasonably

withheld or delayed, resign at any time. Upon receiving such consent, and subject to giving 30 days' prior written notice to the Banks, the Administrative Agent may resign as Administrative Agent hereunder. Upon any such resignation, the Majority Banks, with the written consent of the Borrower (which consent shall not be unreasonably withheld or delayed), shall have the right to appoint from the Banks a successor. If no successor shall have been so appointed by the Majority Banks and shall have accepted such appointment within 30 days after the Administrative Agent gives written notice of its resignation, then the Administrative Agent may, on behalf of the Banks, with the written consent of the Borrower (which consent shall not be unreasonably withheld or delayed) appoint a successor Administrative Agent which shall be a bank having a combined capital and surplus of at least \$50,000,000.00 or an Affiliate of any such bank. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor bank, such successor shall succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative Agent and such retiring Administrative Agent shall be discharged from its duties and obligations hereunder. After the Administrative Agent's resignation hereunder, the provisions of this Article shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as an Administrative Agent.

9A.5 AGENT AS BANK. With respect to the Loans made by it hereunder and the Notes issued to it, the Administrative Agent in its individual capacity and not as an Administrative Agent shall have the same rights and powers as any other Bank and may exercise the same as though it were not the Administrative Agent, and the Administrative Agent and its Affiliates may accept deposits from, lend money to and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if it were not the Administrative Agent.

-54-

9A.6 OWNERSHIP AND POSSESSION OF CREDIT DOCUMENTS. Each Bank shall own an undivided interest in the Advances and the Credit Documents equal to its Commitment from time to time. The Administrative Agent shall hold the Credit Documents in its possession, as agent, at its office at 1162 El Camino Road, Suite 200, San Diego, California 92130, or at such other location as the Administrative Agent shall designate in writing to the Banks, for the pro-rata benefit of itself as one of the Banks and each of the other Banks; PROVIDED, HOWEVER, that the Administrative Agent shall deliver to each Bank an original promissory note executed by the Borrower and evidencing such Bank's Commitment. The Administrative Agent shall keep and maintain complete and accurate files and records of all matters pertaining to the Advances. Upon reasonable prior notice to the Administrative Agent by a Bank, the files and records shall be made available to such Bank and its respective representatives and agents for inspection and copying during normal business hours.

9A.7 INDEMNIFICATION OF ADMINISTRATIVE AGENT. Each Bank agrees (i) to reimburse the Administrative Agent, on demand, in the amount of its pro rata share (based on its Commitment hereunder) of any expenses incurred for the benefit of the Banks by the Administrative Agent, including counsel fees and compensation of agents and employees paid for services rendered on behalf of the Banks, which shall not have been reimbursed by the Borrower and (ii) to indemnify and hold harmless the Administrative Agent and any of its directors, officers, employees or agents, on demand, in the amount of such pro rata share, from and against any and all liabilities, taxes, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may be imposed on, incurred by or asserted against it in its capacity as the Administrative Agent or any of them in any way relating to or arising out of this Credit Agreement or any other Loan Document or any action taken or omitted by it or any of them under this Credit Agreement or any other Loan Document, to the extent the same shall not have been reimbursed by the Borrower; PROVIDED that no Person shall be liable to the Administrative Agent for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting from the gross negligence or wilful misconduct of the Administrative Agent any of its directors, officers, employees or agents.

9A.8 INDEPENDENT CREDIT ANALYSIS. Each Bank acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Bank and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Credit Agreement. Each Bank also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Bank and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Credit Agreement or any other Loan Document, any related agreement or any document furnished hereunder or thereunder.

# ${\tt 9A.9}$ PROCESS FOR OBTAINING APPROVAL OF THE BANKS.

(a) With respect to obtaining the consent, approval, or determination of all of the Banks, of the Majority Banks or of the Required Banks, the Administrative Agent or any Bank may request that the Banks make a determination pursuant to this Credit Agreement. In the case of a request by any such Bank, the request shall be made through the Administrative

-55-

Administrative Agent to the Banks requesting Banks' determination, consent, approval, disapproval and/or joinder shall:

- (i) Be given in the form of a written notice to each of the Banks:
- (ii) Be accompanied by a description of the matter or thing as to which such determination, approval, consent, disapproval or joinder is requested, and shall advise each of the Banks where such matter or thing may be inspected, or shall otherwise adequately describe the matter or issue to be resolved;
- (iii) Include, to the extent not previously provided to the Banks, all written materials (to the extent necessary to make an informed decision) and a description of all oral information (to the extent necessary to make an informed decision) provided to the Administrative Agent in respect of the matter or issue to be resolved; and
- (iv) Include such other information and recommendations as the Administrative Agent may reasonably deem appropriate.
- (b) Subject to Paragraph (c) of this Section, the Banks shall reply within fifteen (15) Business Days after such written notice is given by the Administrative Agent; PROVIDED, HOWEVER, that if the Administrative Agent notifies the Banks that, pursuant to the Credit Documents, the matter with respect to which such consent, approval, disapproval or joinder is sought requires that the Administrative Agent respond within a certain time period and/or provides that if a response is not given within a certain time period such approval or consent shall be deemed given, the Banks shall reply by the earlier of (i) three (3) Business Days before such time period expires (as designated by the Administrative Agent) or (ii) five (5) Business Days after such written notice is given by the Administrative Agent.
- (c) With respect to each Bank, unless a Bank shall give written notice to the Administrative Agent that such Bank does consent to or approve any matter as to which such Bank's consent or approval is sought within the applicable time frame, such Bank shall be deemed to have disapproved of and not consented to such recommendation or determination.
- 9A.10 COMMUNICATIONS TO THE BANKS. All communications from the Borrower to the Banks relating to the Credit Documents and the Advances shall be sent by or through the Administrative Agent, except that Borrower may forward directly to the Banks copies of all notices and information sent by Borrower to the Administrative Agent.
- 9A.11 RELATIONSHIP WITH THE BORROWER. Consistent with the agency established hereunder, the Banks acknowledge and agree that the Administrative Agent, in accordance with its respective rights and duties under the Credit Documents, shall have the sole and exclusive authority to bind the Administrative Agent and the Banks with respect to matters relating to the Credit Documents. To the extent that any matter has been approved by all of the Banks, by the Required Banks or by the Majority Banks in accordance with the provisions of this Credit Agreement, the Administrative Agent is authorized to

-56-

execute such documents and instruments as the Administrative Agent may deem prudent to evidence and confirm such approval. The Borrower shall be entitled to rely upon any approval, communication or written notice it receives from the Administrative Agent as if it had been received from all of the Banks.

## 9A.12 PAYMENTS TO OR BY THE BANKS.

- (a) The Banks shall be entitled to interest on the amount of Advances held by each Bank for the period of time such Advances are outstanding at the rates set forth in this Credit Agreement, to the extent that such payments are actually received from the Borrower. If permitted pursuant to Section 2.4 of this Credit Agreement, the Administrative Agent shall charge and collect interest at the Default Rate unless the Majority Banks otherwise agree.
- (b) Fees, to the extent applicable, shall be paid to the Banks in accordance with Section 3.2.
- (c) Amounts paid by the Borrower pursuant to any provision of the Credit Documents providing for payment, compensation, or reimbursement to one or more, but not necessarily all, of the Banks, shall be paid to the Bank or Banks incurring such expenses or otherwise entitled to compensation under any of those provisions, with each Bank entitled to receive any payment, reimbursement, or compensation pursuant to any of such Sections or

other provisions being obligated to provide to the Administrative Agent and the Borrower a certificate setting forth in reasonable detail the basis for the amount of any request for compensation, payment or reimbursement under any of those Sections or other provisions. Except if incurred with respect to any Event of Default or explicitly provided for herein, Borrower shall not be liable for the expenses, including without limitation, the legal expenses, of the Banks.

- (d) Regular monthly payments of interest and any other payments to the Administrative Agent on behalf of the Banks (other than payments to be applied to the outstanding principal amount of Advances, which payments will be applied as provided in Section 9A.13), received by the Administrative Agent shall be made available to the Banks entitled thereto in accordance with Section 9A.3(d).
- (e) If and to the extent any Bank shall not have made any payment required pursuant to Article II, such Bank agrees to pay the Administrative Agent, forthwith on demand, such amount, together with interest thereon at the Federal Funds Rate, for each day from such date until the date such amount is paid to the Administrative Agent as provided in Article II. The failure of any Bank to make available to the Administrative Agent any amount required pursuant to Section 2.4 shall not relieve any other Bank of its obligation hereunder to make available as aforesaid such payment, as specified above, nor shall any Bank be relieved of its obligations to make such payments for any other reason.
- (f) Funds shall be transferred to the Banks in accordance with the funds transfer instructions given to the Administrative Agent and by the Administrative Agent to the Banks from time to time on or before the times specified in Section 9A.3(d).

-57-

- (g) If and to the extent the Administrative Agent shall not have made any payment required pursuant to Section 9A.3(d) to a Bank, the Administrative Agent agrees to pay such Bank forthwith on demand, such amount, together with interest thereon at the Federal Funds Rate, for each day from such date until the date such amount is paid pursuant to Section 9A.3(d).
- 9A.13 APPLICATION OF PAYMENTS. All monies collected or received by the Administrative Agent on account of the Loans or in respect of security for the Loans, directly or indirectly, shall be applied in the following order of priority, except to the extent otherwise required by Article II of this Credit Agreement, in which case the provisions of Article II shall control:
  - (a) To the payment of all costs and expenses due to the Administrative Agent and/or the Banks pursuant to the Credit Documents, including costs incurred in collection of such monies, including, without limitation, the payment to the Banks of the amounts described in Section 9A.12(d);
  - (b) To outstanding interest on the Loans, which amount shall be allocated between the Banks in accordance with the actual principal amount of Loans held by each Bank throughout the period in question as determined by the Administrative Agent on a daily basis; PROVIDED, HOWEVER, that if amounts received by the Administrative Agent are not sufficient to pay in full all such outstanding interest on the Loans, such amount shall be allocated among the Banks pro rata in accordance to the amount of Loans held by each Bank during the period in question; and
  - (c) To the payment of principal on the Loans in accordance with the principal amount of Loans held by each Bank.

## 9A.14 DEFAULTING BANKS.

(a) If for any reason any of the Banks shall fail or refuse to abide by its obligations under the Credit Documents (each a "Defaulting Bank"), then, in addition to the rights and remedies that may be available to the Administrative Agent and the other Banks at law and in equity, but subject to the notice and cure periods hereinafter set forth, such Defaulting Bank's right to participate in the administration of the Loans and the Credit Documents, including without limitation, any rights to consent to or direct any action or inaction of the Administrative Agent, all of the Banks including without limitation pursuant to Section 9A.3(b), or to be taken into account in the calculation of the Required Banks or the Majority Banks (other than the right to vote with respect to a decision as to its Loans to extend the Maturity Date thereof, or to amend the interest rate and repayment provisions thereof or to modify such Bank's Commitment), shall be suspended during the pendency of such failure or refusal. A Bank shall be deemed to be a Defaulting Bank if (i) such Bank shall have failed to pay to the Administrative Agent any amount due pursuant to this Credit Agreement within five (5) Business Days after written notice by the Administrative Agent to such Bank stating such payment is due from such Bank to the Administrative Agent; (ii) such Bank shall have failed to perform any of its other obligations under this Credit Agreement or the Credit Documents in any material respect, and such failure shall not have been cured within

30 days after written notice by the Administrative Agent to such Bank of such failure, or if such failure cannot reasonably be cured within such 30 day period, within such longer period of time as may be necessary to

-58-

complete such cure, so long as such Bank commences such cure within such 30-day period and thereafter diligently pursues such cure to completion within not more than 120 days after such written notice; or (iii) such Bank shall institute or be subject to any bankruptcy, insolvency, receivership, conservatorship, reorganization, liquidity or similar proceedings under state or federal law; provided, however, in the case of a failure described in clause (i) or clause (ii) of this sentence, if within the 5-Business Day period described in clause (i) or the 30 day period described in clause (ii), as applicable, the Bank in question in good faith disputes such default asserting that such default has not occurred (and provided that such Bank has satisfied its funding obligations pursuant to the provisions of Section 2.4), such Bank shall not be deemed to be a Defaulting Bank until such Bank is found to be in default pursuant to a final judicial or arbitration determination and such Bank does not thereafter take the action necessary to cure the default within 10 Business Days following the date of the final determination.

- (b) With respect to each Defaulting Bank, any Current Bank shall, in addition to any other rights or remedies available at law or equity, be entitled, in the case of the failure of a Defaulting Bank to pay its pro rata share (the "Defaulting Bank's Share") in a Loan made pursuant to Section 2.4, to pay to the Administrative Agent the Defaulting Bank's Share (pro rata if made by more than one Current Bank, based on the pro rata shares of the Current Banks making the payment). If one or more of the Current Banks pays the Defaulting Banks' Share, in addition to any other rights and remedies available to the Banks, each Current Bank making such payment may elect to do either of the following with respect to the payment made by such Current Bank:
  - (i) Notify the Administrative Agent to adjust the pro rata shares of the Defaulting Bank and the Current Bank making payment of the Defaulting Bank's Share, allocating the Defaulting Bank's Share to the Current Bank as of the date the Loan was made; or
  - (ii) Receive all amounts which the Defaulting Bank would otherwise be entitled to receive pursuant to this Credit Agreement with respect to the Defaulting Bank's Share (including interest accruing under the Credit Documents on the Loan, to the extent of the Defaulting Bank's Share of such Loan), pro rata according to the portion of the Defaulting Bank's Share paid by such Current Bank, until such Current Bank has been repaid the full amount of the Defaulting Bank's Share, together with accrued interest paid by the Borrower under this Credit Agreement with respect thereto.

9A.15 PURCHASE OF DEFAULTING BANK'S INTEREST AFTER DEFAULT. If Bank becomes a Defaulting Bank under Section 9A.14, each Bank which is not a Defaulting Bank (a "Current Bank") shall have the right, but not the obligation, in its sole discretion to acquire (or if more than one Current Bank exercises such right, each such Current Bank shall have the right to acquire, pro rata according to its pro rata shares, or in such other proportions as they may mutually agree), the interest in the Commitment and the Loans of a Defaulting Bank. Upon any such purchase, the Defaulting Bank's interest in the Commitment and the Loans and its rights hereunder as a Bank (but not its liability in respect thereof or under the Credit Documents or is Credit Agreement for events occurring prior to such purchase) shall terminate at the date of purchase, and the Defaulting Bank shall

-59-

promptly execute all documents reasonably requested to surrender and transfer such interest including an Assignment and Acceptance agreement and the canceled Note shall be returned to the Borrower. Current Banks exercising purchase rights under this Credit Agreement must, as a precondition to the exercise of such rights, concurrently exercise their corresponding purchase rights under this Credit Agreement.

- 9A.16 PURCHASE PRICE AND PAYMENT FOR DEFAULTING BANK'S INTEREST. The purchase price for the interest in the Commitment and the Loans of a Defaulting Bank shall be equal to the total outstanding Loans owed by the Borrower to the Defaulting Bank as of the date of such purchase, including without limitation any outstanding interest related thereto up to the date of such purchase, together with any accrued but unpaid Fees payable to the Defaulting Bank through such date. Payment of the purchase price for the Defaulting Bank's interest in the Commitment and the Loans so acquired shall be made on the date of such purchase.
- 9A.17 ISSUING BANK. The Issuing Bank shall act on behalf of the Banks with respect to any Letters of Credit Issued by it and the documents associated therewith until such time and except for so long as the Administrative Agent may agree at the request of all the Banks to act for such Issuing Bank with respect thereto; provided, however, that the Issuing Bank shall have all of the benefits and immunities (i) provided to the Administrative Agent in this Article 9A with

respect to any acts taken or omissions suffered by the Issuing Bank in connection with Letters of Credit Issued by it or proposed to be Issued by it and the application and agreements for letters of credit pertaining to the Letters of Credit as fully as if the term "Administrative Agent," as used in this Article 9A, included the Issuing Bank with respect to such acts or omissions, and (ii) as additionally provided in this Credit Agreement with respect to the Issuing Bank.

-60-ARTICLE 10.

#### MISCELLANEOUS

#### 10.1 SUCCESSORS AND ASSIGNS.

- (a) Whenever in this Credit Agreement any of the parties hereto is referred to, such reference shall be deemed to include the successors and assigns of such party; and all covenants, promises and agreements by or on behalf of the Borrower, the Administrative Agent or the Banks that are contained in this Credit Agreement shall bind and inure to the benefit of their respective successors and assigns.
- (b) Each Bank at its own expense may assign to one or more assignees all or a portion of its interests, rights and obligations under this Credit Agreement (including all or a portion of its Commitment, and the Loans at the time owing to it and the Notes held by it); PROVIDED, HOWEVER, that if a Bank is required to make such assignment due to a Regulatory Change, CB&T shall have the first right of refusal to purchase such assignment; otherwise (i) except in the case of an assignment to a Bank or an Affiliate of any Bank, the Borrower and the Administrative Agent must give their prior written consent to such assignment (which consent shall not be unreasonably withheld or delayed), (ii) each such assignment shall be of a constant, and not a varying, percentage of all the assigning Bank's rights and obligations under this Credit Agreement, (iii) except in the case of an assignment to a Bank or an Affiliate of any Bank, the amount of the Commitment of the assigning Bank subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to the Administrative Agent) shall not be less than \$10,000,000.00 or such lesser amount if such amount is the entire Commitment of the assigning Bank, (iv) the parties to each such assignment shall execute and deliver to the Administrative Agent an Assignment and Acceptance, together with the Note or Notes subject to such assignment and, except in the case of an assignment to a Bank or an Affiliate of any Bank, a processing and recordation fee of \$2,500 (which fee shall not in any way be the responsibility of the Borrower), (v) the assignee, if it shall not be a Bank, shall deliver to the Administrative Agent an Administrative Details Reply Form, (vi) any increased costs by reason of any such assignment will not be borne by the Borrower, and (vii) any assignee shall have a net worth of at least \$350,000,000 and total assets of a least \$2.5 billion. Upon acceptance and recording pursuant to paragraph (e) of this Section, from and after the effective date specified in each Assignment and Acceptance, which effective date shall be at least five Business Days after the execution there thereof, (A) the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Acceptance, have all the rights and obligations of a Bank under this Credit Agreement and (B) the assigning Bank thereunder shall, to the extent of the interest assigned by such Assignment and Acceptance, be released from its obligations under this Credit Agreement (and, in the case of an Assignment

-61-

and Acceptance covering all or the remaining portion of an assigning Bank's rights and obligations under this Credit Agreement, such Bank shall cease to be a party hereto (but shall continue to be entitled to the benefits of Sections 3.6, 3.7, 10.3 and 10.16 but only for the period prior to the effective date of the applicable Assignment and Acceptance, as well as to any Fees accrued for its account hereunder and not yet paid)).

(c) By executing and delivering an Assignment and Acceptance, the assigning Bank thereunder and the assignee thereunder shall be deemed to confirm to and agree with each other and the other parties hereto as follows: (i) such assigning Bank warrants that it is the legal and beneficial owner of the interest being assigned thereby free and clear of any adverse claim and that its Commitment and the outstanding balances of its Loans, without giving effect to assignments thereof which have not become effective, are as set forth in such Assignment and Acceptance, (ii) except as set forth in (i) above, such assigning Bank makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Credit Agreement, or the execution, legality, validity, enforceability, genuineness, sufficiency or value of this Credit Agreement, any other Credit Document or any other instrument or document furnished pursuant hereto or the financial condition of the Borrower or any Subsidiary or the performance or observance by the Borrower or any Subsidiary of any of its obligations under this Credit Agreement, any other Credit Document or any other instrument or document furnished pursuant

hereto; (iii) such assignee represents and warrants that it is legally authorized to enter into such Assignment and Acceptance; (iv) such assignee confirms that it has received a copy of this Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 7.1 and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance; (v) such assignee will independently and without reliance upon the Administrative Agent, such assigning Bank or any other Bank and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Credit Agreement; (vi) such assignee appoints and authorizes the Administrative Agent to take such action as agent on its behalf and to exercise such powers under this Credit Agreement as are delegated to the Administrative Agent by the terms hereof, together with such powers as are reasonably incidental thereto; and (vii) such assignee agrees that it will perform in accordance with their terms all the obligations which by the terms of this Credit Agreement are required to be performed by it as a Bank.

(d) The Administrative Agent shall maintain at one of its offices in San Diego, California, a copy of each Assignment and Acceptance delivered to it and a register for the recordation of the names and addresses of the Banks, and the Commitment of, and principal amount of the Loans owing to, each Bank pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive in the absence of manifest

-62-

error and the Borrower, the Administrative Agent and the Banks may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Bank hereunder for all purposes of this Credit Agreement. The Register shall be available for inspection by the Borrower and any Bank, at any reasonable time and from time to time upon reasonable prior written notice.

- (e) Upon its receipt of a duly completed Assignment and Acceptance executed by an assigning Bank and an assignee together with the Note or Notes subject to such assignment, an Administrative Details Reply Form completed in respect of the assignee (unless the assignee shall already be a Bank hereunder), the processing and recordation fee referred to in paragraph (b) above and, if required, the written consent of the Borrower and the Administrative Agent to such assignment, the Administrative Agent shall (i) accept such Assignment and Acceptance, (ii) record the information contained therein in the Register and (iii) give prompt written notice thereof to the Borrower and the Banks. Within five Business Days after receipt of notice, the Borrower, at its own expense, shall execute and deliver to the Administrative Agent, in exchange for the surrendered Note or Notes, a new Note or Notes to the order of such assigning Bank in a principal amount equal to the applicable Commitment retained by it. Such new Note or Notes shall be in an aggregate principal amount equal to the aggregate principal amount of such surrendered Note or Notes; such new Notes shall be dated the date of the surrendered Notes which they replace and shall otherwise be in substantially the form of Exhibit C. Canceled Notes shall be returned to the Borrower.
- (f) Each Bank may without the consent of the Borrower or the Administrative Agent sell participations to one or more banks or other entities in all or a portion of its rights and obligations under this Credit Agreement (including all or a portion of its Commitment and the Loans owing to it and the Notes held by it); PROVIDED, HOWEVER, that (i) such Bank's obligations under this Credit Agreement shall remain unchanged, (ii) such Bank shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) the participating banks or other entities shall be entitled to the benefit of the cost protection provisions contained herein to the same extent as if they were Banks (however no participating bank or entity shall be entitled to claim a greater amount than could have been claimed by the Bank from whom the participation was acquired) and (iv) the Borrower, the Administrative Agent and the other Banks shall continue to deal solely and directly with such Bank in connection with such Bank's rights and obligations under this Credit Agreement, and such Bank shall retain the sole right to enforce the obligations of the Borrower relating to the Loans and to approve any amendment, modification or waiver of any provision of this Credit Agreement. No entity acquiring a participation pursuant to this paragraph (f) shall by virtue of such participation have any direct voting rights under this Credit Agreement.

-63-

- (g) Any Bank or participant may, in connection with any assignment or participation or proposed assignment or participation pursuant to this Section, disclose to the assignee or participant or proposed assignee or participant any information relating to the Borrower furnished to such Bank by or on behalf of the Borrower.
- (h) Any Bank may at any time  $% \left( 1\right) =0$  assign all or any portion of its rights under this Credit Agreement and the Notes issued to it to a Federal Reserve

Bank; PROVIDED that no such assignment shall release a Bank from any of its obligations hereunder.

- (i) The Borrower shall not assign or delegate any of its rights or duties hereunder without the prior written consent of the Banks.
- 10.2 WAIVER. No failure to exercise, and no delay in exercising, on the part of the Banks, any right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise thereof preclude any other further exercise thereof or the exercise of any other right. The rights of the Banks and Administrative Agent hereunder and under the Credit Documents shall be in addition to all other rights provided by law. No modification or waiver of any provision of this Credit Agreement, the Notes or any Credit Documents, nor consent to departure therefrom, shall be effective unless in writing and no such consent or waiver shall extend beyond the particular case and purpose involved. No notice or demand given in any case shall constitute a waiver of the right to take other action in the same, similar or other instances without such notice or demand.

#### 10.3 PAYMENT OF EXPENSES; INDEMNITY.

- (a) Borrower shall pay all costs and expenses of (i) the Administrative Agent (including, without limitation, reasonable attorneys' fees and costs) incurred by the Administrative Agent in connection with the documentation of the Loans, and (ii) the Administrative Agent and the Banks in connection with the preservation and enforcement of rights of the Banks and the Administrative Agent under this Credit Agreement, the Notes, and/or the other Credit Documents. In addition, Borrower shall pay all costs and expenses of the Banks and the Administrative Agent in connection with the negotiation, preparation, execution and delivery of any and all amendments, modifications and supplements of or to this Credit Agreement, the Notes or any other Credit Document.
- (b) Borrower shall indemnify and hold the Banks and the Administrative Agent harmless from and against all claims, costs, expenses, actions, suits, proceedings, environmental claims or conditions, losses, damages and liabilities of any kind whatsoever, including but not limited to attorneys' fees and expenses, arising out of any matter relating, directly or indirectly, to the Loans, whether resulting from internal disputes of Borrower, or whether involving other third persons or entities, or out of any other matter whatsoever related to this Credit Agreement, the Credit Documents, but excluding any claim or liability which arises as the direct result of the gross negligence or willful misconduct of the Banks or the

-64-

Administrative Agent. This indemnity provision shall continue in full force and effect and shall survive not only the making of the Loans and the Advances but shall also survive the repayment of the Loans and the performance of all of Borrower's other obligations hereunder.

10.4 NOTICES. Except for telephonic notices permitted herein, any notices or other communications required or permitted to be given by this Credit Agreement or any other documents and instruments referred to herein must be (i) given in writing and personally delivered or mailed by prepaid certified or registered mail, or (ii) made by telefacsimile delivered or transmitted, to the party to whom such notice or communication is directed, to the address of such party as follows:

Borrower: MERITAGE CORPORATION

6613 North Scottsdale Road, Suite 200

Scottsdale, Arizona 85250

Attention: Larry W. Seay, Chief Financial Officer

Administrative Agent: CALIFORNIA BANK & TRUST

11622 El Camino Real, Suite 200 San Diego, California 92130 Attention: Peggy Standefer, Esq.

with a copy to: CB&T REAL ESTATE FINANCE

3101 North Central Avenue, Suite 520

Phoenix, Arizona 85012 Attention: Eileen Porter

Banks: See Schedule 1.1.

Any notice to be personally delivered may be delivered to the principal offices (determined as of the date of such delivery) of the party to whom such notice is directed. Any such notice or other communication shall be deemed to have been given (whether actually received or not) on the day it is personally delivered as aforesaid; or, if mailed, on the third day after it is mailed as aforesaid; or, if transmitted by telefacsimile, on the day that such notice is transmitted as aforesaid. Any party may change its address for purposes of this Credit Agreement by giving notice of such change to the other parties pursuant to this Section.

- 10.5 GOVERNING LAW. This Credit Agreement has been prepared, is being executed and delivered, and is intended to be performed in the State of Arizona. The substantive laws of the State of Arizona and the applicable federal laws of the United States of America shall govern the validity, construction, enforcement and interpretation of this Credit Agreement and all of the other Credit Documents, without regard to Arizona conflicts of law rules.
- 10.6 INVALID PROVISIONS. If any provision of any Credit Document is held to be illegal, invalid or unenforceable under present or future laws during the term of this Credit Agreement, such provision shall be fully severable; such Credit Document shall be construed and enforced as if such illegal, invalid or

-65-

unenforceable provision had never comprised a part of such Credit Document; and the remaining provisions of such Credit Document shall remain in full force and effect and shall not be affected by the illegal, invalid or unenforceable provision or by its severance from such Credit Document. Furthermore, in lieu of each such illegal, invalid or unenforceable provision there shall be added as part of such Credit Document a provision mutually agreeable to Borrower and the Banks as similar in terms to such illegal, invalid or unenforceable provision as may be possible and be legal, valid and enforceable.

- 10.7 BINDING EFFECT. The Credit Documents shall be binding upon and inure to the benefit of Borrower, the Banks and the Administrative Agent and their respective successors, assigns and legal representatives; provided, however, that Borrower may not, without the prior written consent of the Banks, assign any rights, powers, duties or obligations thereunder.
- 10.8 ENTIRETY. The Credit Documents embody the entire agreement between the parties and supersede all prior agreements and understandings, if any, relating to the subject matter hereof and thereof.
- 10.9 RELATIONSHIP OF THE BANKS AND BORROWER. The Banks and Borrower each have separate and independent rights and obligations under this Credit Agreement. Nothing contained herein shall be construed as creating, forming or constituting any partnership, joint venture, merger or consolidation of Borrower and the Banks for any purpose or in any respect.
- 10.10 TIME OF THE ESSENCE. Time is expressly made of the essence of this Credit Agreement.
- 10.11 GOOD FAITH STANDARD. Except where governed by a specific provision of this Credit Agreement for a specific purpose, whenever the approval or consent of the Banks is required hereunder, the Banks shall consider the request for approval or consent on a timely basis, but the Banks shall have such time as may be reasonably necessary to review and consider such request, as determined in their sole judgment, and the Banks shall have the right to not give their approval or consent or to impose such conditions or additional requirements with respect to their approval or consent as the Banks in their sole judgment shall determine. Approvals or consents by the Banks shall be effective only when given in writing, except when otherwise specifically provided herein. The standard by which the Banks shall be governed with respect to a request for approval or consent shall be "good faith" as that term is defined in the Arizona Uniform Commercial Code.
- 10.12 HEADINGS. Section headings are for convenience of reference only and shall in no way affect the interpretation of this Credit Agreement.
- 10.13 SURVIVAL. All representations and warranties made by Borrower herein shall survive delivery of the Notes and the making of the Loans.
- 10.14 NO THIRD PARTY BENEFICIARY. The parties do not intend the benefits of this Credit Agreement to inure to any third party, nor shall this Credit Agreement be construed to make or render the Banks and the Administrative Agent liable to any materialman, supplier, contractor, subcontractor, purchaser or

-66-

lessee of any property owned by Borrower, or for debts or claims accruing to any such persons against Borrower. Notwithstanding anything contained herein or in the Notes, or in any other Credit Document, or any conduct or course of conduct by any or all of the parties hereto, before or after signing this Credit Agreement or any of the other Credit Documents, neither this Credit Agreement nor any other Credit Document shall be construed as creating any right, claim or cause of action against the Banks and the Administrative Agent, or any of their officers, directors, agents or employees, in favor of any materialman, supplier, contractor, subcontractor, purchaser or lessee of any property owned by Borrower, nor to any other person or entity other than Borrower.

- 10.15 SCHEDULES AND EXHIBITS INCORPORATED. All schedules and exhibits attached hereto are hereby incorporated into this Credit Agreement by each reference thereto as if fully set forth at each such reference.
- 10.16 SETOFF. The Banks are hereby authorized at any time and from time to time, without prior notice to Borrower (any such notice being expressly waived by the Borrower), to set off and apply any and all deposits (general or special,

time or demand, provisional or final) at any time held by the Banks, or any branch, subsidiary, or affiliate of the Banks, and other indebtedness at any time owing by the Banks, or any branch, subsidiary, or affiliate of the Banks, to or for the credit or the account of Borrower, against any and all of the obligations of Borrower now or hereafter existing under the Credit Documents, irrespective of (i) whether or not the Banks shall have made any demand under the Credit Documents and (ii) whether such obligations are contingent, matured, or unmatured. The Banks agree promptly to notify Borrower after any such setoff and application, provided that the failure to give such notice shall not affect the validity or such setoff and application. The rights of the Banks under this Section are in addition to other rights and remedies (including, without limitation, other rights of setoff) which the Banks may have.

10.17 JURY WAIVER. THE BORROWER, THE BANKS AND THE ADMINISTRATIVE AGENT HEREBY VOLUNTARILY, KNOWINGLY, IRREVOCABLY AND UNCONDITIONALLY WAIVE ANY RIGHT TO HAVE A JURY PARTICIPATE IN RESOLVING ANY DISPUTE (WHETHER BASED UPON CONTRACT, TORT OR OTHERWISE) BETWEEN OR AMONG THE BORROWER AND THE BANKS AND THE ADMINISTRATIVE AGENT ARISING OUT OF OR IN ANY WAY RELATED TO THIS DOCUMENT, ANY OTHER CREDIT DOCUMENT, OR ANY RELATIONSHIP BETWEEN THE BANKS, THE ADMINISTRATIVE AGENT AND THE BORROWER. THIS PROVISION IS A MATERIAL INDUCEMENT TO THE BANKS TO PROVIDE THE FINANCING DESCRIBED HEREIN OR IN THE OTHER CREDIT DOCUMENTS.

10.18 COUNTERPARTS. This Credit Agreement may be executed in multiple counterparts, each of which, when so executed, shall be deemed an original but all such counterparts shall constitute but one and the same agreement.

-67-

IN WITNESS WHEREOF,  $\,$  the undersigned have executed this Credit Agreement as of the day and year first above written.

 $\begin{array}{ll} \texttt{MERITAGE CORPORATION, a Maryland} \\ \texttt{corporation} \end{array}$ 

Ву:	
Name:	
Its:	
	BORROWER
	RNIA BANK & TRUST, a California g corporation
Ву:	
Name:	
Its:	
	ADMINISTRATIVE AGENT, BANK

ADMINISTRATIVE AGENT, BANK
AND ISSUING BANK

SCHEDULE 2.1

COMMITMENTS OF THE BANKS as of September 17, 1999

			RLC
	BANK	왕	COMMITMENT
1.	California Bank & Trust	100.0%	\$15,000,000
	Maximum Commitment	100.0%	\$15,000,000

## ADDRESSES

 California Bank & Trust 11622 El Camino Real, Suite 200 San Diego, California 92130 Attention: Peggy Standefer, Esq.

with a copy to:

CB&T Real Estate Finance 3101 North Central Avenue, Suite 520 Phoenix, Arizona 85012 Attention: Eileen Porter

#### SCHEDULE 6.23

#### SUBSIDIARIES

PERCENTAGE OF VOTING STOCK OR EQUITY JURISDICTION OF INTEREST OWNED BY NAME OF INCORPORATION OR BORROWER AND EACH SUBSIDIARY FORMATION OTHER SUBSIDIARY \_\_\_\_\_ \_\_\_\_\_ Monterey Homes Construction I, Inc.\* Arizona 100% 100% Monterey Homes Arizona I, Inc.\* Arizona Monterey Homes Construction, Inc. Arizona 100% Monterey Homes Arizona, Inc. Arizona 100% 100% MTH-Texas GP, Inc. Arizona MTH-Texas LP, Inc. 100% Arizona 1% GP Arizona Legacy/Monterey Homes LP 99% LP 100% 100% Texas Home Mortgage Corporation Texas Meritage Homes of Northern California, Inc. California 100% EMIC Finance Corporation\* Arizona 100% Chandler 110, LLC Arizona Hulen Park Venture, LLC Texas 75%

SCHEDULE 6.24

EXISTING INDEBTEDNESS - BORROWER (as of June 30, 1999)

	COMMITMENT BALANCE		
LENDER	OUTSTANDING BALANCE	DUE DATE	COLLATERAL
Senior Subordinated Notes	\$15,000,000 \$15,000,000	10/15/2001	None-Unsecured
	SCHEDULE 6.25		

EXISTING INDEBTEDNESS - SUBSIDIARIES (as of June 30, 1999)

<TABLE> <CAPTION>

LENDER	COMMITMENT BALANCE OUTSTANDING BALANCE	DUE DATE	COLLATERAL
<s> Norwest/Bank One Revolving Construction Facility</s>	<c> \$60,000,000 \$23,155,000</c>	<c> 12/18/1998</c>	<c> Substantially all housing inventory in Arizona</c>
Norwest/Bank One A&D Guidance Facility	\$20,000,000 Guidance \$ 5,908,000	Various	Various finished lots and lots under development in Arizona and California
Guaranty Federal Revolving Construction Facility	\$70,000,000* \$25,268,000	7/31/1999	Substantially all housing inventory in Texas
ComPAS Bank	\$ 1,060,000 \$ 360,000	4/30/2000	91 Finished Lots in Brighton Subdivision
ComPAS Bank	\$ 3,480,000 \$ 2,064,000	7/13/2000	188 lots under development in Legends Crest Subdivision
Chase Automotive	\$ 32,832 \$ 32,832	9/04/2001	1997 Jaguar
Seller Carryback - Chandler 110 McQueen Riggs 60 LLC 			

 \$ 878,550 \$ 878,550 |  | 298 lots under development at Paseo Crossings ||  |  |  | (Chandler) |
<sup>\*</sup> Increased to \$80 million and extended to 7/31/00 during July, 1999 EXHIBIT "A-1"

COMPLIANCE CERTIFICATE FOR FISCAL QUARTER ENDING \_**,** 19 ("REPORTING QUARTER")

CALIFORNIA BANK & TRUST 11622 El Camino Real, Suite 200 San Diego, California 92130

<sup>\*</sup> Inactive and to be soon dissolved.

with a	сору	to:			
CB&T	Real	Estate	Finance		
3101	North	n Centra	al Avenue,	Suite	520
Phoe	nix, A	Arizona	85012		
Attei	ntion:	Eileer	Porter		

Date:	(1)

Re: Meritage Corporation, a Maryland corporation

Dear Ladies and Gentlemen:

This Compliance Certificate refers to the Credit Agreement dated as of September 17, 1999 (as it may hereafter be amended, modified, extended or restated from time to time, the "Credit Agreement"), among MERITAGE CORPORATION, a Maryland corporation ("Borrower"), the Banks named therein from time to time, and California Bank & Trust, a California banking corporation as Administrative Agent for the Banks. Capitalized terms used and not otherwise defined herein shall have the meanings assigned to such terms in the Credit Agreement.

Pursuant to Section 7.1(c) of the Credit Agreement, the undersigned, an Authorized Officer of Borrower, hereby certifies that:

- 1. Enclosed are the required financial statements for the quarter ending for the Borrower as required under Section 7.1 of the Credit Agreement, which, to the undersigned's knowledge, after due inquiry, fairly present in all material respects the financial position of the Borrower and the results of its operations at the dates and for the periods indicated, and have been prepared in accordance with GAAP.
- 2. To the best of the undersigned's knowledge, no "Event of Default" has occurred [or if so, specifying the nature and extent thereof and any corrective actions taken or to be taken].

<sup>4.</sup> As of the last day of the Reporting Quarter, the computations below were true and correct:

I.	Secti	on 8.10	(a) - LEVERAGE RATIO (CONSOLIDATED)	(in t	housands)
	Numer	ator:	Indebtedness (per GAAP)	\$	
			plus Subordinated Debt(2)	\$	
			plus current amounts due under development(2) agreements/funding obligations	\$	
			plus letters of credit(2)	\$	
			equals Consolidated Indebtedness	\$	A
			divided b		
	Denom	inator:	Consolidated stockholders equity	\$	
			less Consolidated Intangible Assets	\$(	)
			plus Tax Assets	\$	
			equals Consolidated Tangible Net Worth	\$	В
	Equal	s:	Leverage Ratio		A/B
			Maximu		1.75X
II.	Secti	on 8.10	(b) - CONSOLIDATED TANGIBLE NET WORTH		
	Α.	Actual		\$	
	В.	Require	ment: Actual as of March 31, 1999	\$	А
			90% of A	\$	В

<sup>(2)</sup> If not already included in Indebtedness.

<sup>(1)</sup> To be submitted within 60 days after the end of each of the first three fiscal quarters of each fiscal year and within 105 days after the end of the final fiscal quarter of each fiscal year.

<sup>3.</sup> To the best of the undersigned's knowledge, Borrower is in compliance with its most recent RLC Borrowing Base Inventory Report.

plus 50.0% of Net Profit for each fiscal year after March 31, 1999 equals Consolidated Tangible Net Worth Requirement (sum of B, C and D) Minimum: greater of \$45,000,000 or \$ Percentage of Consolidated Tangible Net Worth С. Nonentitled Land Entitled Land not under development Total % of Consolidated Tangible Net Worth 용 25% Maximum Entitled Land not under development (excluding Finished Lots) Entitled Developing Land (excluding Finished Lots) Ś Total % of Consolidated Tangible Net Worth 80% Maximum Nonentitled Land, Entitled Land, Entitled Developing Land, Finished Lots and Contract Option Deposits: % of Consolidated Tangible Net Worth Maximum Land supply (excludes Finished Lots under option contract) Supply \_\_\_\_ Years Maximum 3 years III. Section 8.10(c) - INTEREST COVERAGE RATIO (CONSOLIDATED, ROLLING FOUR QUARTERS) Numerator: Net Profit Before Tax plus interest expense and capitalized interest expensed in cost of sales plus depreciation plus amortization \$ equals EBITDA divided by Denominator: Consolidated Interest Incurred Equals: Interest Coverage Ratio Minimum 2.50X IV. Section 8.10(d) - FIXED CHARGE COVERAGE RATIO (CONSOLIDATED, ROLLING FOUR QUARTERS) Numerator: EBITDA plus rental and operating lease expenses equals divided by Denominator: Consolidated Interest Incurred \$ plus rental and operating lease expenses plus principal amortization payments plus distributions to preferred shareholders(3) less principal payments at loan maturity \$(

		less principal amorti Senior Notes	zation payments -	\$( )
		equals Fixed Charges		\$ B
_	,		5	
E	quals:	Fixed Charge Coverage	Ratio	A/B
		lividends and other distri	hutions	
(3) 1110	crudes (	lividends and other distri		
			-4- Minimum	1.50X
V. Se	ection 8	3.1(f) - SUBORDINATED DEBT		
*	Bala	nce at the end of the Rep	orting Quarter	\$
			Maximum	\$20,000,000
V	I. Sect	ion 8.8 - INVESTMENTS		
( 6	a) Joir	nt venture - aggregate inv	estments	\$
	*	% of Consolidated Tangib	le Net Worth	<del></del> %
			Maximum	
(]	B) Unre	elated business - aggregat	e investments	\$
	*	% of Consolidated Tangib		 ১
			Maximum	5.0%
			MERITAGE CORPORATION, a Mar corporation	yland
			Ву:	
			Name:	
			Its:	
			-5-	
	Ç	EXHI: QUARTERLY INVENTORY REPORT	BIT A-2 AS OF	
Re	e: MER	TAGE CORPORATION, a Maryl	and corporation	
(the "Control thereing "Administration the Book the dasserted the state of the stat	Credit An and ( istrative Credit orrower te set	agreement"), among the un- california Bank & Trust, re Agent"). Except as oth Agreement shall have the certifies to the Banks at forth herein, the state	Credit Agreement dated Septer dersigned, as Borrower, the a California banking corperwise provided herein, all same meaning when used in the nd the Administrative Agentements, representations and ached hereto are true and compared to the september of the septem	e Banks named pration (the terms defined is Report.  t that, as of d warranties
		of the day of		
			MERITAGE CORPORATION, a Mar corporation	yland
			By:	
			Name:	
			Title:	
		EXHIB	IT "B-1	
	RLC BO	RROWING BASE INVENTORY RE RE: MERITAGE CORPORATION	PORT AS OF MONTH ENDING N, A MARYLAND CORPORATION	

<TABLE> <CAPTION>

Inventory Total GAAP Cost (or units) Advance Rate

RLC

	Arizona	California	Texas	(Unencumbered)	(Or Max. %)	Base
 <5> <c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
LAND OWNED Entitled Land (owned 3-12 months) In \$					30%	
Entitled Land (owned 0-3 months) In \$					60%	
 Entitled Developing Land In \$					60%	
 Finished Lots In \$					60% 	
SUBTOTALS						\$
 DETACHED SF UNIT INVENTORY Models** in \$					80%	\$ 
 Spec*** in \$ (not to exceed					***	\$
lesser of \$20M						
or 30% of sale						
value of prior						
4 quarter						
<pre>backlog) Subtotal of Model/Spec. Inv. in \$</pre>						\$
 Backlog in \$ - Prev. Qtr. End						
Presold in \$					90%	
 Subtotals (GAAP Cost of Construction ar	nd RLC Borr	rowing Base)		\$		\$
 TOTAL GAAP COST & RLC BORROWING BASE Total RLC						
Borrowing Base Less:						
OTHER OUTSTANDING SENIOR UNSECURED DEBTOR OF CREDIT) (other than the Senior Notes		G ISSUED AND	UNDRAWN I	LETTERS		
a) Lender Name					le	ss:
b) Lender Name					le	ss:
c) Lender Name					le	ss:
SUBORDINATED DEBT IN EXCESS OF \$20,000,	000				le	ss:
ISSUED AND UNDRAWN LETTERS OF CREDIT U	INDER THE S	UBJECT FACII	ITY (UP TO	\$0)	le	ss:
RLC Borrowing Base Available for RLC Advaless issued & undrawn letters of credit). available						

  | \$15,000,000 | facility o | cap |  | \$ |

# Footnotes:

<sup>\*</sup>No more than \$8,000,000 Entitled Developing Land costs in any Subdivision.

<sup>\*\*</sup>No more than 6 Units per Subdivision in \$.

<sup>\*\*\*</sup>No more than 15 Units in California, 12 Units elsewhere per Subdivision in \$.

\*\*\*80% for 12 months after completion, 70% for next 6 months, 50% for next 6

months, 0% thereafter.

17, 1999 (the "Agreement") among the undersigned as Borrower, the Banks named therein and California Bank & Trust, a California banking corporation, the undersigned hereby certifies to the Administrative Agent that the foregoing information on this RLC Borrowing Base Inventory Report as of month ending \_ is true and correct and was provided from financial information prepared according to GAAP, consistently applied. All capitalized undefined terms used herein have the meaning given them in the Agreement. MERITAGE CORPORATION, a Maryland corporation By: Name: Title: \_\_\_\_\_ EXHIBIT "B-2" RIC BORROWING BASE COMPLIANCE CERTIFICATE as of month ending \_ Re: MERITAGE CORPORATION, a Maryland corporation Pursuant to Section 7.1(f) of that Credit Agreement dated as of September 17, 1999 (the "Agreement") among the undersigned, as Borrower, the Banks named therein, and California Bank & Trust, a California banking corporation (the "Administrative Agent"), the Borrower certifies as follows with respect to that RLC Borrowing Base Inventory Report of even date herewith (the "Report") that: 1. All Inventory shown on the Report is located in a Qualified State. 2. All Inventory shown on the Report is located in a Qualified Subdivision. 3. All Inventory shown on the Report is either owned by Borrower or by a Guarantor and is free of any Mortgage Lien, and for which the Guarantor or Borrower, as applicable, has received a Qualified Title Policy. 4. The sum of the  $\,$  Senior  $\,$  Unsecured  $\,$  Debt  $\,$  shown on the  $\,$  Report  $\,$  plus the outstanding letters of credit under the RLC plus Subordinated Debt in excess of \$20,000,000 and the outstanding RLC Advances is not more than the lesser of the Total RLC Borrowing Base shown on the Report or the RLC Commitment.

- 5. Borrower or, if applicable, the Subsidiary, has received lien waivers for all construction work in process with respect to the property.
- 6. No more than \$8,000,000 in Entitled Developing Land costs in any single Subdivision is included in the RLC Borrowing Base.
- 7. Entitled  $\,$  Land was added to the RLC  $\,$  Borrowing  $\,$  Base within the past 12 months.

The undersigned each hereby certifies to the Administrative Agent that the foregoing information on this RLC Borrowing Base Compliance Certificate as of month ending \_\_\_\_\_ is true and correct and was provided from financial information prepared according to GAAP, consistently applied. All capitalized undefined terms used herein have the meaning given them in the Agreement.

MERITAGE CORPORATION, a Maryland corporation

Зу:			
Name:			
-	 		
Title:			
		D	ODDOMED

BORROWER

-2-EXHIBIT "C"

REVOLVING PROMISSORY NOTE

Phoenix, Arizona

to to pay "Maker"), promises the order (the "Payee"; Payee and each subsequent transferee and/or owner of this Note, whether taking by endorsement or otherwise, are herein successively called "Holder"), at the office of CALIFORNIA BANK & TRUST, a California banking corporation (the "Administrative Agent"), at Central Note Department, 9775 Clairemont Mesa Boulevard, San Diego, California 92124, or at such other place as Holder may from time to time designate in writing, the principal sum of AND NO/100 ) or so much thereof as Holder may advance to or for the benefit of Maker plus interest calculated on a daily basis (based on a 360-day year) for the actual number of days elapsed from the date hereof on the principal balance from time to time outstanding as hereinafter provided, principal, interest and all other sums payable hereunder to be paid in lawful money of the United States of America as follows:

- A. Interest  $% \left( 1\right) =\left( 1\right) =\left( 1\right)$  shall accrue on the unpaid  $% \left( 1\right) =\left( 1\right) =\left( 1\right)$  principal of each RLC Advance at the Variable Rate.
  - B. Interest shall be due and payable on each Payment Date.
- C. The entire unpaid principal balance, all accrued and unpaid interest, and all other amounts payable hereunder shall be due and payable in full on the RLC Maturity Date.

The principal balance of this Note represents a revolving credit all or any part of which may be advanced to Maker, repaid by Maker, and re-advanced to Maker from time to time, subject to the other terms hereof and the conditions, if any, contained in the Credit Agreement, and provided that the principal balance outstanding at any one time shall not exceed the face amount hereof.

Maker agrees to an effective rate of interest that is the rate stated above plus any additional rate of interest resulting from any other charges in the nature of interest paid or to be paid by or on behalf of Maker, or any benefit received or to be received by Holder, in connection with this Note.

Payments are to be made no later than 11:00 a.m. California time on the date when due. If payment is ten (10) days or more late, Maker will be charged five percent (5.0%) of the regularly scheduled payment. This late charge may be assessed without notice, shall be immediately due and payable and shall be in addition to all other rights and remedies available to Holder.

Unless otherwise agreed to, in writing, or otherwise required by applicable law, payments will be applied first to accrued, unpaid interest, then to principal, and any remaining amount to any unpaid collection costs, late charges and other charges; provided, however, upon delinquency or other default, Holder reserves the right to apply payments among principal, interest, late charges, collection costs and other charges at its discretion. All prepayments shall be applied to the indebtedness owing hereunder in such order and manner as Holder may from time to time determine in its sole discretion.

This Note is issued pursuant to that Credit Agreement (the "Credit Agreement") dated as of September 17, 1999 among Maker, the Banks named therein from time to time and the Administrative Agent. The capitalized terms used herein and not otherwise defined shall have the same meanings as set forth in the Credit Agreement.

Time is of the essence of this Note. At the option of Holder, the entire unpaid principal balance, all accrued and unpaid interest and all other amounts payable hereunder shall become immediately due and payable without notice upon the occurrence of any Event of Default.

Upon an Event of Default, including failure to pay upon final maturity, Holder, at its option, may also, if permitted under applicable law, do one or both of the following: (a) increase the applicable interest rate on this Note three (3.00) percentage points, and (b) add any unpaid accrued interest to principal and such sum will bear interest therefrom until paid at the rate provided in this Note (including any increased rate). The interest rate will not exceed the maximum rate permitted by applicable law. Maker shall pay all costs and expenses, including reasonable attorneys' fees and court costs, incurred in the collection or enforcement of all or any part of this Note. In the event of any court proceedings, court costs and attorneys' fees shall be set by the court and not by jury and shall be included in any judgment obtained by Holder.

Maker shall have the option to prepay this Note, in full or in part, as provided in the Credit Agreement.

Failure of Holder to exercise any option hereunder shall not constitute a waiver of the right to exercise the same in the event of any subsequent default or in the event of continuance of any existing default after demand for strict performance hereof.

Maker and all sureties, guarantors and/or endorsers hereof (or of any obligation hereunder) and accommodation parties hereon (all of which, including Maker, are severally each hereinafter called a "Surety") each: (a) agree that the liability under this Note of all parties hereto is joint and several; (b)

severally waive any homestead or exemption laws and right thereunder affecting the full collection of this Note; (c) severally waive any and all formalities in connection with this Note to the maximum extent allowed by law, including (but not limited to) demand, diligence, presentment for payment, protest and demand, and notice of extension, dishonor, protest, demand and nonpayment of this Note; and (d) consent that Holder may extend the time of payment or otherwise modify the terms of payment of any part or the whole of the debt evidenced by this Note, at the request of any other person liable hereon, and such consent shall not alter nor diminish the liability of any person hereon.

-2-

In addition, each Surety waives and agrees not to assert: (a) any right to require Holder to proceed against Maker or any other Surety, to proceed against or exhaust any security for the Note, to pursue any other remedy available to Holder, or to pursue any remedy in any particular order or manner; (b) the benefit of any statute of limitations affecting its liability hereunder or the enforcement hereof; (c) the benefits of any legal or equitable doctrine or principle of marshalling; (d) notice of the existence, creation or incurring of new or additional indebtedness of Maker to Holder; (e) the benefits of any statutory provision limiting the liability of a surety, including without limitation the provisions of Sections 12-1641, ET SEQ., of the Arizona Revised Statutes; (f) any defense arising by reason of any disability or other defense of Maker or by reason of the cessation from any cause whatsoever (other than payment in full) of the liability of Maker for payment of this Note; and (g) the benefits of any statutory provision limiting the right of Holder to recover a deficiency judgment, or to otherwise proceed against any person or entity obligated for payment of this Note, after any foreclosure or trustee's sale of any security for this Note, including without limitation the benefits, if any, to a Surety of Arizona Revised Statutes Section 33-814. Until payment in full of this Note and Holder has no obligation to make any further advances of the proceeds hereof, no Surety shall have any right of subrogation and each hereby waives any right to enforce any remedy which Holder now has, or may hereafter have, against Maker or any other Surety, and waives any benefit of, and any right to participate in, any security now or hereafter held by Holder.

Maker agrees that to the extent Maker or any Surety makes any payment to Holder in connection with the indebtedness evidenced by this Note, and all or any part of such payment is subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid by Holder or paid over to a trustee, receiver or any other entity, whether under any bankruptcy act or otherwise (any such payment is hereinafter referred to as a "Preferential Payment"), then the indebtedness of Maker under this Note shall continue or shall be reinstated, as the case may be, and, to the extent of such payment or repayment by Holder, the indebtedness evidenced by this Note or part thereof intended to be satisfied by such Preferential Payment shall be revived and continued in full force and effect as if said Preferential Payment had not been made.

This Note shall be binding upon Maker and its successors and assigns and shall inure to the benefit of Payee, and any subsequent holders of this Note, and their successors and assigns.

All notices required or permitted in connection with this Note shall be given at the place and in the manner provided in the Credit Agreement for the giving of notices.

This Note shall be governed by and construed according to the laws of the State of Arizona, without giving effect to conflict of laws principles.

JURY WAIVER. THE UNDERSIGNED AND HOLDER (BY ITS ACCEPTANCE HEREOF) HEREBY VOLUNTARILY, KNOWINGLY, IRREVOCABLY AND UNCONDITIONALLY WAIVE ANY RIGHT TO HAVE

-3-

A JURY PARTICIPATE IN RESOLVING ANY DISPUTE (WHETHER BASED UPON CONTRACT, TORT OR OTHERWISE) BETWEEN OR AMONG THE UNDERSIGNED AND HOLDER ARISING OUT OF OR IN ANY WAY RELATED TO THIS DOCUMENT OR ANY OTHER AGREEMENTS, DOCUMENTS OR INSTRUMENTS EXECUTED OR DELIVERED IN CONNECTION WITH, OR OTHERWISE RELATING TO, THE INDEBTEDNESS EVIDENCED HEREBY (TOGETHER WITH THIS NOTE, THE "RELATED DOCUMENTS"). THIS PROVISION IS A MATERIAL INDUCEMENT TO HOLDER TO PROVIDE THE FINANCING DESCRIBED HEREIN OR IN THE OTHER RELATED DOCUMENTS.

IN WITNESS WHEREOF, these presents are executed as of the date first written above.

 $\begin{array}{ll} \texttt{MERITAGE CORPORATION, a Maryland} \\ \texttt{corporation} \end{array}$ 

Ву:				
	 	 	 	-
Name:				
	 	 	 	_
Its:				

-4-EXHIBIT "D"

#### SUBDIVISION ADMISSION CERTIFICATE

RE: MERITAGE CORPORATION,
A MARYLAND CORPORATION

Reference is made to that certain Credit Agreement dated September 17, 1999 (the "Credit Agreement"), among the undersigned, as Borrower, the Banks named therein, and California Bank & Trust, a California banking corporation (the "Administrative Agent"). Except as otherwise provided herein, all terms defined in the Credit Agreement shall have the same meaning when used in this Subdivision Eligibility Certificate.

Borrower certifies to the Banks and the Administrative Agent that, as of the date set forth herein, the following statements, representations and warranties are true and correct as to the Subdivision submitted to the Administrative Agent for approval as a Qualified Subdivision.

1.	ASSET DE	SCRIPTION	
	PMSA/Cit	у:	State:
	Subdivis	ion Name:	
2.	DUE DILI	GENCE COMPLETED (Y/N)	
		Soils and Engineering Rep	ort
		Approved Phase I Report o required)	r Environmental Questionnaire (if
		Approved Phase II Report	(if required)
		Remediation complet	ed by licensed environmental engineer
		Subdivision Plat Map appr	oved by applicable municipality
		Final	
		Preliminary	
		Title Insurance	
3.	LIENS AN	Owner D ENCUMBRANCES	
		No Liens and Encumbrances	in violation of the Credit Agreement
Date	d as of t	he day of	·
			MERITAGE CORPORATION, a Maryland corporation
			By:
			Name:
			Title:
			-2- BIT "E"
		CONTINUIN	C CHARANTEE

CONTINUING GUARANTEE

TO: California Bank & Trust as Administrative Agent for the Banks

1. For valuable consideration, the undersigned (hereinafter called "Guarantor"), whose address is set forth after Guarantor's signature below, jointly and severally, and unconditionally, guarantees and promises to pay to the Banks (hereinafter collectively called "Lender") listed in that Credit Agreement dated as of September 17, 1999 among the Administrative Agent, the Lender and the hereinafter defined Borrower, or order, on demand, in lawful money of the United States, any and all indebtedness of MERITAGE CORPORATION, a Maryland corporation (hereinafter called "Borrower") to Lender. If more than one Borrower is named herein, or if this Guarantee is executed by more than one Guarantor, the word "Borrower" and the word "Guarantor" respectively shall mean all and any one or more of them, severally and collectively. The word "indebtedness" is used in its most comprehensive sense and includes any and all advances, debts, obligations and liabilities of Borrower heretofore, now or

hereafter made, incurred or created, with or without notice to Guarantor, whether voluntary or involuntary and however arising, whether due or not due, absolute or contingent, liquidated or unliquidated, determined or undetermined, and whether Borrower is liable individually or jointly with others, or whether recovery upon such indebtedness may be or hereafter become barred by any statute of limitations, or whether such indebtedness may be or hereafter become otherwise unenforceable, exclusive, however, of any indebtedness of Borrower to Lender presently covered by existing guaranties executed by Guarantor, but without derogation to such existing guaranties, if any, which are hereby ratified and reaffirmed.

- 2. The liability of Guarantor hereunder shall not exceed at any one time the sum of FIFTEEN MILLION AND NO/100 DOLLARS (\$15,000,000.00) for principal, plus all interest thereon and all attorneys' fees and other costs and expenses incurred by Lender in collecting, compromising or enforcing the indebtedness or in protecting or preserving any security for the indebtedness. Lender may permit the indebtedness of Borrower to exceed such maximum liability without impairing the obligation of Guarantor hereunder. Any payment by Guarantor shall not reduce Guarantor's maximum obligation hereunder, unless written notice to that effect is actually received by Lender at or prior to the time of such payment. Any payment by or recovery from Borrower, any other guarantor or any security shall be credited first to that portion of the indebtedness which exceeds the maximum obligation of Guarantor hereunder.
- 3. (a) This is a continuing guarantee that shall remain in full force and effect and includes all indebtedness arising under future transactions or under successive transactions which either continue then existing indebtedness or from time to time renew it after it has been satisfied, but shall not apply to any indebtedness created after actual receipt by Lender of written notice of the revocation of this Guarantee as to future transactions. Any such revocation of this Guarantee at any time by any Guarantor as to future transactions shall not affect the liability of any other guarantor for indebtedness of Borrower and shall not affect the liability of that Guarantor or any other guarantor for indebtedness incurred or credit committed by Lender to Borrower prior to the effective time of that revocation; this Guarantee shall remain in full force and effect as to all such indebtedness. The death of any Guarantor shall not operate as a revocation of liability hereunder of the estate of that Guarantor for indebtedness created or incurred or credit committed by Lender to Borrower subsequent to such death until actual receipt by Lender of written notice of the death of that Guarantor. Guarantor waives notice of revocation given by any other guarantor.
- (b) Guarantor agrees that to the extent Borrower or Guarantor makes any payment to Lender in connection with the indebtedness, and all or any part of such payment is subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid by Lender or paid over to a trustee, receiver or any other entity, whether under any bankruptcy act or otherwise (any such payment is hereinafter referred to as a "Preferential Payment"), then this Guarantee shall continue to be effective or shall be reinstated, as the case may be, and, to the extent of such payment or repayment by Lender, the indebtedness or part thereof intended to be satisfied by such Preferential Payment shall be revived and continued in full force and effect as if said Preferential Payment had not been made.
- 4. Guarantor is providing this Guarantee at the instance and request of Borrower to induce Lender to extend or continue financial accommodations to Borrower. Guarantor hereby represents and warrants that Guarantor is and will continue to be fully informed about all aspects of the financial condition and business affairs of Borrower that Guarantor deems relevant to the obligations of Guarantor hereunder and hereby waives and fully discharges Lender from any and all obligations to communicate to Guarantor any information whatsoever regarding Borrower or Borrower's financial condition or business affairs.
- 5. Guarantor authorizes Lender, without notice or demand and without affecting Guarantor's liability hereunder, from time to time, to: (a) renew, modify, compromise, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of the indebtedness or any part thereof, including increasing or decreasing the rate of interest thereon; (b) release, substitute or add any one or more endorsers, Guarantor or other guarantors; (c) take and hold security for the payment of this Guarantee or the indebtedness, and enforce, exchange, substitute, subordinate, waive or release any such security; (d) proceed against such security and direct the order or manner of sale of such security as Lender in its discretion may determine; and (e) apply any and all payments from Borrower, Guarantor or any other guarantor, or recoveries from such security, in such order or manner as Lender in its discretion may determine.
- 6. Guarantor waives and agrees not to assert: (a) any right to require Lender to proceed against Borrower or any other guarantor, to proceed against or exhaust any security for the indebtedness, to pursue any other remedy available to Lender, or to pursue any remedy in any particular order or manner; (b) the benefit of any statute of limitations affecting Guarantor's liability hereunder or the enforcement thereof; (c) demand, diligence, presentment for payment, protest and demand, and notice of extension, dishonor, protest, demand, nonpayment and acceptance of this Guarantee; (d) notice of the existence,

creation or incurring of new or additional indebtedness of Borrower to Lender; (e) the benefits of any statutory provision limiting the liability of a surety, including without limitation the provisions of A.R.S. Sections 12-1641, ET SEQ.; (f) any defense arising by reason of any disability or other defense of Borrower or by reason of the cessation from any cause whatsoever (other than payment in full) of the liability of Borrower for the indebtedness; and (g) the benefits of any statutory provision limiting the right of Lender to recover a deficiency

-2-

judgment, or to otherwise proceed against any person or entity obligated for payment of the indebtedness, after any foreclosure or trustee's sale of any security for the indebtedness, including without limitation the benefits, if any, to Guarantor of A.R.S. Section 33-814. Guarantor shall have no right of subrogation and hereby waives any right to enforce any remedy which Lender now has, or may hereafter have, against Borrower, and waives any benefit of, and any right to participate in, any security now or hereafter held by Lender.

- 7. All existing and future indebtedness of Borrower to Guarantor is hereby subordinated to the indebtedness of Borrower to Lender and such indebtedness of Borrower to Guarantor, if Lender so requests, shall be collected, enforced and received by Guarantor as trustee for Lender and shall be paid over to Lender on account of the indebtedness of Borrower to Lender, but without reducing or affecting in any manner the liability of Guarantor under the other provisions of this Guarantee.
- 8. In addition to all liens upon, and rights of setoff against, the monies, securities or other property of Guarantor given to Lender by law, Lender shall have a lien and a right of setoff against, and Guarantor hereby grants to Lender a security interest in, all monies, securities and other property of Guarantor now and hereafter in the possession of or on deposit with Lender, whether held in a general or special account or deposit, or for safekeeping or otherwise; every such lien and right of setoff may be exercised without demand upon or notice to Guarantor. No lien or right of setoff shall be deemed to have been waived by any act or conduct on the part of Lender, by any neglect to exercise such right of setoff or to enforce such lien, or by any delay in so doing.
- 9. If Borrower is a corporation or partnership, it is not necessary for Lender to inquire into the powers of Borrower or the officers, directors, partners or agents acting or purporting to act on its behalf, and any indebtedness made or created in reliance upon the professed exercise of such powers shall be guaranteed hereunder.
- 10. Guarantor agrees to pay all attorneys' fees and all other costs and expenses which may be incurred by Lender in enforcing this Guarantee.
- 11. The obligations of Guarantor hereunder are joint and several if Guarantor is more than one person or entity, are separate and independent of the obligations of Borrower and of any other guarantor, and a separate action or actions may be brought and prosecuted against Guarantor whether action is brought against Borrower or any other guarantor or whether Borrower or any other guarantor is joined in any action or actions. The obligations of Guarantor hereunder shall survive and continue in full force and effect until payment in full of the indebtedness is actually received by Lender, notwithstanding any release or termination of Borrower's liability by express or implied agreement with Lender or by operation of law and notwithstanding that the indebtedness or any part thereof is deemed to have been paid or discharged by operation of law or by some act or agreement of Lender. For purposes of this Guarantee, the indebtedness shall be deemed to be paid only to the extent that Lender actually receives immediately available funds and to the extent of any credit bid by Lender at any foreclosure or trustee's sale of any security for the indebtedness.

-3-

- 12. This Guarantee sets forth the entire agreement of Guarantor and Lender with respect to the subject matter hereof and supersedes all prior oral and written agreements and representations by Lender to Guarantor. No modification or waiver of any provision of this Guarantee or any right of Lender hereunder and no release of Guarantor from any obligation hereunder shall be effective unless in a writing executed by an authorized officer of Lender.
- 13. This Guarantee shall inure to the benefit of Lender and its successors and assigns and shall be binding upon Guarantor and its heirs, personal representatives, successors and assigns. Lender may assign this Guarantee in whole or in part without notice.
- 14. This Guarantee shall be governed by and construed according to the laws of the State of Arizona.
- 15. Guarantor represents and warrants to Lender that: (a) (if Guarantor is not a natural person) it is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization; (b) Guarantor has full capacity and authority to execute, deliver and perform this Guarantee, and the execution, delivery and performance of this Guarantee will not (i) violate any law or regulation, (ii) (if Guarantor is not a natural person) violate any provision of Guarantor's organizational documents, (iii) violate or

constitute (with due notice or lapse of time or both) a default under any indenture, agreement, license or other instrument to which Guarantor is a party or by which Guarantor or any of Guarantor's properties may be bound, (iv) violate any order of any court, tribunal or governmental agency binding on Guarantor or any of Guarantor's properties, or (v) result in the creation or imposition of any lien of any nature whatsoever on any of Guarantor's properties or assets, (c) no approval or consent of, or filing or registration with, any federal, state or local regulatory authority is required in connection with the execution, delivery and performance of this Guarantee; and (d) this Guarantee constitutes the legal, valid and binding obligation of Guarantor, enforceable against Guarantor in accordance with its terms. These representations and warranties shall survive the execution of this Guarantee. As used in this paragraph, "insolvent" means the present fair saleable value of assets is less than the probable amount required to be paid on existing debts when and as they mature.

16. JURY WAIVER. THE UNDERSIGNED AND LENDER (BY ITS ACCEPTANCE HEREOF) HEREBY VOLUNTARILY, KNOWINGLY, IRREVOCABLY AND UNCONDITIONALLY WAIVE ANY RIGHT TO HAVE A JURY PARTICIPATE IN RESOLVING ANY DISPUTE (WHETHER BASED UPON CONTRACT, TORT OR OTHERWISE) BETWEEN OR AMONG THE UNDERSIGNED AND LENDER ARISING OUT OF OR IN ANY WAY RELATED TO THIS DOCUMENT OR ANY OTHER RELATED DOCUMENT OR ANY RELATIONSHIP BETWEEN THE UNDERSIGNED AND LENDER. THIS PROVISION IS A MATERIAL INDUCEMENT TO LENDER TO PROVIDE THE FINANCING DESCRIBED HEREIN OR IN THE OTHER RELATED DOCUMENTS.

IN WITNESS WHEREOF these presents are executed as of the \_\_\_\_ day of \_\_\_\_\_, 19\_\_\_.

-4-GUARANTOR:

Ву:			
-		 	
Name:	:		
Its:			

Address: 6613 North Scottsdale Road, Suite 200 Scottsdale, Arizona 85250

-5-EXHIBIT "F"

ASSIGNMENT AND ACCEPTANCE

\_\_\_\_\_, 19\_\_\_\_

Reference is made to the Credit Agreement dated as of September 17, 1999 (the "Credit Agreement"), among MERITAGE CORPORATION, a Maryland corporation (the "Borrower"), the lenders named therein (the "Banks") and CALIFORNIA BANK & TRUST, a California banking corporation, as Administrative Agent for the Banks (in such capacity, the "Administrative Agent"). Terms defined in the Credit Agreement and not otherwise defined herein shall have the meanings assigned to such terms in the Credit Agreement.

- 1. The Assignor hereby sells and assigns, without recourse, to the Assignee, and the Assignee hereby purchases and assumes, without recourse, from the Assignor, effective as of the Effective Date set forth on the reverse hereof, the interests set forth on the reverse hereof (the "Assigned Interest") in the Assignor's rights and obligations under the Credit Agreement including, without limitation, the interests set forth on the reverse hereof in the Commitment of the Assignor on the Effective Date and the Loans owing to the Assignor which are outstanding on the Effective Date, together with unpaid interest accrued on the assigned Loans to the Effective Date and the amount, if any, set forth on the reverse hereof of the Fees accrued to the Effective Date for the account of the Assignor. Each of the Assignor and the Assignee hereby makes and agrees to be bound by all the representations, warranties and agreements set forth in Section 10.1 of the Credit Agreement, a copy of which has been received by each such party. From and after the Effective Date (i) the Assignee shall be a party to and be bound by the provisions of the Credit Agreement and, to the extent of the interests assigned by this Assignment and Acceptance, have the rights and obligations of a Bank thereunder and under the Credit Documents and (ii) the Assignor shall, to the extent of the interests assigned by this Assignment and Acceptance, relinquish its rights and be released from its obligations under the Credit Agreement.
- 2. This Assignment and Acceptance is being delivered to the Administrative Agent together with (i) the Notes evidencing the Loans included in the Assigned Interest, and (ii) if the Assignee is not already a Bank under the Credit Agreement, an Administrative Details Reply Form in the form of Exhibit "G" to the Credit Agreement.

		ith the laws	and Acceptance of the State of		governed by	and construed in
Lega	ıl Name o	f Assignor:				
Lega	ıl Name o	f Assignee:				
Assi	.gnee's A	ddress for N	Motice:			
be f		n 5 Business	ment (may not Days after the			
				forth, to percentage	at least 8 of the Com Commitments	Commitment (set decimals, as a mitment and the of all Banks
		Principal A	mount Assigned			
	nitment gned:	\$		o		
Loan	ıs:					
	: Assigne any):	d				
on t		t forth above se side hered d to:			Accepted	
		, as Assig	nor			
Ву _				Ву		
	Its			Its		
		, as Assig	mor			
Ву _				Ву		
	Its			Its		
				2- BIT "G"		
			ADMINISTRATIVE D		FORM	
			ADMINISTRATIVE L	LIAILS KEFLI	rorm	
Re:	Loans f	or Meritage	Corporation			
1.	NAME OF	ENTITY FOR	SIGNATURE PAGE:			
2.	APPEAR	ENTITY AS I IN ANY PUBLI ferent than	CITY:			
3.		PERSON TO F AGREEMENT AT	ECEIVE DRAFT			
4.		PERSON TO S	IGN			
5.	CONTACT	S:	CREDIT CONTRACT	OPERATIC	NS CONTACT	LEGAL COUNSEL
	Name:					
	Title:					
	Address	:				
					<del></del>	
	m=1 · 1					
	Telepho	ne:				

Facsimile #:		
Telex #:		
Answerback: PAYMENT INSTRUCTIONS	3:	
Method of Payment:	Fedwire	Chips
Pay to:		
Name of Bank:		
City, State, Zip:		
ABA Number:		Reference:
Account Number:		Account Name:
Attention:		

6.

## <ARTICLE> 5

<CURRENCY> U.S. DOLLARS

<\$>	<c></c>	
<period-type></period-type>	9-MOS	
<fiscal-year-end></fiscal-year-end>		DEC-31-1999
<period-end></period-end>		SEP-30-1999
<exchange-rate></exchange-rate>		1
<cash></cash>		3,835,754
<securities></securities>		0
<receivables></receivables>		2,004,365
<allowances></allowances>		0
<inventory></inventory>		180,151,276
<current-assets></current-assets>		201,238,452
<pp&e></pp&e>		6,903,751
<pre><depreciation></depreciation></pre>		2,871,277
<total-assets></total-assets>		224,850,764
<current-liabilities></current-liabilities>		44,930,765
<bonds></bonds>		95,733,724
<preferred-mandatory></preferred-mandatory>		54,749
<preferred></preferred>		0
<common></common>		0
<other-se></other-se>		83,944,589
<total-liability-and-equity></total-liability-and-equity>		224,850,764
<sales></sales>		204,739,270
<total-revenues></total-revenues>		204,739,270
<cgs></cgs>		164,364,352
<total-costs></total-costs>		12,479,899
<other-expenses></other-expenses>		8,960,296
<loss-provision></loss-provision>		0
<interest-expense></interest-expense>		4,404
<income-pretax></income-pretax>		18,930,319
<income-tax></income-tax>		8,037,298
<pre><income-continuing></income-continuing></pre>		10,893,021
<discontinued></discontinued>		0
<extraordinary></extraordinary>		0
<changes></changes>		0
<net-income></net-income>		10,893,021
<eps-basic></eps-basic>		2.00
<eps-diluted></eps-diluted>		1.80

</TABLE>

# PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995 SAFE HARBOR COMPLIANCE STATEMENT FOR FORWARD-LOOKING STATEMENTS

In passing the Private Securities Litigation Reform Act of 1995 (the "PSLRA"), Congress encouraged public companies to make "forward-looking statements"(1) by creating a safe harbor to protect companies from securities law liability in connection with forward-looking statements. Meritage intends to qualify both its written and oral forward-looking statements for protection under the PSLRA.

To qualify forward-looking statements for protection under the PSLRA, a readily available written document must identify important factors that could cause actual results to differ materially from those in the forward-looking statements. Meritage provides the following information in connection with its continuing effort to qualify forward-looking statements for the safe harbor protection of the PSLRA.

Important factors currently known to management that could cause actual results to differ materially from those in forward-looking statements include, but are not limited to, the following: (i) changes in national and local economic and other conditions, such as employment levels, availability of mortgage financing, interest rates, consumer confidence, and housing demand; (ii) risks inherent in homebuilding activities, including delays in construction schedules, cost overruns, changes in government regulation, increases in real estate taxes and other local fees; (iii) changes in costs or availability of land, materials, and labor; (iv) fluctuations in real estate values; (v) the timing of home closings and land sales; (vi) Meritage's ability to continue to acquire additional land or options to acquire additional land on acceptable terms; (vii) a relative lack of geographic diversification of Meritage's operations, especially when real estate analysts are predicting that new home sales in certain markets may slow in future periods; (viii) Meritage's inability to obtain sufficient capital on terms acceptable to Meritage to fund its planned capital and other expenditures; (ix) changes in local, state and federal rules and regulations governing real estate development and homebuilding activities and environmental matters, including "no growth" or "slow growth" initiatives, building permit allocation ordinances and building moratoriums; (x) expansion by Meritage into new geographic or product markets in which Meritage has little or no operating experience; (xi) the inability of Meritage to identify acquisition candidates that will result in successful combinations; (xii) the failure of Meritage to make acquisitions on terms acceptable to Meritage, or to successfully integrate acquired operations, into Meritage; and (xiii) the loss of key employees of the Company, including Steven J. Hilton and John R. Landon.

Forward-looking statements express expectations of future events. All forward-looking statements are inherently uncertain as they are based on various expectations and assumptions concerning future events and they are subject to numerous known and unknown risks and uncertainties which could cause actual events or results to differ materially from those projected. Due to these inherent uncertainties, readers and others are urged not to place undue reliance on forward-looking statements. In addition, Meritage undertakes no obligations to update or revise forward-looking statements to reflect changed assumptions, the occurrence of anticipated events or changes to projections over time.

<sup>(1) &</sup>quot;Forward-looking statements" can be identified by use of words such as
 "expect,""believe," "estimate," "project," "forecast," "anticipate,"
 "plan," and similar expressions.